UNIVERSITY OF CAPE COAST

.

INTENTION TO USE AND ACTUAL USE OF POSTPARTUM CONTRACEPTIVES AMONG PREGNANT WOMEN IN THE MFANTSEMAN MUNICIPALITY OF CENTRAL REGION, GHANA

BY

SEBASTIAN ELIASON

THESIS SUBMITTED TO THE DEPARTMENT OF POPULATION AND HEALTH OF THE FACULTY OF SOCIAL SCIENCES, COLLEGE OF HUMANITIES AND LEGAL STUDIES, UNIVERSITY OF CAPE COAST IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR AWARD OF DOCTOR OF PHILOSOPHY DEGREE IN POPULATION AND HEALTH

CAL	L No.	
ACCES	278	
CAT. CHECKED	FINAL CHECKED	N

MAY 2016

.

SAM JONAH LIBRARY UNIVERSITY OF CAPE COAST

DECLARATION

Candidate's Declaration

I hereby declare that this thesis is the result of my own original work and that no part of it has been presented for another degree in this university or elsewhere.

Candidate's Name: Sebastian Eliason

Date 18/07/16 Signature:.....

Supervisors' Declaration

We hereby declare that the preparation and presentation of the thesis were supervised in accordance with the guidelines on supervision of thesis laid down by the University of Cape Coast.

Principal Supervisor's Name: Professor Kofi Awusabo-Asare

Date 18/07/2016 Signature.

Co-Supervisor's Name: Professor Akwasi Kumi-Kyereme

UGUAD Date 18/07/16 Signature:

ABSTRACT

Opportunities to reach women with family planning services, can occur along the continuum of care, throughout a woman's pregnancy, childbirth and postpartum. Evidence shows that integrating Postpartum Family Planning (PPFP) into Maternal, Newborn and Child Health (MNCH) programmes and services provides opportunities for increased use of family planning among women. This study assessed the intention of pregnant women in the Mfantseman Municipality of the Central Region of Ghana, to use and use postpartum contraceptives and the associated determinants. Changes in reproductive health behaviour were studied in the context of the proximate determinants framework.

It was a prospective panel study carried out in two phases (1 and 2). Initially, a total of 1,914 pregnant women attending antenatal clinics at Saltpond hospital, Mankessim, Anomabo and Biriwa health centres, were interviewed. The second phase was a follow up of 1,359 women who consented to be followed up in the postpartum period. Of that number, 1,004 were re-interviewed at least twelve months after delivery.

Prenatal intention to use family planning was a strong predictor of postpartum family planning use and decisions of women to adopt PPFP were influenced by the woman's acceptability and approval of family planning, supported by the partner. Pregnant women attending antenatal clinic and their partners need to be targeted for education and counseling on postpartum family planning by the midwives and doctors before they deliver.

iii

4

ACKNOWLEDGEMENTS

I am extremely grateful to my supervisors, Professor Kofi Awusabo-Asare and Professor Akwasi Kumi-Kyereme for their advice, constructive inputs, corrections and meticulous review of the thesis. My sincere appreciation also goes to Dr Frank Baiden for his support and encouragement.

I wish to acknowledge the cooperation of pregnant women in the four health facilities where the study was conducted. I also wish to recognize the work of the interviewers in the respective facilities, the heads and staff of the facilities who supported the conduct of the study in diverse ways.

I am particularly grateful to the Mfantseman Municipal Health Directorate and the Management of the Saltpond Municipal Hospital headed by Ms Yvonne Graham-Hayfron and Dr. Derek Bonsu respectively, for allowing the study to be undertaken in facilities in the Municipality.

DEDICATION

To the Eliason family of Accra and Takoradi, Ghana

TABLE OF CONTENTS

Content	Page
DECLARATION	ii
ABSTRACT	iii
ACKNOWLEDGEMENTS	iv
DEDICATION	v
TABLE OF CONTENTS	vi
LIST OF TABLES	xii
LIST OF FIGURES	xiv
LIST OF ACRONYMS	xv
CHAPTER ONE: POSTPARTUM FAMILY PLANNING: ISSUES	
AND PERSPECTIVES	1
Introduction	1
Background to the Study	3
Determinants of Fertility	3
Historical Perspectives in Family Planning	4
Family Planning as a Human Right	8
Poverty Reduction and Socio-Economic Benefits of Family Planning,	12
Family Planning and the Health Care System	14
Family Planning Service Delivery	16
Family Planning Products and Technologies	19
Financing Family Planning Programmes	22
Human Resource for Family Planning	24

Health Information for Family Planning	25
Leadership and Governance in Family Planning	25
Postpartum Family Planning	27
Statement of Problem	29
Objectives of the Study	32
Hypotheses of the Study	33
Rationale of the Study	33
Organization of the Thesis	36
CHAPTER TWO: CONCEPTUAL, THEORETICAL AND	
EMPERICAL ISSUES ON INTENTION TO USE AND USE OF	
POSTPARTUM FAMILY PLANNING	38
Introduction	38
Unintended Pregnancy	38
Intention to Use and Use of Family Planning	41
Factors Influencing Intention to Use Contraceptives	42
Factors Influencing Actual Use of Contraceptives	43
Theories and Models	44
Health Belief Model	46
Social Cognitive Theory	49
Diffusion of Innovations Theory	52
Transtheoretical (Stages of Change) Model	55
Theory of Planned Behaviour	58
Steps to Behaviour Change Model	62

Postpartum Contraceptive Use and Postpartum Behaviours Framework	65
Davies and Blake Model of Fertility	67
Conceptual Frameworks Adapted for the Study	70
Summary	71
CHAPTER THREE: METHODS OF DATA COLLECTION AND	
ANALYSES	75
Introduction	75
Study Design	82
Study Population	83
Target Population	84
Sampling Procedures	86
Methods of Data Collection	88
Data Collection Instruments	88
Preparatory Visits to Study Area	90
Pre-testing of Study Instruments	91
Data Collection	93
Fieldwork Challenges	98
Ethical Issues	100
Data Processing and Analyses	101
Study Limitations, their Mitigation and Quality Assurance	104
CHAPTER FOUR: SOCIO-DEMOGRAPHIC CHARACTERISTICS	
AND DETERMINANTS OF UNINTENDED PREGNANCIES	108
Introduction	108

viii

Socio-Demographic Characteristics of Respondents in Phase 1	109
Socio-Demographic Characteristics of Follow-up (Phase 2) Respondents	113
Wantedness of Current Pregnancy	113
Factors Influencing Status of Pregnancy	118
Summary	124
CHAPTER FIVE: FACTORS INFLUENCING THE INTENTION	
OF WOMEN TO USE POSTPARTUM FAMILY PLANNING	131
Introduction	131
Knowledge of Family Planning Methods by Socio-Demographic	
Characteristics	132
Past Use of Family Planning Methods	132
Intention to Adopt Postpartum Family Planning	136
Preferred Postpartum Family Planning Methods Intended to be Used	139
Knowledge of Family Planning Methods and Intention to Use PPFP	141
Past Use of Family Planning Methods and Intention to Use PPFP	145
Acceptability of Family Planning and Intention to Use Postpartum Family	
Planning	149
Summary	153
CHAPTER SIX: POSTPARTUM CONTRACEPTIVE USE:	
TRANSLATING INTENTION INTO BEHAVIOUR	160
Introduction	160
Translation of Prenatal Intentions into Actual PPFP Use	161
Timing and Preferred Postpartum Contraceptive Methods	170

Predictors of Actual Postpartum Family Planning Use	172
Summary	183
CHAPTER SEVEN: SEX COMPOSITION OF CHILDREN EVER	
BORN AND INTENDEDNESS FOR POSTPARTUM FAMILY	
PLANNING AND POSTPARTUM CONTRACEPTIVE USE AND	
FERTILITY BEHAVIOURS	185
Introduction	185
Sex Composition of Children	187
Sex Composition, Pregnancy Intendedness and Intention to Use PPFP	1 89
Concept of Breastfeeding and Sexual Abstinence as Elements for Spacing	191
Duration of Breastfeeding, Postpartum Amenorrhoea and Sexual Abstinence	196
Duration of Breastfeeding	196
Duration of Postpartum Amenorrhoea	200
Duration of Postpartum Sexual Abstinence	200
Postpartum Contraceptive Use Before and After Susceptibility	201
Summary	202
CHAPTER EIGHT: SUMMARY, CONCLUSIONS, POLICY	
IMPLICATIONS AND RECOMMENDATIONS	208
Introduction	208
Summary	208
Key Findings of the Study	209
Conclusions	213
Evaluation of Conceptual Framework	215

Contribution to Knowledge	219
Policy Implications	221
Recommendations	222
Areas for Further Research	223
BIBLIOGRAPHY	225
APPENDICES	259
1: Ethical Clearance	259
2: Phase 1 Questionnaire	260
3: Phase 2 Questionnaire	270
4: Published Articles	281

.

. . .

LIST OF TABLES

Ta	ble	Page
1:	Steps to Behaviour Change Constructs	64
2:	Trends in Maternal Health Indicator Coverages 2008 - 2010	80
3:	Pregnant Women 15 to 49 years Attending Antenatal Clinic at Least	
	Once at Selected Health Facilities in Mfantseman Municipal (2008 – 2010) 85
4:	Sample Size by Selected Health Facility	87
5:	Number of Instruments Administered and Response Rates by Health	
	Facility	93
6:	Number and Percentage of Women Targeted and Interviewed per	
	Health Facility Attended in Phases 1 and 2	97
7:	Socio-Demographic Characteristics of Respondents	110
8:	Socio-Demographic Characteristics of Women by Pregnancy Status	116
9:	Logistic Regression Analyses of Unintended Pregnancy and	
	Selected Independent Variables: Models I and II (Bivariate) and	
	Model III (Multivariate)	120
10	: Knowledge of Family Planning Methods by Socio-Demographic	
	Characteristics (%)	133
11	: Past Use of Family Planning Methods	136
12	: Intention to Use PPFP by Socio-Demographic Characteristics	137
13	: PPFP Methods Women Intended to Use	140
14	: Knowledge and Intention to Use PPFP	142
15	: Logistic Regression of Intention to Use PPFP and Knowledge of FP	144

16: Past Use of Family Planning Methods and Intention to Use PPFP	147
17: Logistic Regression of Intention to Use PPFP and Past Use of FP	148
18: Acceptability of Family Planning and Intention to Use PPFP	151
19: Logistic Regression of Intention to Use PPFP and Acceptability	
of FP	152
20: Logistic Regression of Intention to Use PPFP and Most Significant	
Knowledge, Past Use and Acceptability Variables	154
21: Prenatal Intention and Actual PPFP Use	163
22: Association of Prenatal Intention and PPFP Use	167
23: Prenatal FP Intentions and PPFP Outcomes of Women	168
24: Timing of PPFP Use Following Delivery	171
25: Postpartum FP Methods Used by Women Following Birth	171
26: Reasons for Method Choices	172
27: Logistic Regression of Socio-Demographic Characteristics,	
Postpartum Behaviours and Postpartum Contraceptive Use	174
28: Reasons for Use and Non-Use of PPFP	182
29: Family Sex Composition, Pregnancy Intendedness and PPFP Intentions	190
30: Sex Composition and Women's Reproductive Health Outcomes	192
31: Breastfeeding and Sexual Abstinence by Background Characteristics	194
32: Median Duration (Months) of Breastfeeding, Sexual Abstinence and	
Postpartum Amenorrhoea	197
33: PPFP Use Before and After Susceptibility to Pregnancy	203
34: Study Findings Related to Steps to Behavior Change Constructs	216

LIST OF FIGURES

.

Figure	
1: Pillars of Health System	15
2: Family Planning Methods Appropriate for Women in the	
Postpartum Period	28
3: Health Belief Model	48
4: Diffusion of Innovation Model	54
5: Stages of Change Model	56
6: Theory of Planned Behaviour Model	60
7: Steps to Behaviour Change Model	63
8: Postpartum Contraceptive Use and Postpartum Behaviours Framework	66
9: Davies and Blake Model of Fertility	69
10: Adaptation of Davies and Blake Model 1956	
11: Family Planning Acceptor Rate by District (%), Central Region	
2007 – 2009	76
12: Map of Central Region Showing Mfantseman Municipal Area	77
13: Map of Mfantseman Municipal Area Showing Selected Health	
Facilities Used for the Study	79
14: Age and Unintended Pregnancy	114
15: Education and Unintended Pregnancy	114

xiv

.

1. C.

LIST OF ACRONYMS

AAK	Abura Asebu kwaman
AEE	Agoafo Eyan Esiam
AIDS	Acquired Immune Deficiency Syndrome
ANC	Antenatal Clinic
AOR	Adjusted Odds Ratio
BCC	Behaviour Change Communication
CBA	Cost-Benefit Analysis
CBS	Community-Based Service Providers
CBV	Community-Based Volunteer
Cc	Index of Contraception
CC	Cape Coast
CEDAW	Convention on the Elimination of all forms of
	Discrimination Against Woman
	Discrimination Against women
CHOs	Community Health Officers
CHOs CHP S	Community Health Officers Community Based Health Planning and Services
CHOs CHP S Cm	Community Health Officers Community Based Health Planning and Services Index of Proportion Married
CHOs CHP S Cm CPR	Community Health Officers Community Based Health Planning and Services Index of Proportion Married Contraceptive Prevalence Rate
CHOs CHPS Cm CPR CR	Community Health Officers Community Based Health Planning and Services Index of Proportion Married Contraceptive Prevalence Rate Central Region
CHOs CHP S Cm CPR CR CRC	Community Health Officers Community Based Health Planning and Services Index of Proportion Married Contraceptive Prevalence Rate Central Region Convention on the Right of the Child
CHOs CHP S Cm CPR CR CR CRC DFID	Community Health Officers Community Based Health Planning and Services Index of Proportion Married Contraceptive Prevalence Rate Central Region Convention on the Right of the Child Department for International Development
CHOs CHPS Cm CPR CR CR CRC DFID DHS	Community Health Officers Community Based Health Planning and Services Index of Proportion Married Contraceptive Prevalence Rate Central Region Convention on the Right of the Child Department for International Development Demographic and Health Survey

DRC Democratic Republic of Congo Expanded Programme on Immunization EPI Female Genital Mutilation FGM Family Planning FP Family Planning and Maternal and Child Health Programme FPMCH Ghana Demographic and Health Survey GDHS GHS Ghana Health Service Ghana Statistical Service GSS HBM Health Belief Model Human Immune Deficiency Syndrome HIV International Conference on Population and Development ICPD POA Programme of Action IPPF International Planned Parenthood Federation IUD Intrauterine Device JHPIEGO Johns Hopkins Program for International Education in Gyneacology and Obstetrics KEEA Komenda Edina Eguafo Abirem LAM Lactional Amenorrhoea Method LMIC Low- and Middle-Income Countries MCHIP Maternal and Child Health Integrated Program MDG Millennium Development Goals MHD Municipal Health Directorate MICS Multiple Indicator Cluster Survey

MNCH	Maternal Newborn and Child Health
MOH	Ministry of Health
MSI	Marie Stopes International
NGO	Non-Governmental Organization
NPC	National Population Council
OHCHR	Office of the High Commission for Human Rights
OPD	Out Patient Department
PhD	Doctor of Philosophy
PPAG	Planned Parenthood Association of Ghana
PPFP	Postpartum Family Planning
RCH	Reproductive and Child Health
RH	Reproductive Health
SSA	Sub-Saharan Africa
SSS	Seniour Secondary School
STBCM	Steps to Behaviour Change Model
SWOP	State of the World Population
TBA	Traditional Birth Attendants
TFR	Total Fertility Rate
THLD	Twifo Heman Lower Denkyira
ТМ	Total Marital Fertility
TNM	Total Natural Marital Fertility
TPB	Theory of Planned Behavior
TTM	Transtheoretical Model

UCC Unive	sity of Cape Coast
-----------	--------------------

UDHR Universal Declaration of Human Rights

UN United Nations

UNFPA United Nations Population Fund

US United States

USAID United States Agency for International Development

USCB United States Census Bureau

USDHHS United States Department of Health and Human Services

WHO World Health Organization

WRA Women in Reproductive Age

CHAPTER ONE

POSTPARTUM FAMILY PLANNING: ISSUES AND PERSPECTIVES

Introduction

According to the World Health Organization (WHO), family planning (FP) involves the promotion of maternal and child health (MCH), by helping individuals and couples attain the desired number and composition of their children and empowering them to decide, if and when to become pregnant, thus preventing unintended pregnancies and reducing the need for unsafe abortions (WHO, 2013). Global estimates indicate that about 200 million women who want to avoid pregnancy are either not using any means of modern family planning methods or are using only traditional family planning methods, which are considered ineffective (Singh, Darroch, Ashford, & Vlassoff, 2009). Similarly, 40 percent of the 186 million pregnancies that occur annually in developing countries are unintended, with about 82 percent of them having unmet need for modern family planning (Singh et al., 2009). Reasons for these observations include cultural or religious opposition, gender-based barriers, poor quality of available maternal health services, fear or experience of side effects of modern contraceptives, limited choice of modern contraceptive methods and limited access to contraception, especially among young people, poorer segments of populations, or unmarried people (World Health Organization, 2013).

Opportunities to reach women with family planning services, can occur along the continuum of maternal care, throughout pregnancy, childbirth and postpartum. Evidence shows that, 50 percent to 60 percent of pregnant women make prenatal visits (Ross & Winfrey, 2001) or have multiple contacts with health care providers soon after delivery, during infant care and for other child health services (Ross & Winfrey, 2001; Singh et al., 2009). Integrating Postpartum Family Planning (PPFP) into Maternal, Newborn and Child Health (MNCH) programmes and services provides opportunities for improved care and expansion of services within the first year after delivery for women, increased use of family planning among women, reduction of high risk pregnancies, reduced unmet need for family planning and improvements in the health and survival of mothers and children (WHO, 2013).

This thesis examines the intentions of pregnant women to adopt postpartum family planning and the extent to which their intentions are translated into actual use after delivery in the Mfantseman Municipality of the Central Region of Ghana. The determinants of unintended pregnancies among these pregnant women were also examined. These objectives were informed by the fact that the municipal area is noted for low family planning uptake, high unmet need for contraception and high teenage pregnancy (Ghana Health Service, Central Region, 2010). Sex composition and its influence on fertility behaviour is also assessed and was informed by a study on the dynamics of reproductive health behaviours in rural coastal communities of southern Ghana by Akinyoade (2007), which revealed that, individuals in southern coastal communities preferred sons to daughters for propagation of their business.

2

.

Background to the Study

Determinants of Fertility

Fertility, an outcome of reproductive behaviour, has been a subject of debate in the past amongst ancient philosophers such as Plato, Aristotle and followers of Hippocrates. It is said that Plato when asked how population stability could be attained, answered, "if too many children are being born, there are measures to check propagation; on the other hand, a high birth-rate can be encouraged and stimulated by conferring marks of distinction or disgrace" (Pollard, 1994:p. 275). Debates on fertility have been ongoing since those ancient times till date, with the development of various perspectives, frameworks and theories, to explain fertility determinants and transitions, from the pre- to post-transition periods.

•

Davis and Blake (1956) proposed an analytical framework for the comparative sociology of fertility. They classified intermediate or direct determinants, through which, and only through which any indirect determinants (social, economic, cultural and environmental factors) could influence fertility. Three necessary steps required in the process of human reproduction were suggested - intercourse, conception and gestation and parturition, which were further classified into eleven (11) factors. Bongaarts (1978) re-classified the eleven (11) factors into eight and later seven proximate variables which, included the proportion of women of reproductive age that is married (as a measure of the proportion exposed to sexual intercourse); the use and effectiveness of contraception; induced abortion; postpartum infecundability (as primarily

3

determined by the duration and intensity of breast-feeding); the frequency of intercourse (including the effect of temporary separation and abstinence practices); the onset of permanent sterility (particularly as related to menopause) and spontaneous intrauterine mortality. Marriage, contraceptive use, abortion and postpartum infecundability have been shown to have by far the greatest effects on fertility and fertility transitions (Bongaarts & Potter, 1983; Bongaarts, 1993) These variables are so important that researchers have made them the focus of their research rather than fertility, and are looking for the determinants of these proximate determinants instead (De Bruijn, 2006). Contraceptive or family planning use, one of the most important proximate determinants of fertility is the central argument of this thesis.

Historical Perspectives in Family Planning

Available data reveal that the world's population has grown considerably, since the end of the Great Famine and Black Death in 1350 in Europe, from 370 million (Biraben, 1979) to one billion in 1804, and doubling to two billion after 123years in 1927 (Kollodge, 2011). Growth rates of over 1.8 percent per annum occurred in the early 1950s through to the 1960s and 1970s. Annual population growth rate peaked at 2.2 percent in 1963, declining to below 1.1 percent as at 2010 (Census Bureau United States, 2011). The world's population was 7 billion in 2011 (Kollodge, 2011).

These population increases over the years gave rise to streams of thought and action that culminated in the global family planning movement. The first was concerned about unintended pregnancies, unsafe abortions, high maternal deaths and women's rights and empowerment, a view which gave rise to the first family planning movement, pioneered by Margaret Sanger, Marie Stopes and others in the early years of the 20th century (Sinding, 2007). The second stream was led by neo–Malthusians such as John D. Rockefeller III in the post-World War II era who argued that imbalances between rapid population growth and a wide variety of resources, including food, had the potential to lead to political instability in developing economies (Sinding, 2007).

The main point of the first stream was that individuals could have control over fertility through universal access to contraception. The second stream believed in promoting policies that restricted population growth, but which had the tendency to restrict individual reproductive choices and freedoms. A third stream of thought championed by rights advocates, argued that, even though contraception is important, focusing on it obscures the much broader struggle for comprehensive women's rights. They envisioned high fertility rates and increased population as consequences of broader economic and gender inequities and therefore believed that premising family planning initiatives on reproductive and human rights, rather than fertility reduction, would yield the greatest benefits to women, their families, and the environment (Zehner, 2011.)

Over time, convergence of these streams of thought emerged in the proposition that: "Voluntary birth control programmes would be sufficient to bring about significant declines in fertility, and hence in population growth" (Sinding, 2007: p. 2). The commitment to voluntarism and non-coerciveness embedded in the proposition ensured that, women and men had control over their reproductive lives and avoided unintended childbearing through improved access to and information about contraceptives. This policy response to rapid population growth, occurred alongside improved educational and health policies from the 1960s (Bongaarts & Sinding, 2009).

Despite renewed debates globally on the belief by pessimists that overpopulation and over-consumption account for mankind's doom and optimists' belief that technological advancement can improve living standards and that population growth may be a minor issue (Lomborg, 2001; Ehrlich & Ehrlich, 2009), voluntary family planning programmes still remain the principal global policy for reducing fertility rates as articulated in global commitments like the International Conference on Population and Development Programme of Action (ICPD POA) and the Millennium Development Goals (MDGs).

Some critics have questioned this global position by making assertions to the contrary. First, the view of a number of economists is that, family planning programmes have little or no effect on fertility (Pritchett, 1994; Connelly, 2009) because couples have the number of children they want and can afford (Becker, 1960; Pritchett, 1994). A second view is that, family planning programmes are no longer required because of global fertility declines which imply adequate solution to the population problem (Connelly, 2009). Despite widely accepted evidence on the impact of family planning programmes, these small and determined number of critics managed to influence funding cuts to the global family planning programme by about 30 percent from the 1990s (Bongaarts & Sinding, 2009).

6

In order to restore hope in, and integrity of, the global family planning programme, counter-views to the assertions made by the critics have been proffered. Bongaarts and Sinding (2009), in their response to the first critical view stated that, the views of the economists was simplistic and did not reflect the real world. They illustrated the impact of family planning by using experiences in the rural Matlab district of Bangladesh and Iran. In Bangladesh, a 20-year fall in fertility from over six to nearly three children per woman and an increase in contraceptive use from 10 to 50 percent was experienced from the 1970s, following the introduction of a comprehensive family planning and reproductive health programme (Cleland, Phillips, Amin & Kamal, , 1994; Reseach N.I. Bangladesh , 2005), whilst in Iran, fertility levels declined from over five children per woman from 1989 to two children per woman in 2000, following the provision of free contraceptive services throughout the country by an extensive network of village health workers (Roudi-Fahimi, 2002).

In response to the second critical view, Bongaarts and Sinding (2009) stated that, instead of the unprecedented population expansion of the past 50 years getting to an end in the face of declining fertility, it continues to increase every year since 1970, with large increases in population expected in Africa (by 1.1 billion), Asia (by 1.3 billion) and Latin America (by 0.2 billion). They gave three reasons for this observation. First, the average global decline in fertility from six to three births per woman still leaves fertility substantially above two surviving children per woman and every generation larger than the preceding one, thus propagating the population expansion. Second, declines in mortality (historically

the main cause of population growth) will almost certainly continue because of higher standards of living, better nutrition, expanded health services and greater investments in public health measures which have increased life expectancy by 50 percent since 1950. Thirdly, population momentum, which is the tendency for a population to keep growing even if fertility could immediately be brought to the replacement level of 2.1 births per woman, will produce more than enough births to drive population growth for decades to come.

Views and counter-views about fertility and population dynamics have been expressed historically and will continue to be expressed for a long time to come. In the face of all these, the fact still remains that individuals and couples hold the rights to determine their own fertility options, which eventually impact on population dynamics and socio-economic development globally.

Family Planning as a Human Right

Human rights are usually understood to be the inalienable fundamental rights to which a person is inherently entitled simply because he or she is a human being. For much of human history, rights and liberties have existed in alternative forms as conceptions of justice, political legitimacy and human flourishing that sought to realize human dignity, flourishing, or well-being entirely independent of human rights (Donnelly, 2013).

The "Cyrus Cylinder" has been recognized by the United Nations as the world's first charter of human rights, which decreed religious freedoms and racial equality in Babylon around 539 B.C. Subsequently, human rights laws have been derived from natural laws in India, Greece and later Rome and documents such as the Magna Carta (1215), the Petition of Right (1628), the US Constitution (1787), the French Declaration of the Rights of Man and of the Citizen (1789), and the US Bill of Rights (1791) (United for Human Rights, n.d.).

Modern human rights began to emerge following the end of the Second World War in 1945, when the United Nations Charter was launched. It was soon followed in 1948 by the adoption of the Universal Declaration of Human Rights (UN General Assembly, 1948). Although, this declaration did not mention family planning specifically, it mentioned the Family as the basic unit of society. It was upon this premise that the World leaders at the Human rights day celebration in 1966 made a declaration on population that "The majority of parents desire to have the knowledge and the means to plan their families; that the opportunities to decide the number and spacing of children are a basic human right." To add impetus to the declaration, the United Nations Secretary-General, U Thant, in a statement during the signing of the 1966 declaration by World Leaders in 1967 on human rights day, said:

"The Universal Declaration of Human Rights describes the family as the natural and fundamental unit of society. It follows that any choice and decision with regard to the size of the family must inevitably rest with the family itself, and cannot be made by anyone else. But this right of parents to free choice will remain illusory unless they are aware of the alternatives open to them. Hence, the right of every family to information and the availability of services in the field is derived from natural laws in India, Greece and later Rome and documents such as the Magna Carta (1215), the Petition of Right (1628), the US Constitution (1787), the French Declaration of the Rights of Man and of the Citizen (1789), and the US Bill of Rights (1791) (United for Human Rights, n.d.).

Modern human rights began to emerge following the end of the Second World War in 1945, when the United Nations Charter was launched. It was soon followed in 1948 by the adoption of the Universal Declaration of Human Rights (UN General Assembly, 1948). Although, this declaration did not mention family planning specifically, it mentioned the Family as the basic unit of society. It was upon this premise that the World leaders at the Human rights day celebration in 1966 made a declaration on population that "The majority of parents desire to have the knowledge and the means to plan their families; that the opportunities to decide the number and spacing of children are a basic human right." To add impetus to the declaration, the United Nations Secretary-General, U Thant, in a statement during the signing of the 1966 declaration by World Leaders in 1967 on human rights day, said:

"The Universal Declaration of Human Rights describes the family as the natural and fundamental unit of society. It follows that any choice and decision with regard to the size of the family must inevitably rest with the family itself, and cannot be made by anyone else. But this right of parents to free choice will remain illusory unless they are aware of the alternatives open to them. Hence, the right of every family to information and the availability of services in the field is increasingly considered as a basic human right and as an indispensable ingredient of human dignity.' (Thant, 1967)

Following these preparatory declarations and statements, rights to family planning finally emerged as an integral part of human rights in the 1968 Teheran proclamation, during the United Nations International Conference on Human Rights (United Nations, 1968). The proclamation extended the 1945 UN Charter and the 1948 Universal Declaration of Human rights into family planning and reproductive health and stated that:

"Parents have a basic right to decide freely and responsibly on the number and spacing of their children and a right to adequate education and information in this respect" - (United Nations, 1968:p.15)

Implicit in this is the right of men and women to have access to safe, effective, affordable and acceptable methods of contraception of their choice, as well as other methods of their choice to regulate fertility which are not against the law (Krishnamurty, 2000). The right to family planning was considered as part of the right to life, the right to the highest attainable standard of health, the right to decide the number and spacing of one's children, the right to privacy, the right to information, and the right to equality and non-discrimination (International Planned Parenthood Federation, 2012)

10

Globally influential statements and declarations that have reaffirmed family planning as a human right include the Programme of Action of the International Conference on Population and Development (ICPD POA), Convention on the Elimination of All Forms of Discrimination Against Women (CEDAW), Convention on the Rights of the Child (CRC), Beijing Platform for Action, the Millennium Declaration, Paris Declaration on Aid Effectiveness & Accra Agenda for Action, the UN Global Strategy for Women's & Children's Health, Muskoka Initiative on Maternal, Newborn & Child Health, and the protocol to the African Charter on Human and Peoples Rights (IPPF, 2012).

Family planning as a human right has not been spared the Universalism versus cultural relativism debates and controversies. The 1993 Vienna declaration sought to protect and promote strongly, universalism of family planning, like all other human rights. The declaration stated thus:

"All human rights are universal, indivisible and interdependent and interrelated. The international community must treat human rights globally in a fair and equal manner, on the same footing, and with the same emphasis. While the significance of national and regional particularities and various historical, cultural and religious backgrounds must be borne in mind, it is the duty of States, regardless of their political, economic and cultural systems, to promote and protect all human rights and fundamental freedoms" (United Nations, 1993:p.20) The declaration took note of the threat of cultural relativism and sought to further re-emphasize the need for all states despite their cultural differences, to still protect human rights and fundamental freedoms. However, cultural relativists have presented a challenge to the propagation of universal human rights worldwide.

Views of cultural relativists vary. One view is that human rights, legal rights and obligations have not considered cultural histories, traditions and regional perspectives such as those of Africa or Asia (Cobbah, 1987; Bell, Nathan, & Peleg, 2001). Another view is that cultures have been hurriedly subjected to human rights and legal norms that those cultures were not prepared to accept (Teson, 1984). A third view is that, it is wrong for cultures to be subjected to human rights and legal norms that are foreign to them (Mayer & Mayor, 1991). In all these views, one common position held is that human rights were perceived by the West, implemented by the West and continues to be upheld by the West (Perrin, 2005). Although, evidence exists that non-western states played major parts in the process, there is still not a universal acceptance of these rights among all cultures and therefore new approaches to universalizing human rights are required (Perrin, 2005) to ensure socio-economic development.

Poverty Reduction and Socio-Economic Benefits of Family Planning,

Another argument which has emerged is the cost-effectiveness of family planning in socio-economic development in the developing world (Reproductive Health Supplies Coalition, 2009; Smith, Ashford, Gribble, & Clifton, 2009). A global cost-benefit analysis revealed that expenditures of up to US\$3.9 billion on

contraceptives for women in the poorest of developing countries could prevent 52 million unintended pregnancies and 22 million abortions, and save on health care costs (Singh et al., 2009). Evidence from Bangladesh also showed that long-term investment in an integrated family planning and maternal and child health (FPMCH) programme contributed to improved economic security for families, households, and communities through larger incomes, greater accumulation of wealth, and higher levels of education (Smith et al., 2009). Further evidence showed that every US dollar spent on family planning saved at least US\$4 that could have been spent treating complications from unintended pregnancies (Singh . et al., 2009; Speidel, Sinding, Gillespie, Maguire, & Neuse, 2009) and saved governments up to US\$31 in health care, water, education, housing, sewers, and many more (Singh et al., 2009)

The argument is that, if individuals and couples plan their pregnancies and families with improved use of contraception, they are more likely to have fewer and healthier children. This will reduce the socio-economic burden on them and allow more investment in each child's care including education, helping to break the poverty cycle. Opposing views to the assertion that family planning programmes are cost-effective exist. The assertion is that since many couples in developing countries perceive that they are better off with large families, the best way to reduce fertility is not by investing in family planning, but by improving women's education and improving their health and status in society (Pritchett, 1994).

Despite these opposing views, the fact still remains that many lives are being lost through unintended pregnancies. For example, 76 million unintended pregnancies in the developing world in 2003 resulted in 184,000 pregnancyrelated deaths and 1.8 million infant deaths (Vlassoff, Singh, Darroch, Carbone & Bernstein, 2004). According to the World Bank, part of these deaths could have been averted by improving contraceptive use and hence made gains of \$100 per life year saved (Jamison et al., 2006). These gains would be difficult to make without improvements in access to contraceptives and strengthened family planning services. According to Ringheim (2011), the highly effective and economical way of preventing these deaths was by educating women about and offering them family planning in the postpartum period. The global postpartum family planning programme started on this premise.

Family Planning and the Health Care System

Family planning services have seen evolution since the launch of the pioneering works of Margaret Sangers and Marie Stopes (Bongaarts et al 2012). The evolution involved the changing content of services from the provision of only condoms and diaphragms to include family planning education, counseling, contraception for sexually active young people and women and men with disabilities, treatment for infertility, safe abortion (where legal) and post-abortion care, reproductive health needs of peri-menopausal women, advocacy related to eliminating practices such as female genital mutilation (FGM) and Sexual and

gender-based violence depending on the socio-cultural setting (Bongaarts et al., 2012).

Family planning services are now understood to be an integral part of reproductive health in particular and the health system in general. Improving family planning service performance depends on six factors, namely: service delivery, human resource, health information systems, access to essential medicines and technologies, health financing and leadership/governance (WHO, 2010) Leadership/governance and health information systems are cross-cutting components that provide the basis for the overall policy and regulation of the other system blocks. Financing and human resource are key input components, while medical products/technologies and service delivery reflect immediate system outputs (see Figure 1)



Figure 1: Pillars of Health System

Source: WHO (2010)

Family Planning Service Delivery

This is the most critical of all the pillars of health systems strengthening and ensures delivery of effective, safe, and high-quality family planning services, in an efficient and acceptable way to individuals, especially those most in need, and to communities served (Bongaarts, Cleland, Townsend, Bertrand, & Gupta, 2012). The current context of family planning service provision is multi-dimensional. First, clients' family planning providers play dual roles as their usual source of health and serve as gateways for their clients to other essential health care services (Frost, 2013). Second, primary care and other providers who provide a range of health-care services are expected to integrate family planning services for all persons of reproductive age, including those whose primary reason for their health-care visit might not be family planning. Third, clients could be referred for specialist care after initial screening at a family planning visit, or could be referred from specialists to family planning providers if clients have family needs (Bongaarts et al., 2012).

In the past, static clinics were the only means of providing family planning services. These were limited and provided very few options to clients. Currently, means of providing family planning services have expanded to include door-step services, pharmacies, community-based distribution, social marketing and social franchising. These have greatly improved contraceptive uptake and access to contraceptives (Prata, Vahidnia, Potts & Dries-Daffner, 2005; Azmat et al., 2013). The main service providers for contraceptives in Ghana are the Ministry of Health/Ghana Health Service public health facilities, followed by pharmacies (including chemists and drug stores), private clinic facilities and NGOs such as Planned Parenthood association of Ghana (PPAG) and Marie Stopes International (MSI). These stakeholders provide services through Static Clinics; Community Based Distribution Points, Outreaches, Social Marketing and Social Franchising (Directory of MOH/GHS - Ghana National Reproductive Health Policy and Standards, 2003.). There continues to be a shift towards the commercial or private sector as sources of contraceptives in Ghana (Ghana: Market Segmentation Analysis, 2011). The package of services provided include Behaviour Change Communication (BCC), counseling on human sexuality and contraception, provision of full range of contraceptives, management of side effects, prevention and management of STI/condom use/safer sex, voluntary counseling and testing, logistic management and referral (Directory of MOH/GHS - Ghana National Reproductive Health Policy and Standards, 2003).

Equity in access to family planning services has been another consideration that affects contraceptive uptake. The richer have greater ability to procure contraceptives compared to those in the lower wealth quintile. In some countries like Bangladesh , door-to-door service delivery of contraceptive services ensured bridging of the equity gap, whilst in Kenya for example, the over-reliance on static clinics and private sources amongst other factors contributed to the wide gap between lower and upper wealth quintiles (Gwatkin et al., 2007). In Ghana, the richest (31%) use contraceptives more than the poorest (14%), with the richest relying more on the private facilities and pharmacies (65%) whilst the poorest rely
on the static public health clinics (62%) (Ghana: Market Segmentation Analysis, 2011).

A review of the service delivery environment in Ghana in 2011 revealed several opportunities and challenges (Meeting the Commodity Challenge: Ghana's National Reproductive Health Commodity Security Strategy 2011–2016)

Some of the opportunities were that:

- The active involvement of the private sector, community distribution networks and provision of outreach services encourage wide access to contraceptive methods in terms of geographic location and time. Furthermore, short term methods are easily available and also ensure easy access;
- Couples and individuals have strong desires and willingness to use family planning services and contraceptives and are continually aroused through education campaigns;
- The environment is conducive for education on various family planning methods and use of services and commodities;
- Most reasons for not using contraceptives are misconceptions, which can be corrected with education;
- The public sector is better trained and equipped especially for long-term and permanent methods; and
- Trainers and curriculum are available to train staff in family planning service delivery and supervision.

•

Some of the challenges identified with service provision are:

- The quality of services provided especially in the private sector remains poor. Many chemical and drug stores in the country are not manned by well trained personnel and give wrong information to clients;
- Increased demand for services are difficult to meet because of inadequate trained personnel, commodity availability and limited opening hours, especially in rural areas;
- Myths and misconceptions about some of the methods may be limiting uptake and contributing to high unmet need; and
- Inadequate number of providers trained to provide long-term and permanent methods in public and private sectors, limits access to and utilization of voluntary surgical contraception, implants and IUDs.

Therefore, interventions to improve family planning uptake, especially in rural settings need to bear in mind some of these challenges and opportunities.

Family Planning Products and Technologies

The assurance of consistent supply of safe, effective and affordable supplies of contraceptives is essential to meet the objects of all family planning programmes. To ensure contraceptive security, countries have developed Contraceptive Security Strategies which combine efficient procurement techniques and efforts to make contraceptives affordable and accessible. This is facilitated by the WHO's Essential Drug List. Ghana is currently implementing its second National Reproductive Commodity Security Strategy 2011-2016 after the expiry of the first which was implemented from 2004 to 2010. The contraceptive methods and technologies currently being provided in Ghana according to the Ghana Reproductive Health Policy and Standards (2003) are classified under Temporary (short- term and long-term) and Permanent methods as follows:

A. Temporary Methods

- a. Short --term
- Oral Microgynon (combination pill), Ovrette (Progestin-only pill), Micronor (Progestin-only pill), Pregnon (Emergency contraceptive pill); Duofem (Combination pill); Lo-femenal (combination pill), Microlut.
- 2. Injectables Depo-Provera, Norigynon, Pregnon, Secure
- Condoms Generic, Be Safe male condom, Protector Plus male condom, Female Condom
- 4. Spermicides Neo Sampoon, Today
- 5. Lactational Amenorrhoea Method (LAM)

b. Long-term

- 1. Intrauterine device (IUD) Copper T
- 2. Hormonal Implants Jadelle, Sino-implants, Implanon
- 3. Natural family planning methods

B. Permanent (Surgical)

- 1. Male-Vasectomy
- 2. Female Tubal ligation
- C. Emergency Contraception
- 1. Oral contraceptive pills

of the first which was implemented from 2004 to 2010. The contraceptive methods and technologies currently being provided in Ghana according to the Ghana Reproductive Health Policy and Standards (2003) are classified under Temporary (short- term and long-term) and Permanent methods as follows:

A. Temporary Methods

- a. Short –term
- Oral Microgynon (combination pill), Ovrette (Progestin-only pill), Micronor (Progestin-only pill), Pregnon (Emergency contraceptive pill); Duofem (Combination pill); Lo-femenal (combination pill), Microlut.
- 2. Injectables Depo-Provera, Norigynon, Pregnon, Secure
- Condoms Generic, Be Safe male condom, Protector Plus male condom, Female Condom
- 4. Spermicides Neo Sampoon, Today
- 5. Lactational Amenorrhoea Method (LAM)

b. Long-term

- 1. Intrauterine device (IUD) Copper T
- 2. Hormonal Implants Jadelle, Sino-implants, Implanon
- 3. Natural family planning methods
- B. Permanent (Surgical)
- 1. Male-Vasectomy
- 2. Female Tubal ligation
- C. Emergency Contraception
- 1. Oral contraceptive pills

of the first which was implemented from 2004 to 2010. The contraceptive methods and technologies currently being provided in Ghana according to the Ghana Reproductive Health Policy and Standards (2003) are classified under Temporary (short- term and long-term) and Permanent methods as follows:

A. Temporary Methods

- a. Short –term
- Oral Microgynon (combination pill), Ovrette (Progestin-only pill), Micronor (Progestin-only pill), Pregnon (Emergency contraceptive pill); Duofem (Combination pill); Lo-femenal (combination pill), Microlut.
- 2. Injectables Depo-Provera, Norigynon, Pregnon, Secure
- Condoms Generic, Be Safe male condom, Protector Plus male condom, Female Condom
- 4. Spermicides Neo Sampoon, Today
- 5. Lactational Amenorrhoea Method (LAM)

b. Long-term

- 1. Intrauterine device (IUD) Copper T
- 2. Hormonal Implants Jadelle, Sino-implants, Implanon
- 3. Natural family planning methods

B. Permanent (Surgical)

- 1. Male-Vasectomy
- 2. Female Tubal ligation
- C. Emergency Contraception
- 1. Oral contraceptive pills

2. Intrauterine Device (IUD)

The following methods are provided at the different levels of the health system based on the category of staff at each level:

- a. Community level condoms (male/female), spermicides, oral contraceptives and LAM;
- b. Sub-district level condoms (male/female), spermicides, oral contraceptives, LAM, natural family planning methods, hormonal implants, injectables, IUD and emergency contraception;
- c. District level condoms (male/female), spermicides, oral contraceptives,
 LAM, natural family planning methods, hormonal implants, injectables,
 IUD, voluntary surgical contraception (tubal ligation and vasectomy) and
 emergency; and
- d. Regional and Teaching hospitals same as at district level.

Opportunities for ensuring contraceptive commodity security in Ghana, include but not limited to the fact that there is a policy that enables local procurement and manufacture of commodities and a strong donor support for commodity procurement, there exist sufficient quality control procedures and protocols and clear registration guidelines that are implemented by the Food and Drugs Authority and the current social marketing promotion strategies motivate more clients to accept commodities, especially the short-term contraceptive methods. Stockouts mainly due to inadequate flow of funds, transportation challenges, poor stock and financial management at the national, regional and facility levels respectively, is the main challenge affecting availability of commodities and technologies in Ghana (Ghana Health Policy Project, 2012).

Financing Family Planning Programmes

Financing of family planning programmes is essential to ensure access to family planning services and contraceptive methods and technologies and also to provide incentives as motivation for family planning service providers. Investments in family planning provide very high return in savings in other development sectors like education and health (Singh et al., 2009). In developing countries, the main source of funding for contraceptives has been from development partners since the 1990s. Funding had increased from an estimated US\$79 million in 1990 to an estimated US\$223 million by 2007 (Gribble, 2010) and as much as \$408 million may be required by 2015 to respond to unmet need for family planning (Reproductive Health Supplies Coalition, 2009). Changing donor priorities worldwide like donors transition to more sustainable partnerships with national governments, health sector-wide approaches (SWAPs), direct budget Support from donors, special global funds that focus on specific health issues Such as HIV/AIDS & malaria and national development agendas based on poverty reduction strategies which may not include family planning, have prompted a concern that family planning programmes may be adversely affected if other sustainable ways of funding are not explored.

In response to changes in development assistance, many governments have become more active in supporting family planning efforts. Some strategies that have been developed to meet the challenges include creation of budget line items in national budgets and committing to respond favourably to them, including contraceptives on essential drug lists and designating them as "essential commodities" and integrating family planning into national health insurance programmes and other benefit packages (Gribble, 2010).

In Ghana, family planning commodities had traditionally been provided to the Ghana Family Planning Programme through donations largely from USAID, DFID, and the UNFPA. The Ministry of Health (MOH) had contributed about 21percent of the total costs in recent years Increasingly, the World Bank has supported a substantial portion of the annual contraceptive financing gap mainly through basket funds (World Bank credits) (USAID, 2009)

The private sector also provides contraceptives and is the dominant source for short-term methods, while the public sector is the primary source for longerterm methods. Although family planning services including counseling are provided "free-of-charge" at public health facilities, there is usually a small charge for the commodity (averaging 10 percent of the international price). A policy to include family planning commodities under the National Health Insurance Package has been approved but the total cost of the package is yet to be computed before implementation of the policy begins (Ghana Family Planning Programme, 2015). The main challenge for family planning programme financing is that commitments by some partners are not always fulfilled (Ghana Family Planning Programme, 2015).

Human Resource for Family Planning

Health care is labour intensive and determining and establishing the right mix of personnel to safely deliver family planning services has been a major challenge for most health care organizations and health systems (World Health Organization, 2010) . Human resources should be deployed to achieve the best health outcomes within a given context, however, in developing countries where the ratio of trained health personnel to population is small and are poorly distributed, it is difficult to achieve the right mix of personnel to deliver the family planning services. Task-sharing (Higher level or lower level cadres perform the same task or perform different aspects of the same task) and task-shifting (Moving a task from higher level cadre completely to a lower level cadre in order to free the hands of the higher level cadre to concentrate on other important tasks) are being explored in developing countries where few trained medical practitioners are interested in reproductive health including family planning (Bongaarts et al., 2012).

As set out in the Ghana Reproductive Health Policy and Standards (2003), family planning services in Ghana are provided by trained service providers at various service delivery points as follows:

- a. Community level- Community based Service providers (CBS), trained Community Health Officers (CHOs), traditional birth attendants (TBAs), chemical sellers and pharmacists;
- b. Sub-district levels Midwives, Nurses, physician /medical assistants, medical practitioners;

•

- c. District level medical practitioners, physician/medical assistants, midwives, nurses, specialists;
- d. Regional level Midwives, nurses, medical practitioners and specialists; and
- e. Teaching /Specialist Centres Same as for regional.

Health Information for Family Planning

Governments, donors, programme planners and service providers require timely and accurate information for policy formulation or change, for decision making, to improve quality and coverage of family planning services, to improve programme performance and for resource mobilization among others.

For family planning programming, information generated are usually on results of surveys such as the Demographic and Health Surveys on contraceptive prevalence and unmet need by population segments, location and distribution of service delivery points, client records, procurement and supply information, trends of family planning services and products provided, and information on the location, skills, and training needs of providers and for identification of research needs (Directory of MOH/GHS - Ghana National Reproductive Health Policy and Standards, 2003; Bongaarts et al., 2012).

Leadership and Governance in Family Planning

Leadership in family planning involves bringing together multiple partners that work in family planning toward a common goal of improving family planning services and increasing contraceptive prevalence; whilst Governance refers to the establishment of systems and procedures used to ensure the functioning of key organizations working in family planning. Governance and leadership in family planning requires accountability, transparency, integrity, and commitment to the common goal (National Strategic Plan for Family Planning in the DRC, 2014-2020).

In Ghana leadership and commitment to family planning has been shown by the government's development of family planning policies and legislation consistent with international conventions, by approval of the Roadmap for Repositioning Family Planning and other reproductive health policy documents and by effective collaboration with the private sector and development partners.

Although, the Ministry of Health (MOH) and Ghana Health Service (GHS) play leadership roles in strengthening the family planning programme and ensuring contraceptive security, there exists a national coordination mechanism that ensures the development and implementation of a coordinated strategy from which specific activities of various stakeholders derive. The membership of the coordination mechanism includes government, development partners, non-governmental organizations /civil society organizations and the private sector (Meeting the Commodity Challenge: Ghana's National Reproductive Health Commodity Security Strategy 2011–2016).

In addition to the coordination mechanism, there is an efficient logistics system which links the lower level facilities to the central level. The logistic system includes a logistic forecasting and quantification, supply and procurement, inventory control, storage and warehousing, transport and distribution and logistics management information systems. Challenges with the leadership and governance system are the poor reporting system that fails to provide the required information for coordination, lack of follow-up on agreed strategies and plans and irregular coordination meetings (Meeting the Commodity Challenge: Ghana's National Reproductive Health Commodity Security Strategy 2011–2016, n. d.)

Postpartum Family Planning

Conceptually, the postpartum period may be divided into the following (Figure 2): (1) post-placental- within 10 minutes after delivery of the placenta, (2) Immediate postpartum - up to one week after delivery, (3) later postpartum - one week up to six weeks after delivery and (4) the Extended postpartum - six weeks up to one year after delivery (McKaig & Chase, 2007). Figure 2 depicts appropriate modern contraceptive methods that could be employed within the postpartum period for all women, breastfeeding women and non-breastfeeding women.

Postpartum family planning programme was initiated and launched by the Population Council in 1966. Its success led the Population Council to propose a global plan in 1971, with the dual aim of improving maternal and child health and providing better access to modern contraception (Taylor Jr & Berelson, 1971). Although the aims of this plan were not realized initially, postpartum family planning was considered to be an opportunity to offer family planning along the continuum of maternal health care: from pre-pregnancy and antenatal care, through the extended postpartum period (Ringheim, 2011). Postpartum family management information systems. Challenges with the leadership and governance system are the poor reporting system that fails to provide the required information for coordination, lack of follow-up on agreed strategies and plans and irregular coordination meetings (Meeting the Commodity Challenge: Ghana's National Reproductive Health Commodity Security Strategy 2011–2016, n. d.)

Postpartum Family Planning

Conceptually, the postpartum period may be divided into the following (Figure 2): (1) post-placental- within 10 minutes after delivery of the placenta, (2) Immediate postpartum - up to one week after delivery, (3) later postpartum - one week up to six weeks after delivery and (4) the Extended postpartum - six weeks up to one year after delivery (McKaig & Chase, 2007). Figure 2 depicts appropriate modern contraceptive methods that could be employed within the postpartum period for all women, breastfeeding women and non-breastfeeding women.

Postpartum family planning programme was initiated and launched by the Population Council in 1966. Its success led the Population Council to propose a global plan in 1971, with the dual aim of improving maternal and child health and providing better access to modern contraception (Taylor Jr & Berelson, 1971). Although the aims of this plan were not realized initially, postpartum family planning was considered to be an opportunity to offer family planning along the continuum of maternal health care: from pre-pregnancy and antenatal care, through the extended postpartum period (Ringheim, 2011). Postpartum family planning according to the WHO is, the prevention of unintended and closely spaced pregnancies through the first 12 months following childbirth. It is the time when a subsequent pregnancy holds the greatest risk for mother and baby and also when there is the greatest number of contacts with health care services (WHO, 2013)



Figure 2: Family Planning Methods Appropriate for Women in the Postpartum Period

Source: World Health Organization (2013)

planning according to the WHO is, the prevention of unintended and closely spaced pregnancies through the first 12 months following childbirth. It is the time when a subsequent pregnancy holds the greatest risk for mother and baby and also when there is the greatest number of contacts with health care services (WHO, 2013)



Figure 2: Family Planning Methods Appropriate for Women in the Postpartum Period

Source: World Health Organization (2013)

The 2006 World Health Report proposed a 2-3 year birth interval following birth, and a six month interval following miscarriage or abortion, because they ensure good maternal and child health outcomes (WHO, 2006). Although most women would want to wait for more than two years before having another baby, there is still a high level of unmet need for family planning during the first year following birth when women are especially vulnerable to unintended pregnancy (Duong, Lee, & Binns, 2005). According to an analysis of DHS data from 27 countries, 65 percent of women who are 0–12 months postpartum, had an unmet need for family planning (Ross & Winfrey, 2001). The postpartum period, especially the immediate postpartum period, is a time during which couples generally have multiple encounters with the health-care system. Providing contraception during this time is cost-effective and efficient as a mechanism of family planning service provision (Singh et al., 2009; Warren, Mwangi, Oweya, Kamunya, & Koskei, 2010; World Health Organization, 2013).

Statement of Problem

United Nation's Millennium Development Goal five (MDG 5), aims at reducing maternal mortality by three quarters, between 1990 and 2015. An important intervention towards achieving this target is the promotion of modern FP among women (Sachs & McArthur, 2005; Cates et al., 2010). Within the first twelve (12) months postpartum, closely spaced pregnancies are the riskiest for mother and baby resulting in a 7.5-fold increased risk for induced abortion, a 1.6fold increased risk of stillbirth, preterm births, low birth weight and small for

.

gestational age (DaVanzo, Hale, Razzaque, & Rahman, 2007). Evidence exists that if couples can space their pregnancies by at least two years through the use of postpartum family planning, up to 35 percent of maternal deaths and up to 13 percent of child mortalities could be averted (Cleland et al., 2006; Rutstein, 2008; Stover & Ross, 2010). Additionally, 25 percent of under – five mortalities could be averted if birth intervals were at least three years (Rutstein, 2008). Uptake of modern family planning methods remains low in Sub-Saharan Africa (SSA) and this is associated with a high incidence of unintended pregnancies, unsafe abortions, unplanned deliveries and maternal mortalities (Crossette, 2005; Cates et al., 2010).

In Ghana, unintended childbearing is estimated at about 0.7 births per woman (Johnson & Madise, 2011). According to the Ghana Demographic and Health Survey (GDHS), fertility rate, contraceptive prevalence rate (CPR) and unmet need in Ghana are 4.2, 27% and 30% respectively, with spatial and sociodemographic disparities. Women in rural areas have an average of two children more than those in urban areas (5.2 versus 3.4) and use modern family planning to the same extent as their urban counterparts (27% versus 26%%) [GSS & Macro, 2014]. Women with at least some secondary education are nearly t twice as likely to use contraception as women with no education; fewer women in the lowest wealth quintile are using a contraceptive method, compared with their counterparts in the highest wealth quintile (22% versus 29%) [GSS, Macro, & ICF, 2014].

At current rates, the targets set in Ghana's population policy are unlikely to be met by 2020. These targets are to reduce total fertility rate from 5.0 to 3.0, and to improve contraceptive prevalence rate from 15 to 50 percent between 2000 and 2020. The postpartum period remains a very important 'window of opportunity" that could be utilized to improve contraceptive uptake. This period is critical because postpartum women have a high need for contraception and have multiple contacts with the health facility either for postnatal or child immunization visits. Although, the majority of postpartum women indicate a desire to use contraceptives, family planning methods are often not offered to, or taken up by women after delivery or in the first year postpartum (Ross & Winfrey, 2001; Adeyemi, Ijadunola, Orji, Kuti, & Alabi, 2005; Adegbola & Okunowo, 2009). Furthermore, lack of knowledge about the return of fertility in the postpartum period increase women's susceptibility to unintended pregnancy as many of them incorrectly believe that fertility returns with resumption of menses (Adeyemi et al., 2005; Borda, Winfrey, & McKaig, 2011; Ndugwa, Cleland, Madise, Fotso, & Zulu, 2011). Many women also resume sexual activity prior to the resumption of menses, putting themselves at risk of a mistimed or unwanted subsequent pregnancy in this period. Seizing the opportunity to offer family planning services to women in the postpartum period may result in a steeper increase in the uptake of family planning methods and a reduction in unintended pregnancies (Speizer, Fotso, Okigbo, Faye, & Seck, 2013).

Very little is known about how pregnant women in Sub-saharan Africa (SSA), including Ghana arrive at their PPFP decisions. This information is nevertheless critical to the design of strategies to increase the uptake of PPFP (Adegbola & Okunowo, 2009; Warren et al., 2010).

Objectives of the Study

The main objective was to assess the factors that influence pregnant women's intention to use and use postpartum family planning.

The specific objectives were to:

- 1. Ascertain the level and determinants of unintended pregnancies among pregnant women
- 2. Examine the influence of socio-demographic characteristics of pregnant women on intention to use family planning postpartum
- 3. Assess the influence of partner socio-demographic characteristics on women's intention to use postpartum family planning
- 4. Analyse the influence of sex composition of children (in the context of sex preference) on intention to use and use of postpartum family planning
- 5. Determine the relationship between intention to use postpartum family planning and actual use
- Estimate the timing of postpartum family planning use in relation to postpartum susceptibility to pregnancy (influenced by breastfeeding, postpartum amenorrhoea and postpartum sexual abstinence)

Hypotheses of the Study

This study tested three main hypotheses, including:

- H₀: There is no significant relationship between socio-demographic background characteristics of respondents and their intention to use postpartum family planning;
- H₀: There is no significant relationship between sex composition of children ever born by the women and unintended pregnancy, intention to use and use of postpartum family planning; and
- 3. H₀: There is no significant relationship between intention to use and actual use of postpartum family planning among women.

Rationale of the Study

The Programme of Action of International Conference on Population and Development (ICPD/POA) in Cairo in 1994 established a clear link between population dynamics and sustained economic development. Since then, the Ghana National Population Policy (developed in 1969 and revised in 1994) has sought to ensure acceleration of socio-economic growth by 2020. Targets set included reduction of Total Fertility Rate (TFR) from 5.5 to 3.0; increase of contraceptive prevalence rate to 50 percent for modern methods and reduction of population growth rate from 3.0 to 1.5 percent by 2020. To ensure the attainment of these targets, the government of Ghana has worked to build a positive policy and programmatic environment by the development of subsequent policy documents such as the Ghana National Reproductive Health Policy and Standards 1994 (revised in 2003), a Roadmap for Repositioning Family Planning, Family Planning Protocols and the institution of a strong family planning programme. Some progress has been made: Total Fertility Rate is 4.2, Contraceptive Prevalence Rate is 27% (GSS, Macro & ICF 2014), and population growth rate is 2.5(Ghana Statistical Service, 2012), whilst unintended births improved from 42 to 37 percent between 1999 and 2008 (GSS & Macro, 2009). Ghana has since moved from a low income to a lower middle income country. At current rates, the targets set in the population policy may not be achieved despite the progress made. Ghana could have attained a higher socio-economic growth if the targets set out in the population policy had been achieved. The postpartum period offers good opportunity for family planning but has not been utilized. The reasons may reside in individual, partner, programme, institutional or socio-cultural factors and need to be ascertained.

Although fertility declines have been observed in the past three decades in some countries in the sub-region, certain socio-cultural factors and household, kinship and community institutions that favour childbearing remain highly influential (Makinwa-Adebusoye, 2007). Sex composition in the context of "son preference" is only one dimension of these socio-cultural factors which may influence reproductive health outcomes. The influence of sex composition of children on various reproductive health outcomes has been extensively studied in South and South East Asia, where an estimated 30-70 million women are believed to be "missing" (Qian, 2008), and sex ratios at birth as high as 130 males to 100 females have been observed (United Nations, 2011). Among the reasons identified to account for the situation are, the patriarchal nature of the society, the attitude that sons are more important and valuable than daughters in carrying out important religious roles, that sons have the right to inherit land to support aged parents, and perpetuate the family name (Calhoun, Nanda, Speizer, & Jain, 2013).

Of the twenty-eight (28) sub-Saharan African countries reviewed, balance was found to be the most popular preference in 24 countries in sub-Saharan Africa (Fuse, 2010). The study further revealed that although son preference is observed in every sub-region within sub-Saharan Africa, it appeared to be particularly prevalent in West Africa, especially Mali, Senegal and Burkina Faso. Daughter preferences have also been observed in some West African countries including Ghana, Liberia and Sierra Leone. A study on the dynamics of reproductive health behaviours in rural coastal communities of southern Ghana by Akinyoade (2007) revealed that individuals in southern coastal communities preferred sons to daughters for propagation of their business.

In spite of these studies, there appeared to be paucity of research on sex composition or sex preferences and reproductive health behaviours in Africa which take into account the various social arrangements. One of the identified factors influencing son preference is the patriarchal nature of the society. However, there are no studies on sex preference, for instance, in the matrilineal societies. One of the questions then is whether there will be preference for one sex or the other in matrilineal areas.

This thesis addresses some of these issues by examining the postpartum family planning intentions of pregnant women and the determinants associated with translation or otherwise of family planning intentions during pregnancy into reality after delivery. The study will then provide evidence of some of the strategies that family planning programme managers can use to target women and match interventions to actual performance to promote the health of both mother and child.

Organization of the Thesis

The study is organized into nine chapters. Chapter One captures the introduction to the study and sets the background for the study. It discusses some historical perspectives of family planning; family planning as a human right; cost benefit-analysis of family planning; family planning and the health service; and postpartum family planning. It also discusses the problem statement, purpose, objectives and rationale of the study.

Chapter Two puts the study within the context of empirical body of knowledge on intention to use and use of family planning and discusses various theories and concepts related to behaviour change and fertility.

Chapter Three discusses the methodological approach to the study. It also discusses the limitations and challenges of the study and how they were mitigated. Chapter Four presents the background characteristics of the respondents in general and also compared the characteristics of those who intended to be followed up and those who were actually followed up. Determinants of unintended pregnancies, an outcome of poor family planning use are also discussed in this chapter, whilst factors influencing the intention of women to use postpartum family planning are discussed in Chapter Five.

Postpartum family planning use, the relationship between intention to use and actual use of postpartum family planning, the most methods used and the determinants of actual use are assessed and discussed in Chapter Six. Chapter Seven describes the sex composition of children ever-born and assesses the possible existence of sex preference within the matrilineal inheritance system of the Mfantseman Municipality and further assessed the influence of sex composition on reproductive health outcomes (unintended pregnancy, intention to use and actual use of postpartum family planning). In addition, this chapter estimates the average timing of postpartum contraceptive use and discusses its relationship with breastfeeding, postpartum amenorrhoea and postpartum sexual abstinence among the postpartum women.

Chapter eight concludes the study with summary of main findings, conclusions, and contributions to knowledge, policy implications, recommendations and future research perspectives.

37

CHAPTER TWO

CONCEPTUAL, THEORETICAL AND EMPERICAL ISSUES ON INTENTION TO USE AND USE OF POSTPARTUM FAMILY PLANNING

Introduction

There is growing consensus that behavioural change is likely to occur if people are motivated to undertake the change and if environmental barriers are reduced or removed (Fishbein & Cappella, 2006). Empirical research over the past three decades has led to the recognition that specific behaviours can be predicted with considerable accuracy by assessing intentions to engage in the behaviours under consideration (Fishbein, 2008). The presumption that intentions can be reasonably used to predict behaviours is based on the view that individuals translate intentions to actual behaviour (Agha, 2010; Callahan & Becker, 2014). This chapter discusses unintended pregnancy, a core consideration in intentions to use and use of contraceptives; examines existing body of knowledge on intention to use and use of contraceptives and the extent to which intentions are translated into actual behaviour. Theories and models of behaviour change and fertility that will guide the study are also discussed.

Unintended Pregnancy

Prevention of unintended pregnancy is a core consideration in individuals' and couples' intention to use and use of contraceptives. It refers to pregnancies that are not wanted or those that are mistimed at the time of conception (Centers for Disease Control, 2010). Out of the 208 million pregnancies estimated worldwide, in 2008, 41percent were unintended (Singh, Sedgh, & Hussain, 2010). Rates of unintended pregnancies, though declining world-wide, are still high. Rates declined by 20 percent from 71 to 57 per 1000 from 1995 to 2008 among women aged 15 to 44 years in Low- and Middle -Income Countries (LMICs) (Singh et al., 2010) . In 2008, 75 million women in LMICs reported that their pregnancies were unintended, with 23 percent of these pregnancies occurring in SSA (Singh, Wulf, et al., 2009). Unintended pregnancies were estimated in 2008 at 49 per 1000 pregnancies in Asia, 72 per 1000 in Latin America and the Caribbean and for women aged 15 - 44 years in Africa, 86 per 1000 (Singh et al., 2009); that of Africa was rated as the highest and stated that in Ghana, 37 percent of all births are unintended (Sedgh, 2010).

These pregnancies may carry serious consequences for women and their families, including possible unsafe abortion, delayed prenatal care, poor maternal mental health, reduced mother/child relationship quality, poor developmental outcomes for children, physical abuse and violence of women, increased risk of low birth weight of babies as well as increased maternal morbidity and mortality (Santelli, 2003; Singh et al., 2010). Available data suggests that induced abortion and related complications are the most common outcomes of unintended pregnancies.

The concept of unintended pregnancy has been found to have many facets (Santelli et al., 2003): First, planning or intending to become pregnant may be

distinct from wanting to be pregnant; second, the concept of planning a pregnancy may not be meaningful because of contraceptive failure and a woman's happiness or unhappiness after discovering she is pregnant; and third, having mixed feelings or contradictory ideas about avoiding pregnancy may be expressed in imperfect contraceptive use (Trussell, Vaughan, & Stanford 1999).

A major problem with surveys on unintended pregnancy is that most measure women's intentions after a birth has occurred. This situation can be influenced by the presence of the baby so that an otherwise unintended pregnancy may become more positive over time (Joyce, Kaestner, & Korenman, 2000). Another problem with measurement of unintended pregnancy is the inherent heterogeneity. According to Luker (1999), unwanted and mistimed pregnancies commonly represent different life choice considerations: unwantedness reflects the intentions or desires of a woman and her partner after the couple has had all the children they want. In contrast, mistimed pregnancies can occur throughout the reproductive years but are most common among adolescent and young adult women (Pulley, Klerman, Tang, & Baker, 2002)

Unintended pregnancy may be a reflection of ambivalence about sexuality and may be expressed through inconsistent use of contraceptives. Emotional and psychological factors influence, to a large extent the complex fertility decisions women make and their pregnancy intentions; furthermore, current measures of pregnancy and fertility intentions are not designed for individual level assessment but for population level assessment (Santelli et al., 2003). In view of these factors, some researchers have suggested that intended and unintended pregnancies should not be seen as dichotomous but be considered as two ends of a spectrum (Bachrach & Newcomer, 1999). This spectrum should include the desire for a baby or affective dimension, which is related to the community, partner and individual values about childbearing; and a pregnancy dimension which is concerned with preparation for pregnancy, life goals and education (Stanford, Hobbs, Jameson, DeWitt, & Fischer, 2000). A woman's feelings about a particular pregnancy and decision about abortion may be shaped by changes in the relationship with her partner, pressure from family and friends, antenatal diagnostic tests and medical, psychiatric and psychological conditions (Santelli et al., 2003).

Anthropological studies of reproduction have noted a complex set of relationships within which pregnancies occur, thereby raising the question "intended by whom" (Santelli et al., 2003). Answers to this question will require a critical analysis of how gender inequality and culturally constructed ideals about relationships and sexuality shape women's interactions with their partners, as well as on women's interactions with their lineage and kinship groups, peers and healthcare providers (Santelli et al., 2003). In this regard, anthropological literature suggests that public health research on pregnancy intendedness should take into consideration contextual factors or socio-cultural factors such as gender inequalities and structural factors within the health delivery system.

Intention to Use and Use of Family Planning

Studies in family planning on intention to use, and its further translation into actual use are very few. Adegbola and Okunowo (2009) in their study on intended postpartum contraceptive use amongst 423 pregnant and puerperal women in a teaching hospital in Nigeria reported that fifty-four percent (54%) of the respondents intended to use contraceptives after delivery when the prevalence of previous contraceptive use was 35.5%. In a longitudinal study conducted in rural Bangladesh (Callahan & Becker, 2014), it was reported that women (pregnant and non-pregnant) who were not using a method of contraception but said they intended to use a method in the future were more likely to go on to use a method than women with no intention to use. It was further reported that women who intended to use a specific method were even more likely to begin use than women who were unsure of what method they wanted to use (Callahan & Becker, 2014).

Keogh (2012) and colleagues, in their study, followed 5,284 pregnant women throughout the extended postpartum period in Northern Tanzania. The study outcome revealed that demand for delaying births was matched by intention to use family planning (over 77 percent) both at baseline and follow-up. They further reported that intentions often did not translate into actual contraceptive use in the postpartum period; while 72 percent of women intended at baseline to start family planning in the first year postpartum only 40 percent had actually managed to (or still intended to) at follow-up.

Factors Influencing Intention to Use Contraceptives

Some studies have observed strong association between some sociodemographic, socio-cultural and socio-economic characteristics of women and significant others, and intention to use and/or use of contraceptives while others have not. For instance, age (Adegbola & Okunowo, 2009) and high parity (Adegbola & Okunowo, 2009; Kariuki, 2011) have been found to predict intention to use postpartum contraceptives. Level of education (Adegbola & Okunowo, 2009; Kariuki, 2011; Di Giacomo, Sbarlati, Bagnasco, & Sasso, 2013) and family planning counseling by providers (doctors and nurses) increased the intention to use contraceptives, including postpartum contraceptives (Adegbola & Okunowo, 2009). According to Kariuki (2011), being married is a significant predictor of intention to use postpartum contraceptives among first time mothers.

Agha (2010) observed that, the perception that one's in-laws support family planning, belief in the importance of spacing children, availability of method of choice and facilities with competent staff as well as spousal discussion were drivers of intentions to use contraceptive methods. The obstacles to a woman's intention to use contraceptive methods were her belief that family planning decisions were made by her husband, fertility was determined by God's will and fears that family planning would harm a woman's womb (Agha, 2010)

Factors Influencing Actual Use of Contraceptives

Factors that influence actual use of contraceptives are also generally sociodemographic and socio-economic. Older age (Jabeen, Gul, Wazir, & Javed, 2012; Sharma, Mohan, Das, & Awasthi, 2012), higher educational status (Arbab, Bener, & Abdulmalik, 2011; Yihunie, Tamene, Benedict, & Deribe, 2011; Jabeen et al., 2012; Buyinza & Hisali, 2013), high parity (Jabeen et al., 2012; Sharma et al., 2012; Yihunie et al., 2013), higher household income and highest wealth quintile (Arbab et al., 2011; Yihunie et al., 2013) are associated with actual use of family planning.

Yihunie and colleagues (2013) analysed 10,204 women from the 2011 demographic and health survey data of Ethiopia and reported that being employed, being in a monogamous relationship, attending community conversation and being visited by a health worker at home increased the likelihood of using modern contraception. Some factors that were found to negatively influence contraceptive use included living in rural areas, being in polygamous relationship, being Muslim and witnessing one's own child's death (Yihunie et al., 2011).

Theories and Models

Most interventions to change and improve health-related behaviour are guided by formally stated theories, or by less formally defined ideas about behaviour that are derived from people's experiences, opinions and intuitions (Glanz, Rimer, & Viswanath, 2008). The design of interventions that yield desirable changes can best be done with an understanding of theories of behaviour change and an ability to use them skillfully in research and practice (Grol, Bosch, Hulscher, Eccles, & Wensing, 2007). According to McGuire (1983), the adequacy of a theory is often assessed using three criteria: (1) its logic, or internal consistency in not yielding mutually contradictory derivations, (2) the extent to which it is *parsimonious*, or broadly relevant while using a manageable number of concepts, and (3) its *plausibility* in fitting with prevailing theories in the field.

44

Not many studies that use behaviour change theories, measure actual behaviour change, or use rigorous methodologies and designs; neither do they compare multiple behaviour change theories. According to Noar and Zimmerman (2005), out of nearly three thousand (3000) studies reporting the use of behaviour change theories, only nineteen (19) of them compared different behaviour change models. Of these, only three (3), according to Weinstein and Rothman (2005), used rigorous designs (longitudinal as opposed to cross sectional) and behavioural outcomes.

Behaviours are influenced by many factors. However, McLeroy, Bibeau, Steckler and Glanz (1988) identified five factors (affected by existing social environment) that influence health-related behaviours, namely: Intrapersonal or individual factors; interpersonal factors; institutional or organizational factors; community factors; and policy level factors.

Individual and interpersonal health-behaviour theories are generally classified under Cognitive – Behavioural Theories, which have behaviour and knowledge as major concepts influencing them, despite the presence of other factors in the social environment. Although, group, community and national level behaviour change interventions are important, it is extremely useful to target individuals who make up the groups and communities, in order for a particular behaviour change to be effective. Individual level targeting is important, especially for sensitive and sometimes controversial issue such as use of family planning.

45

Several philosophical ideas, theories and models have been proffered to explain behaviour modification. Among them are the Health Belief Model; Stages of Change (Transtheoretical), Theory of Planned Behaviour (TPB); Social Cognitive Theory; Diffusion of Innovations Theory (DOI), Steps to Behaviour Change Model (STBCM); Postpartum Behaviours Model; and Davis and Blake Model of fertility. The core issues of these models are examined to inform the study.

Health Belief Model

In order to understand why individuals fail to use behaviours such as disease prevention strategies or undergo screening tests for early detection of disease, Social Scientists in the United States of America (USA) developed the Health Belief Model in the 1950s (Hochbaum, 1958). Susceptibility, Seriousness, Benefits and Barriers, Cues to action and Self efficacy are primary constructs that predict an individual's ability to prevent and control disease conditions in this model. Individuals are likely to take mitigating actions to reduce their risks if their susceptibility leads them to believe that a given condition may have serious consequences, a certain course of action will reduce their susceptibility and possible severity of the condition, and the anticipated benefits of taking that action far outweigh the barriers. Some other demographic and socio-psychological variables influence perceptions and may indirectly affect behaviours, working through the constructs (Figure 3). The theory's intuitive logic and clearly stated central tenets make it appealing (Brewer & Rimer, 2008). Other strengths are its ability to explain health related behaviours and provide useful theoretical basis for analyzing cognitive determinants of behaviour (Orji, Vassileva, & Mandryk, 2012). Furthermore, it can help to develop change strategies and develop messages that are likely to persuade individuals to make healthy decisions. Finally, it is a proven way of helping to identify health behaviour correlates (Brewer & Rimer, 2008).

These strengths notwithstanding, the HBM has some weaknesses. First, it does not account for a person's attitudes, beliefs, or other individual determinants that dictate a person's acceptance of health behaviour. Second, behaviours that are taken for non-health-related issues and habitual behaviours are also not accounted for by the theory(Champion & Skinner , 2008). Third, like all the other theories, it does not integrate socio-economic and environmental variables into the framework. Fourth, its assumptions that cues to action are widely prevalent and that individuals have equal amounts of information on health conditions are flawed. Finally, a major weakness worth noting is also that the relationships between the various constructs are not clearly defined and make measurements quite difficult (Armitage & Conner, 2000; Orji et al., 2012).

Overall, the Health belief model has stood the test of time despite its limitations. It is a cognitive model that is widely used by researchers in the area of behaviour change. The cognitive sequences are very ordered and work better; and depending on the setting and the type of behaviour change applied to, the weight of the various constructs may vary.

Individual Beliefs



Figure 3: Health Belief Model

Source: Rosenstock et al. (1994)

The theory is designed to predict behaviour change, but it is better at describing present or current behaviour status than predicting change. The combination of susceptibility and severity makes it stronger at predicting behaviour than models that use threat alone.

Social Cognitive Theory

This theory developed by Albert Bandura explains human behaviour in terms of a three-way, dynamic, reciprocal theory in which personal factors, environmental influences, and behaviour interact continually (Bandura, 1997). The key constructs of this theory can be grouped into five categories namely: psychological determinants of behaviour (outcome expectations and self-efficacy); observational learning; environmental determinants of behaviour (environmental determinism, incentive motivation and facilitation); self-regulation, and moral disengagement (McAlister, Perry, & Parcel, 2008).

The psychological determinants include outcome expectations and selfefficacy. Outcome expectations are expectations about how different people will evaluate our behaviour and our willingness to be guided by their evaluation(McAlister et al., 2008). Inherent in this is the concept of self-evaluative outcome expectation which is also very critical. It functions like a social outcome, however, it can be governed partly by people's anticipation of how they will feel about themselves if they do or do not perform certain behaviours. Evidence shows that expectations about self-evaluative outcomes can be more powerful than expectations about social and material outcomes for some individuals (McAlister
et al., 2008). The creation of this category of outcome expectation explains how individuals can resist social pressures to meet their own standards of approvable conduct.

Self-efficacy, for which the social cognitive theory is most widely known, consists of a person's beliefs and confidence about her ability to take actions that affect her life and the quality of actions taken (Bandura, 1997). Four ways in which self-efficacy can be developed include mastery experience, social modeling, improving physical and emotional states, and verbal persuasion (Bandura, 2004). Of all the four ways, mastery experience is considered the strongest influence on self-efficacy. It involves creating an enabling environment for a person to succeed in attainable but increasingly challenging performances of desired behaviours. This requires extensive practice.

The second source of influence on self-efficacy is observational learning or modeling whereby people judge their capabilities in relation to others whom they regard as similar to themselves. For example, a person who intends to overcome a certain phobia, benefits from seeing others who are in similar situation and are making progress with overcoming their phobias. This helps individuals to overcome the difficulties associated with the process by learning the experiences of the models. The third source which is verbal persuasion, involves encouraging individuals to boost their confidence enough to induce the initial efforts at behavior change. Finally, improvement in physical and emotional states goes a long way to boost self-efficacy. Acquiring skills to reduce uncomfortable physiological reactions such as stress and depression by relabeling these as "excitement" helps individuals overcome these.

The social cognitive theory has a reciprocally deterministic view point and posits that no amount of observational learning will lead to behaviour change unless the observers' environments support the new behaviours (Bandura & Bryant, 2002). Incentive motivation and facilitation are two basic forms of environmental change that influence behaviour. Incentive motivation involves use of rewards or punishment which increase or decrease respectively, the probability that behaviours will be enhanced. Similarly, facilitation (more empowering than motivation) is the provision of new structures or resources that enable behaviours or make them easier to perform (Bandura, 1998).

Self-regulation is the extent that one accurately reflects on his or her progress toward a learning goal, and appropriately adjusts his or her actions to maximize performance. Three characteristics of self-regulation include selfobservation (monitoring one's activities), seen as the most important of these processes; self-judgment (self-evaluation of one's performance) and self-reactions (reactions to performance outcomes) (Zimmermann, 2000).

Weaknesses of the social cognitive theory are that: it is so broad and lacks any unifying principle or structure and is difficult to implement in its entirety; it does not take into consideration biological differences and the influence of hormonal responses on one's behavior. In addition, it ignores genetic factors that could lead to disparities between people's cognitive abilities and behavior; it does not consider maturation and lifespan behavior changes in that as people move

51

physiological reactions such as stress and depression by relabeling these as "excitement" helps individuals overcome these.

The social cognitive theory has a reciprocally deterministic view point and posits that no amount of observational learning will lead to behaviour change unless the observers' environments support the new behaviours (Bandura & Bryant, 2002). Incentive motivation and facilitation are two basic forms of environmental change that influence behaviour. Incentive motivation involves use of rewards or punishment which increase or decrease respectively, the probability that behaviours will be enhanced. Similarly, facilitation (more empowering than motivation) is the provision of new structures or resources that enable behaviours or make them easier to perform (Bandura, 1998).

Self-regulation is the extent that one accurately reflects on his or her progress toward a learning goal, and appropriately adjusts his or her actions to maximize performance. Three characteristics of self-regulation include selfobservation (monitoring one's activities), seen as the most important of these processes; self-judgment (self-evaluation of one's performance) and self-reactions (reactions to performance outcomes) (Zimmermann, 2000).

Weaknesses of the social cognitive theory are that: it is so broad and lacks any unifying principle or structure and is difficult to implement in its entirety; it does not take into consideration biological differences and the influence of hormonal responses on one's behavior. In addition, it ignores genetic factors that could lead to disparities between people's cognitive abilities and behavior; it does not consider maturation and lifespan behavior changes in that as people move

51

through life, their behavioral patterns can change drastically with little change in their environment. The strengths are that it: is accurate and easy to understand, optimistic in a good way, offers opportunity to explain how behaviour is learned, provides a way to integrate social and cognitive theories, is able to explain a large number of behaviours, can handle inconsistencies in behaviour and can allow and account for cognitive processes.

Diffusion of Innovations Theory

The concept of diffusion had been studied extensively prior to the publication of Everett Roger's theory of Diffusion of Innovations (DOI) (Pemberton, 1936). The DOI is a broad socio-psychological theory that seeks to describe the patterns of use, explains and helps to predict whether and how innovations, strategies or ideas are used (Rogers, 2008).

According to Rogers (2008), diffusion is the process by which ideas, strategies or innovations spread to members of a social system over time, through certain communication channels. Thus, the author deduced that five elements which influence the spread of innovations or ideas are: the innovations or ideas themselves; communication channels; time; existing social systems and the messenger. In addition to these, various diffusion stages exist. The first stage is innovation development, followed by dissemination, adoption, implementation, maintenance, sustainability, and institutionalization (Orlandi, Landers, Weston, & Haley, 1990; Oldenburg & Glanz, 2008).The decision to adopt an innovation occurs through knowledge, persuasion, decision, implementation and confirmation (Cain & Mittman, 2002; Rogers, 2008). The end result of diffusion is that individuals within a social set up use new behaviours and ideas. Perception of an idea or behaviour as new or innovative forms the main basis or backbone for use.

Within a social system, individuals do not use an idea or behaviour at the same time. Some use earlier than others, whilst some may be for or against the innovation. It is therefore important to stratify the population of interest, based on their different characteristics, in other to design different strategies to target them. Five (5) user audiences exist according to Rogers (2003) (Figure 4): Innovators; Early adopters; Early Majority; Late Majority; Laggards.

Innovators are usually people who are not afraid to use new ideas or behaviours, are innovative, want to be first to use them and are risk takers. Generally, little to nothing needs to be done to convince such individuals. Early adopters are usually aware of the need to change and are not afraid to use new ideas. They have characteristics of opinion leaders and enjoy leadership roles. Such individuals only need information on how to implement or use the idea or behaviour. Late users are skeptics and need to be convinced beyond reasonable doubt that an idea or a change of behavior pays off. They will usually try new ideas or behaviours if they are convinced that many people have tried and tested them. Strategies that appeal to such people are success stories and statistics about numbers of individuals who have used ideas and behaviours successfully. Laggards are the most difficult to convince, are conservative and bound by entrenched traditions and beliefs. They may have little or no formal education, likely to be rural dwellers and may be in the lower wealth quintiles compared to innovators and early users. These people will require success stories, targeting by peer groups and fear to eliminate fear and provide reassurance.





At play, to different extents, at the four user categories are: the idea's or behaviour's relative advantage over the previous or existing one; its compatibility with the users' needs, experiences and values; how complex or difficult the behavior or idea is to understand; the ease with which the innovation could be tried before being used and the extent to which it provides results which are tangible (observability) (Oldenburg & Glanz, 2008).

The importance of the theory are first, it strengthens the adoption-decision process by introducing innovation, communication channels and the nature of the society to which the innovation is targeted into the model. Second, it adds the extent of change agents' promotion efforts to the contextual factor of the adoptiondecision model. Third, it confirms adoption is a mental process in which the individual's attitude influences decision to adopt or reject the innovation. Fourth, the concept of the adopter's perception about innovation strengthens the adoptiondecision model through the five characteristics of innovation (Botha & Atkins; 2005).

The theory is however limited by the fact that, it assumes a hundred percent adoption of innovation and does not allow for rejection (Waterman, 2004). The theory also assumes the characteristics of the innovation do not change over time (Wolfe, 1994). According to Kole (2002) the model has pro-innovation and individual biases in that it assumes all individuals should adopt innovation and do it more quickly and that it only focuses on the individual adopter and ignores the individual's resources or the social support to use the behaviour.

For behaviour such as adoption of postpartum family planning, the diffusion of innovation theory will help us understand the characteristics of the women being studied, their knowledge base and the influence it may have on their subsequent use of postpartum family planning, the social system (culture, beliefs and practices) in which they live and which may influence the use of the experience, timing of the use and factors that influence the rate of use.

Transtheoretical (Stages of Change) Model

This was developed by Proschaska and DiClemente in 1983 whilst studying experiences of smokers who quit on their own (Prochaska, DiClemente, & Norcross, 1992). It is an Intentional Change Model which focuses on the decision –making of an individual. The model is based on the assumption that change in behaviour occurs through a cyclical process (Figure 5) and that behaviours are not changed quickly or decisively. Different theories where effective, can be applied to different stages of the cycle.

According to them, six stages are proposed namely (Figure 5): Precontemplation; Contemplation, Preparation, Action, Maintenance and Termination. The last stage was not part of the original model and may not apply to health behaviour change. At the Precontemplation stage, people lack insight into their negative behaviours and do not intend to change in the immediate future. They tend to focus more on the negative rather than the positive consequences of behaviour changes that are proposed to them.



Figure 5: Stages of Change Model

Source: Proschaka and Di Clemente (2008)

In the Contemplation stage, individuals have some insight into their problems but still have contradictory feelings towards changing behaviour. They now have intentions to change their behaviours in the immediate future (next six months). Equal emphases are placed on both positive and negative outcomes of using new behaviours and experiences.

At the Determination stage, people begin to take some minor steps towards changing their behaviours because they are now convinced of the positive outcomes of it. They may be likely to take major steps in the following month. Some individuals have a change of mind as they progress along this stage, but may re-enter to complete and enter the action stage. In the Action stage individuals have actually changed their behaviours in the last six months and may be progressing towards maintaining their behaviours. Some individuals are unable to sustain their behaviours and may relapse. Some may remain in the relapse stage and fail to progress, whilst some others may take up the new behaviours again and progress to the Maintenance stage. In this stage, individuals make a strong effort not to relapse. They usually have maintained the new behaviours for over six months and may be progressing. The Termination stage is difficult to attain in health behaviour-change programmes and is often not considered. Here individuals are convinced strongly that they will never return to their old ways.

The strengths of the model are that it helps effective targeting of interventions at individuals in different stages, to ensure positive behaviour-change outcomes and treats behaviour change as a dynamic rather than an "all or nothing" phenomenon (Marshall & Biddle, 2001). The limitations of applying this model to

this study are that, the socio-cultural and socio-economic environment in which behaviour-change occurs are not considered; furthermore, the theory assumes the following: that individuals planning to use new behaviours do so logically and coherently, whilst this is not observed in practice; assigning individuals to particular stages may be arbitrary because no set criteria exist and no clear durations exist for how long individuals spend in each stage (BUMC, 2013).

The application of this model to postpartum family planning may be practical, in the sense that there may be some women in the pre-contemplation stage who, for fear of side effects of contraceptives, may have no intentions of using postpartum family planning whilst others may be in the contemplation, determination, action or maintenance stage because they appreciate the benefits of postpartum family planning.

Theory of Planned Behaviour

This theory started as the Theory of Reasoned Action in the 1980s which assumed that people had absolute control over their behaviours. However, this was limited by the fact that some individuals believe they have little or no control over their behaviours. To overcome this limitation, Ajzen introduced the element of perceived behavioural control into the Theory of Reasoned Action, thus transforming it into the Theory of Planned Behaviour (Figure 6). According to Ajzen and Fishbein (1980), the most important determinant of behaviour is the individual's behavioural intent. They described the constructs as follows:

- Behaviour: The transmission of intention or perceived behavioral control into action;
- 2. Behavioral Intention: Indication of how hard people are willing to try and of how much an effort they are planning to exert, in order to perform the behaviour. These are influenced by a person's *attitude* toward performing the behaviour, the perceived social pressure, called *subjective norm* and *perceived behavioural control*;
- Attitude: It is the degree to which the person has a favourable or unfavourable evaluation of the behaviour under consideration and is the first determinant of behavioural intention;
- 4. Subjective Norm: It is the second predictor of behavioural intention and defined as the influence of social pressure that is perceived by the individual (normative beliefs) to perform or not perform acertain behaviour and weighted by the individual's motivation to comply with those perceived expectations (motivation to comply); and
- 5. Perceived Behavioral Control: Described as the third antecedent of behavioral intention, this construct is defined as the individual's belief concerning how easy or difficult performing the behaviour will be and often reflects actual behavioural control.

They assumed that individuals are usually rational, make systematic use of information available to them and consider the implications of their actions before they decide to engage or not engage in a given behaviour. According to Ajzen (1991), the Theory of Planned Behaviour assumes a causal link between attitudes, subjective norms and perceived behavioural control, and behaviours, through behavioural intention (Figure 6).



Figure 6: Theory of Planned Behaviour Model

Source: Ajzen (1991)

The strengths of the theory of planned behaviour are that it can explain the relationship between behavioral intention and actual behaviour, help predict health-related behavioral intention and explain individuals' social behaviour by considering "social norm" as an important variable (Ajzen, 1989).

Brewer and Rimer (2008) assessed several aspects of the causal link claim by Ajzen and Fishbein and think that, highly specific behavioural intention measures that match the intended behaviours are required. Firstly, they suggested that intentional measures like planning, desire and expectations are likely to match behavioural intentions. Based on these suggestions, the following examples could be used in this study as intentional measures: "do you plan to use family planning after delivery?", "would you like to use family planning after delivery?", "is it likely that you would use family planning after delivery"? . Gollwitzer (1999) however asserts that behavioural willingness is a stronger predictor of behaviours compared to behavioural intentions. Secondly, empirical evidence suggests that intentions do not always lead to behaviour as postulated by the Theory of Planned Behaviour.

Despite the strong intention-behaviour relationship observed in some longitudinal studies (Callahan & Becker 2014), the relationship is weaker for very well designed experimental studies, studies of intentions measured further away in time from behaviours, risk behaviours performed within a social context and behaviours influenced by habits (Webb & Sheeran, 2006). Thirdly, Brewer and Rimmer (2008) further observed that the influence of attitude on behaviour is not always mediated by intentions. They assert that both intentions and attitudes change, depending on factors such as fear, threat, mood changes or past experiences, making their relationships to behaviour something of a moving target.

Despite consideration of normative influences and regardless of the individual's intention, socio-economic and environmental factors, and opportunities and resources needed to be successful at performing the behaviour, were not considered. The theory assumed that, perceived behavioural control predicted actual behavioural control, but this may not always be the case and thus may create some measurement problems. Applying the theory to this study, use of postpartum family planning would be the behaviour of interest, whilst attitude towards the behaviour may be the woman's thinking that, subsequent birth less than two years from the last one could be detrimental to her health and that of the child. Approval or non-approval of postpartum contraceptive uptake based on normative beliefs of partner, motheror father-in-law, religious group or any other social influence defines the subjective norms, whilst the woman's perception that she can still use postpartum family planning despite disapproval by significant others, implies a perceived behavioural control.

Steps to Behaviour Change Model

This model was developed by Piotrow et al. (1997) to explain behaviour change in their studies on family planning. They described the five major constructs of Knowledge, Approval, Intention, Practice and Advocacy in adopting family planning (Table 1, Figure 7). According to Piotrow and colleagues, individuals and groups progress from knowledge to sustained behaviour change and advocacy; and at any point in time people will be at different stages of change and will constitute distinct audiences that require different approaches, to meet their needs. They further observed that, not all individuals go through each step of the process in the same order, same speed, or at the same time; and that as knowledge and approval reach high levels in advanced programmes, emphases shift to later steps. The strengths of the model are that it helps effective targeting of interventions at individuals in different stages, to ensure positive behaviourchange outcomes (Piotrow et al., 1997)

Because this model was developed within the context of behaviour change related to family planning, it is easy to assume that it is the most suitable for this study. Some limitations that make the model inadequate are that: it focuses on the individual without considering effect of structural and environmental issues on ability to enact change; it does not account for social norms and public policy which also have effect on behavior change; the steps present a descriptive rather than a causative explanation of behaviour. It is important to point out that there is no direct relationship between knowledge and practice.



Figure 7: Steps to Behaviour Change Model

Source: Piotrow et al., (1997)

Table 1: Steps to Behaviour Change Constructs

Construct Description

- Knowledge Individuals recall family planning messages, understand what family planning means and can name family planning method(s) and/or source of supply.
- Approval Individuals approve of family planning and respond favourably to family planning messages. They further discuss family planning with personal networks (family, friends, community and any significant others) and have the perception that they approve of family planning.
- Intention Individuals recognize that use of family planning can produce positive outcomes and can meet personal needs; therefore, they intend to consult a provider and also intend to practice family planning at some point.
- Practice Individuals go to a provider of information/supplies/services to seek information, choose methods and begin family planning use. Some individuals who may have started family planning use at some point in time continue usage.
- Advocacy Individuals experience positive outcomes of family planning and acknowledge personal benefits; they advocate practice to others and support programmes in the community.

Source: Piotrow et al. (1997)

Postpartum Contraceptive Use and Postpartum Behaviours Framework

This framework was developed by Gebreselassie and colleagues in 2008 to study the relationship between contraceptive use and postpartum behaviours consisting of breastfeeding, sexual abstinence and amenorrhoea (Gebreselassie, Rutstein, & Mishra 2008). In this dynamic framework (Figure 8), the trajectory from birth to use of contraception is described. For example, a woman may use a reversible contraceptive like intrauterine device (IUD) or sterilization method immediately after birth, thus preceding the commencement of menses, sexual activity and weaning. Furthermore, the period of insusceptibility, which is influenced by sexual abstinence and breastfeeding, lengthens the time until the next birth. Use of contraception within this period offers extra protection against the risk of a subsequent pregnancy. The timing of such use should not fall more than a month after this period.

Postpartum sexual abstinence tends to have additional contraceptive benefits if the duration of abstinence exceeds that of postpartum amenorrhea. It should be noted that socio-cultural norms strongly influence the length of postpartum abstinence. The contraceptive effect of breastfeeding is felt when the mother fully or nearly fully breastfeeds, whilst remaining amenorrheic. Over 98 percent protection against pregnancy is offered when these conditions are fulfilled (Gebreselassie et al., 2008)

The relationship between contraceptive use after birth and amenorrhea is complex. The use of a particular method, especially, hormonal methods, before the resumption of menses may have a direct influence on the duration of postpartum amenorrhea. Injectables for example are known to produce extended periods of amenorrhea in users and thus, if started before the resumption of fecund menstrual cycle, may lead to an abnormally long duration of postpartum amenorrhea. In contrast, women who use the pill have a shortened length of postpartum amenorrhoea. A major limitation of this framework is that the pathway is not sequential, in that depending on the type of contraceptive method selected and the postpartum situation of the mother, the decision to start using contraception takes place any time after birth.



Figure 8: Postpartum Contraceptive Use and Postpartum Behaviours Framework

Source: Gebreselassie, Tesfayi, Shea O. Rutstein, and Vinod Mishra(2008).

Davies and Blake Model of Fertility

This model was developed by Davis and Blake in 1956, in their work on 'Social Structure and Fertility'. Davis and Blake (1956) proposed eleven (11) intermediate variables through which any social or indirect factors influencing the level of fertility must operate. They proposed that these factors are those that are directly associated with the process of reproduction (intercourse, conception and gestation & parturition) as shown in Figure 9.

In the framework, the indirect factors consist of individual characteristics such as demographic and socio-economic characteristics of the women, and the environmental factors which include (a) the social factors such as parental / family characteristics and Partner characteristics; (b) institutional factors such as Religion, community norms and values, health systems and the economy; and (c) policy, legal and political factors. These indirect factors do not bring about a direct change in fertility but do so through the intermediate determinants.

The intermediate determinants are those that directly affect fertility and through which and only through which the indirect factors work to influence fertility. They consist of intercourse factors, conception factors and gestation/parturition factors. The intercourse factors determine the probability of a couple attaining conception. They consist of age of entry into sexual union, proportion of women never entering sexual unions (permanent celibacy), amount of reproductive period spent after or between unions (when unions are broken by divorce, separation or desertion or when unions are broken by death of partner),

67

voluntary and involuntary abstinence (from impotence, illness, unavoidable but temporary separations) and coital frequency (excluding periods of abstinence).

The conception factors consist of fecundity or infecundity as influenced by involuntary factors, use or non-use of contraception and fecundity or infecundity as affected by voluntary factors such as sterilization, sub-incision or medical treatment. The gestation variables are those that are associated with foetal loss mediated by involuntary factors such as spontaneous abortion miscarriage or foetal loss mediated by voluntary factors such as induced abortion.

From a comparative sociological point of view, Davies and Blake (1956) quantified the influence of the intermediate determinants on fertility in different societies depending on their level of economic development. In developing countries where this model was mostly applied, they classified the intermediate determinants into those with high, intermediate, low and indeterminate values or influence on fertility.

Those with high influence included age of entry into unions, permanent celibacy, contraception and fecundity or infecundity as affected by voluntary factors like sterilization; whilst those with intermediate influence included time between unstable unions, post -widowhood celibacy and voluntary foetal loss. The intermediate determinants with low influence included voluntary abstinence and involuntary foetal loss and those with indeterminate influence consisting of involuntary abstinence, coital frequency and involuntary sterility (Davis and Blake, 1956).



Figure 9: Davies and Blake Model of Fertility

Source: Davies and Blake (1956)

The strength of the model is its easy application using widely available data to decompose the contribution of each of the intermediate variables selected on the current levels of fertility over time and across regions (Anyara & Hinde, 2006). The fact that the model considers the social, economic and environmental factors in its framework also makes it easily applicable to this work. The model was limited by its application at the levels of population and society and not at individual levels and also by the fact that values for all the intermediate variables were difficult to derive (De Bruijn , 2006). This latter limitation made (Bongaarts, 1978) translate the social model of Davies and Blake into a bio-social model comprising eight proximate determinants classified as follows: (a) Exposure factors: proportion married (1); (b) Deliberate marital fertility control factors: Contraception (2), Induced abortion (3); and (c) Natural marital fertility factors: Lactational infecundability (4), frequency of intercourse (5), sterility (6), spontaneous intrauterine mortality (7), duration of fertile period (8).

Conceptual Frameworks Adapted for the Study

Following the review of all the theories, models and frameworks, two models were adopted to inform this study. These were the Davies and Blake model of fertility developed in 1956 and the Steps to behaviour change model developed by Piotrow et al. in 1997. All the constructs in the steps to behaviour change model (Knowledge, Approval, Intention, Practice and Advocacy) were fully adopted in this study (see Figure 7 and Table 1). For the Davies and Blake model of fertility, not all the constructs were important to this study and therefore the model was adapted to suit the study. The Indirect and proximate determinants that influence fertility were the main factors considered. The Indirect determinants, mainly Socio-demographic (age, sex, gravidity, parity, marital status, religion), socio-economic (education, occupation, living conditions) and socio-cultural (family sex composition). The proximate determinants mainly considered were the intercourse (voluntary sexual abstinence) and conception factors (infecundity, breastfeeding and lactational amenorrhoea and contraceptive use) [Figure 10].



Figure 10: Adaptation of Davies and Blake Model 1956 Source: Davies and Blake (1956)

Summary

This chapter identified some of the issues on postpartum contraceptive use and some models used in studying behaviour change and fertility which have implications for postpartum contraceptive use. Among the issues on postpartum contraceptive use reviewed were intention as a predictor of behaviour, basis for interest in women's intention to use and use of family planning, factors influencing intention and actual use of contraceptives and unintended pregnancies and associated determinants. The review showed growing consensus about the likelihood of occurrence of a given behaviour if individuals have strong intentions to carry out the behaviour. This was however premised on condition that there are no environmental barriers and the individuals have some skills and abilities to carry out the behaviour. Evidence exists that intentions reasonably predict behaviours (Fishbein, 2008; Adegbola & Okunowo 2009; Agha, 2010; Keogh, 2012; Callahan & Becker, 2014). This study assessed the value of intentions of pregnant women as a predictor of family planning adoption in the postpartum period.

A basis for interest in contraceptive use by individuals, families, communities and programmes is unintended pregnancy, which for purposes of this study refers to pregnancy that is not wanted or mistimed at the time of conception (Center for Disease Control, 2010). High rates of unintended pregnancies have been observed in SSA including Ghana and are linked to various negative outcomes which predispose families to poverty, high tendency to maternal morbidity and mortality. Non-use of contraception, inconsistent or incorrect use of contraceptives and contraceptive failure have been linked to unintended pregnancies which is an indication of unmet need for contraceptives is estimated whilst its determinants are also elucidated to help determine interventions to reduce its incidence.

72

Factors influencing intention to use and actual use of family planning are varied and depend on the setting and the context. These have generally been observed to be socio-demographic and socio-economic. Age, educational status, marital status, wealth quintile and parity for example, have been the most common predictors in various studies to influence contraceptive use intention and actual use (Adegbola & Okunowo, 2009; Agha 2010; Arbab et al., 2011; Kariuki, 2011;; Sharma, Mohan, Das, & Awasthi, 2012; Yihunie et al., 2013). Obstacles to contraceptive use intention are commonly linked to socio-cultural factors (see Agha, 2010) whilst those linked to actual use are mainly socio-cultural and socioeconomic (see Agha, 2010; Arbab et al., 201; Yihunie et al., 2013). This study contributes to the search for predictors of contraceptive use intention and actual use and will enrich the existing body of knowledge. It will also help design interventions to increase contraceptive adoption.

This chapter also discussed theories, models and frameworks that influence human behaviour. Those discussed include the Health Belief Model, Social Cognitive Theory, Transtheoretical (Stages of change) Model, Diffusion of Innovations Model, Theory of Planned Behaviour, Steps to Behaviour Change Model, Framework for Postpartum Contraceptive Use and Contraceptive Behaviour and Davies and Blake Model of Fertility. These have been used in different contexts and settings to explain human behaviour apart from contraceptive use intention and actual use.

Steps and processes involved in changing contraceptive behaviour which have theoretical underpinnings help improve contraceptive prevalence. This is

73

more applicable and relevant in areas of low contraceptive use and among poorer women (Creanga, Gillespie, Karklins, & Tsui, 2011). Theories based on contraceptive behaviour are often inspired by processes through which people learn to adopt the new behaviour. This involves acquiring knowledge of the risks and benefits of adopting contraceptives, developing the confidence that one has the power to adopt contraceptive behavior (belief in self-efficacy), and determining outcome expectations based on a cost-benefit analysis of adopting or not adopting contraceptive behaviour (Michie et al., 2005). Evidence exists that although health behaviour theories nominally have universal relevance, many questions remain about their applicability to contexts outside the ones in which they were developed(Michie et al., 2005). The two models that were adopted to inform this study were the adapted Davies and Blake model of fertility developed in 1956 and the Steps to Behaviour Change model developed by Piotrow et al. in 1997.

CHAPTER THREE

METHODS OF DATA COLLECTION AND ANALYSES

Introduction

This chapter describes the research design, data sources, sampling procedures (sample size and sampling methods), methods of data collection, study implementation, data, processing and management; data analysis and limitations to data collection. It also covers the experiences and challenges encountered during the study, ethical issues and data quality assurance.

Profile of Study Area

The study was carried out in the Mfantseman Municipal Area of the Central Region of Ghana. The Central region was selected because the Ghana Demographic and Health Survey (GDHS) (2008) report revealed that, the region had the highest rate of teen pregnancies (23%), lowest rate of discussion of family planning with health providers, poor attitudes of men towards family planning and the highest unmet need for family planning in the country (50%) (GSS & Macro, 2009).

The Mfantseman Municipal was selected because according to the annual report (2010) of the municipality, teen pregnancies were high (13.6% of all pregnancies; i.e. totaling 1,024), abortion rates amongst the teens were high and family planning uptake in general were lowest in the region over the three year period from 2007 to 2009 (Figure 11). This Low uptake of family planning

prompted a comment in the 2010 report of the municipality thus: "this trend is worrying and calls for urgent action to address this challenge" (Mfantseman Municipal Health Directorate, Annual Report, 2010:25)



Figure 11: Family planning Acceptor Rate by District (%), Central region 2007 – 2009

Source: Central Regional Report, Ghana Health Service (2010)

The Mfantseman Municipality is one of the 20 Metropolitan, Municipal, and Districts in the Central Region of Ghana, with Saltpond as its administrative capital. The Municipal area is located along the Atlantic coastline of the Central Region of Ghana and extends from latitudes 5° T to 5° 20" North of the Equator and longitudes 0° 44" to 1° 11" West of the Greenwich Meridian, stretching for about 21 kilometers along the coastline and for about 13 kilometers inland. It covers an area of 612 square kilometers. The Municipality shares boundaries with Gomoa West District to the East, to the West with Ekumfi District, to the North with Ajumaku-Enyan-Essiam District and to the South with the Gulf of Guinea (Figure 12).



Figure12: Map of Central Region Showing Mfantseman Municipal Area Source: University of Cape Coast, GIS Unit, UCC (2014)

The Mfantseman Municipal Area is situated in a low-lying area at an elevation of less than 60 metres above sea level and is drained by a number of rivers, lagoons and streams. It is characterized by milder temperatures of between 24 °C and 28 °C because of its close proximity to the Atlantic Ocean. It has a relative humidity of about 70 percent, and experiences double maximum rainfall, with peaks in May–June and October. The vegetation consists of dense scrub tangle and grass that grow to an average height of 4.5 metres. Such flora together with the fauna offers good opportunities for agriculture and fishing. The major towns, Biriwa, Anomabo, Saltpond, Otuam, Abandze and Kormantse, are all along the coast. The major inland town is Mankessim (Figure 13)

According to the 2010 population and housing census, the Mfantseman Municipality has a total population of 196,563, consisting of 89,025 males and 107,538 females respectively (sex ratio – 82.8). The annual population growth rate of the municipality is estimated to be 2.8 percent. Approximately, 27.9 percent of the total population of the municipality lives in urban settlements (population exceeding 5,000), while 72.1 percent falls into the category of rural residence (average population for most rural settlements is about 950). The population is predominantly Fanti, however, Ewes, Grushies, Dargatis, Moshis and Mamprusis are also found in in all the major towns. Fishing and farming constitute the main economic activities of the Mfantseman Municipal. Occupational distribution is as follows – fishing -51 percent, farming -30 percent, and commerce -19 percent.

The Municipal area has 27 health facilities consisting of hospitals, health centres, CHPS centres and outreach points. The two main hospitals in the municipal area are the Saltpond Government hospital which serves as the main referral facility for most health facilities in the Municipal area, and Fynnba hospital a private hospital located in Mankessim. The health centres are located at Anomabo, Mankessim, Abeadze Dominase, Essuehyia, Otuam and Biriwa (affiliated to Amicus Onlus). Nsanfo, Taido, Narkwa, Nanaben, Ekumpoano, Edumafa, Eyisam and Ekrawful, host functional Community Based Health Planning and Services (CHPS) compounds. Two maternity homes, St Anthony's Maternity home located at Mankessim and God's Gift Maternity home located at Ekrawful exist in the Municipal area.

A Reproductive and Child Health (RCH) Clinic and Ghana Health Service Community clinics which are government owned are located at Saltpond/Kormantse, Ebuakwa and Srafa Kokodo respectively. An outreach post is also located within the Mankessim market. Five other private clinics which complement the government sector are found in Yamoransa, Mankessim and Eyisam.



Figure 13: Map of Mfantseman Municipal Area Showing Selected Health Facilities Used for the Study

Source: University of Cape Coast, GIS Unit, UCC (2014)

Maternal health services provided by the hospitals and health centres in the Municipal area include antenatal, skilled delivery, postnatal, family planning and emergency obstetric and neonatal care services. Antenatal care coverage has been high (Table 2). For example, between 2008 and 2010, average coverage has been over 90 percent (93.3%), with average antenatal visits of 4.5 per woman within the period. Pregnant women attending antenatal care, four or more times between 2008 and 2010 was also quite high (average 74.3%).

INDICATOR	2008	2009	2010	2010
	(actual)	(actual)	(target)	(actual)
Antenatal coverage (%)	104.0	94.0	95.0	87.5
Average No. of visits	4.2	4.7	5.0	4.7
Pregnant women with 4+ visits	61.1	84.7	85	77.3
Teenage pregnancies (%)	15.2	14.5		13.6
Skilled deliveries (%)	41.2	42.5	45.0	44.0
TBA deliveries (%)	28.0	15.7	15.0	20.8
Maternal deaths	6.0	2.0	0.0	0.0
Postnatal coverage (%)	64.7	62.2	62.0	49.5

Table 2: Trends in Maternal Health Indicator Coverages 2008 - 2010

Source: GHS, Municipal Health Directorate, Mfantseman, Annual Report (2010)

These performances notwithstanding, antenatal care coverage fell short of the 95 percent and 85 percent targets set in 2010 for single and 4+ visits respectively.

Teenage pregnancies are high risk pregnancies and are also generally unintended. In 2010, 1,024 teen pregnancies representing 13.6 percent of all pregnancies were recorded. This was an improvement over those recorded in 2008 and 2009 (Table 2). This improvement has been attributed largely to improvement in the adolescent friendly services in the municipality, with the establishment of four adolescent friendly centres and community libraries/recreational centres.

A total of 4,864 deliveries were recorded in all health facilities in the municipal area in 2010. Of these, 44 percent (3,292) were by skilled attendants. This was a 1.5 percentage point increase over that reported in 2009. One of the goals of the national safemotherhood programme is to increase deliveries by skilled attendants and reduce considerably those conducted by Traditional Birth Attendants (TBAs). This notwithstanding, the deliveries by TBAs in the municipal area increased from 15.7 percent to 20.8 percent (over and above the target of 15%).

Maternal deaths in the municipal have reduced to zero from six and two respectively in 2008 and 2009. This achievement had been attributed to early referral system, use of partograph, life-saving skills training for midwives and emergency obstetric and neonatal care services. Postnatal care coverage declined from 64.7 percent in 2008 to 49.5 percent in 2010. The decline could be explained by entrenched socio-cultural factors and wrong perception by couples that mother and child have low risk of postnatal morbidity and mortality (Ghana Health Service, Municipal Health Directorate, Mfantseman, Annual report, 2010)

Family planning uptake in the Mfantseman Municipal area has been the lowest in the region from 2007 to 2009. Efforts had been made in the past to improve the uptake of family planning. For instance, a family planning campaign was launched by Ghana Health Service to promote injectables, condoms and combined pills; but these were not successful because of myths and negative beliefs about modern contraceptives (Ghana Health Service, Municipal Health Directorate, Mfantseman, Annual Report, 2010). The Municipal Health Directorate is also actively engaging the communities using community-based volunteers (CBVs) and Traditional Birth Attendants at outreach sites and CHPS compounds to improve uptake of family planning.

Study Design

The data for this study was primary and was obtained using a prospective panel design. This involved the collection of quantitative data in two Phases. In Phase 1, data was collected from pregnant women attending antenatal clinics in four selected health facilities in the Mfantseman Municipal area within the survey period to find out about their intention to use family planning after they had delivered. In Phase 2, the same groups of women were followed up to two years after the first interview in Phase1 to collect data about the translation of their intention to actual use of postpartum family planning among other postpartum issues. Attrition is accounted for in subsequent sections:

Study Population

The study population consisted of pregnant women attending antenatal clinic (ANC) at the Saltpond hospital, Mankessim health centre, Anomabo and Biriwa health centres. These health facilities were purposively selected because they have well established and well patronized Reproductive and Child Health (RCH) centres and because they give a good mix of semi-urban and rural characteristics. In Ghana, an urban area is defined as a locality with a population of 5,000 or more (Ghana Statistical Service, 2012), with an economic sector that is mainly non-agricultural. On the other hand, a rural area is defined by populations less than 5,000 (Ghana Statistical Service, 2012) and with an economy that is predominantly agro-based.

This simplistic rural-urban dichotomy has led to the use of the concept such as semi-urban, which strides rural and urban areas. Several researchers have criticized the dichotomous basis of urban and rural definitions, arguing that many of the characteristics that define rural and urban areas exist along a continuum within which individuals, households and communities and institutions distribute themselves (Iaquinta & Drescher, 2000). These descriptions best fit the two semiurban settlements in this study i.e., Mankessim and Saltpond. They both have populations greater than 5,000 but their economies are predominantly agro-based. Mankessim's economy is characterized by crop farming whilst that of Saltpond is characterized by fishing. The two other areas selected (Anomabo and Biriwa) are mainly rural in nature. Their economies are characterized mainly by fishing. Pregnant women in these areas were selected because their reproductive health behaviours would be influenced directly by socio-cultural practices and economic activities in their areas of residence, by reproductive health (including family planning) interventions put in place by the Municipal Health Directorate and the health facilities in the area and by their individual characteristics.

Target Population

The target population consisted of pregnant women between 15 and 49 years who resided in the municipality and attended antenatal clinic in any of the selected health facilities (Saltpond Government hospital, Mankessim health centre, Biriwa health centre and Anomabo health centre) at the time of the survey (first quarter of the year 2012). To be included in the study, participants had to be Ghanaian women, pregnant at the time of the study and living in communities within the Mfantseman Municipality of the Central Region. Verification of their Ghanaian status was not done because their reported citizenship in the Ghanaian languages spoken was considered adequate. Ownership of an approved Ghana Health Service antenatal card bearing an OPD number was a necessary condition imposed by the study. The rationale for choosing the ages 15 to 49 years is that, it is the defined age group for women in reproductive age (WRA) in Ghana (GSS &Macro, 2009).

Antenatal attendance and subsequent delivery in any of the health institutions within the municipal ensures maximum contact with the health system and its resultant influence on uptake of reproductive health services including
family planning. Table 3 shows the number of pregnant women aged 15-49 years attending antenatal clinic at the selected health facilities in the municipal area from 2008 to 2010. Because the survey was planned for the first quarter of the year 2012 the target population was estimated based on average first quarter ANC attendance from 2008 to 2010 (available data). The target population was computed by adding the first quarter ANC attendance for the three years (2008 to 2010) and dividing the total by 3 to obtain a target population of 4,218 pregnant women likely to be encountered during the survey period and from which sample for the study was taken.

Table 3: Pregnant women 15 to 49 years Attending Antenatal Clinic at LeastOnce at Selected Health Facilities in Mfantseman Municipal (2008 –
2010)

Total number of women attending antenatal clinic								
Year	Saltpond	Mankessim	Biriwa	Anomabo	Total	Total per quarter (N/4=4218)		
2008	9,459	2,700	2,578	2,639	17,376	4,344		
2009	8,756	2,826	2,812	2,819	17,213	4,303		
2010	8,155	2,640	2,608	2,624	16,027	4,007		

*Average total number of women visiting ANC at all selected health facilities per quarter, per year from 2008 to 2010

Target population

Source: Central Region Reproductive and Child Health Reports (2008-2010)

4,218*

Sampling Procedures

Based on the estimated target population of 4,218, and the assumption that 50 percent of pregnant women intend to adopt postpartum family planning, within a margin of error of 3 percent, a minimum sample size S_T , was estimated as follows: For a finite population, the sample size S_T , is estimated by the formula

 $S_T = A / [1 + (A-1)/T]$ (Creative Research Systems, 2012)

Where A is given by $[Z^{2*}P^{*}(1-P)] / C^{2}$; T = estimated target population; Z= Z value (1.96 for 95% confidence); P= Proportion of pregnant women who intend to adopt postpartum family planning; and C = margin of error.

$$A = [1.96^{2}(0.5) (0.5)]/0.03^{2} = 1,067$$

 $S_T = 1,067 / [1 + (1,067 - 1)/4218] = 852$; (approximated to 900).

For effective antenatal care and early identification and management of complications, it is important and advisable for all women to report to antenatal clinic as soon as they realize they are pregnant; implying that all pregnant women would be assumed to register in the first trimester of pregnancy. Evidence shows that this is not the case: only 43 percent register in their first trimester, whilst the remainders register in the second or third trimesters (Ghana Health Service, Central Regional Family Health Report, 2011). It is also recommended that all antenatal registrants would deliver at the health facilities at term to ensure adequate supervision during delivery by skilled personnel. However, only 44% of them deliver at the institutions within the municipal area (Ghana Health Service, Central Regional Family Health Report, 2011). This implied a default rate of close to 60 percent. To take care of defaults and late registrations, the minimum sample size (Sr= 900) computed was doubled to 1,800 with an additional 10% mark-up

for women who would decline to be interviewed. The estimated total sample size is 1,980 (Table 4).

Facility	Estimated sample size
Saltpond hospital (Sm)	990
Mankessim health centre (S_{hc})	330
Anomabo health centre (Shc)	330
Biriwa health centre (Shc)	330
Total (ST')	1,980

Table 4: Sample Siz	e by Selected	Health	Facility
---------------------	---------------	--------	----------

Source: Field data (2012)

The estimated minimum sample size ($S_{T'}$) of 1,980 eligible pregnant women attending antenatal clinics in the municipality were selected from the Saltpond hospital and Mankessim, Anomabo and Biriwa health centres based on trends in antenatal attendance records from 2008 to 2010. Records of the Municipal Health Directorate indicated that the ratio of ANC attendance at the Saltpond hospital to each of the health centres was approximately 3:1 (Table 3). Therefore, the estimated sample size for each health centre is given by $S_{hc} = S_m/3$; where S_m is the sample size for the Saltpond municipal hospital. The estimated sample size for the Saltpond municipal hospital was approximately 990 and that of each health centre was 330 (Table 4). Within the survey period, each of the selected health facilities was visited during the days designated for antenatal. At the selected health facilities, all antenatal registrants (pregnant women), irrespective of the period of gestation, who lived in the Mfantseman Municipal area and who were aged 15 to 49 years, were targeted to be part of Phase 1 of the study. They were selected if they gave their consent. They were also requested to give their consent for follow-up (i.e Phase 2). For Phase 2 (follow-up) of the study, it was assumed that all eligible respondents in Phase 1 would agree to be followed up. However, only 71% of the 1,914 respondents agreed to be followed up, giving an estimated Phase 2 sample size of 1,359.

Methods of Data Collection

A quantitative survey technique using questionnaires were employed in collecting data for the study in Phases 1 and 2. Two types of data collection strategies were employed in this study. They are Face-to-face and Telephone interviews. Face-to-face interviews were employed in Phase 1, whilst, Telephone and face-to face interviews were employed in Phase 2. Details of actual data collection are discussed in subsequent sections.

Data Collection Instruments

Two types of instruments were used (Appendices 2 and 3): one for Phase 1 and the other for Phase 2. As much as possible, survey questions were adapted from standardized questionnaires used in Demographic and Health Surveys, Maternal Health Surveys and Multiple Indicator Cluster Surveys in Ghana.

The questionnaire used for Phase 1 consisted of twelve sections (A-L). Section A was on background information of the pregnant women such as age, highest completed education, ethnicity, marital status, number of children ever born, area of residence, facility where client was interviewed (facility code), religion and occupation. Section B was on background characteristics of respondent's partner (if any) such as age, ethnicity, highest completed education, religion, occupation, previous marital relationships and other children besides those with respondent. Section C had questions on relationship issues such as number of years relationship with present partner has lasted, living arrangements with current partner (living in same house or not), whether partner has other wives or spouses apart from respondent.

Section D covered socio-economic status of respondent, whilst Section E had questions on reproductive history and current pregnancy such as gravidity, history of abortions/miscarriages, history of loss of children and status of current pregnancy (whether intended or unintended). Section F, had questions on knowledge and ever use of family planning methods (modern and traditional) whilst Sections G,H and I had questions on knowledge of lactational amenorhoea method (LAM), acceptability of postpartum family planning by respondent and significant others (partner, mother-in law, father-in-law, religious influence) and past experience with postpartum family planning respectively.

Sections J and K had questions on postpartum family planning intentions (to use or not and which methods) and whether respondents would allow to be followed up in Phase 2 respectively. Section L is supplementary to Sections A and B and consisted of detailed questions about religious denomination, number of children ever born (number dead or alive), relationship with partner and partner's relationship with other women.

The follow-up (Phase 2) questionnaire was in four sections (A-D). Section A was the same as that of Phase 1 and was to confirm that, it is the same person who was engaged in Phase 1. Section B had questions on delivery or postpartum outcomes such as, outcome of pregnancy (live term, stillbirth, pre-term or miscarriage), place of delivery, sex of newborn, mode and date of delivery. Section C asked questions on some postpartum behaviours such as amenorrhoea, sexual abstinence, breastfeeding, information about family planning and spousal discussion about family planning. Finally, Section D enquired about postpartum contraceptive use. It explored whether the respondent had used or was using family planning following delivery and at what point within the extended postpartum period that was done. For those who had not used any method, this section explored reasons behind their decision not to use family planning.

Preparatory Visits to Study Area

Before the study began, letters were written to the Central Regional and Mfantseman Municipal Directorates of the Ghana Health Service informing and introducing the researcher and the study to them and indicating the time for discussion with the Directorate. Meetings were held with the Municipal Health Directorate and the Medical Superintendent of the Saltpond Municipal Hospital on 20th June 2011 and on 21st November 2011 respectively. The meetings offered opportunity to discuss and explain the study, and to recruit research assistants and supervisors. The next visits were to the Mankessim, Anomabo and Biriwa Health Centres where meetings were held with the Medical Assistants in charge of the facilities, to explain the objectives of study and sought permission and their support.

Pre-testing of Study Instruments

Selection of field assistants and supervisors was carried out with the help of the Municipal Director of Health Services and the Medical Superintendent of the Saltpond hospital using the following criteria: Prior research and data collection experience within the district, tertiary level education, fluency in English and Akan languages and knowledge of the communities in the municipality. Based on these criteria, nine field assistants (8 females and one male) were recruited out of fifteen (15) persons who applied for the Phase 1 data collection in January 2012. The only male recruited was made the supervisor based on his extensive research experience. Six field assistants were also recruited based on the same criteria for Phase 2 of the data collection in January 2014.

For both phases of data collection, three-day training sessions were organized. Day one was used for the theoretical basis of family planning and postpartum family planning and family planning programming in Ghana and the

91

. .

district. Day two was used for review of the data collection instruments, whilst day three was used for translation, role playing and pre-testing. Resource persons included the principal investigator, Municipal Director of Health services and the Medical Superintendent. An experienced district court registrar who doubles as a translator was employed to support the translation and role plays.

The study instruments were pre-tested and this was carried out at the Cape Coast Metropolitan Hospital and the Adisadel Health Centre in Cape Coast for both Phases 1 and 2. For Phase 1, pre-testing was conducted amongst pregnant women, whilst that of Phase 2 was conducted amongst postpartum women attending postnatal clinic at sixth week post-delivery.

In the pre-test of the Phase 1 questionnaire, it was noted that the information on the cover page that explained details of the study were not adequate and needed to be beefed up. It was also noticed that it was necessary to capture the specific location of respondents to aid follow-up in Phase 2. Other aspects of the questionnaire that were refined included details of religion and number of children ever born. The religion was disaggregated into denominations, whilst details such as number of children born but later died were included. No major differences in responses were elicited amongst the respondents generally.

It was realized during the pre-testing of the Phase 2 questionnaire that eliciting responses was more difficult than the Phase 1 questionnaire. This was because respondents had to provide several dates and time durations (date of delivery, duration of breastfeeding, onset of sexual intercourse, period of amenorrhoea, etc.) with reference to their time of delivery. Recalls of these were difficult for most of the clients. More time was spent on improving skills of field assistants to elicit responses and pre-testing the questionnaire (seven days instead of three days) to ensure errors were reduced to the minimum. This made three of the research assistants leave the study, with the excuse that remuneration was not commensurate with the amount of effort exerted. These were replaced with new assistants.

Data Collection

Phase 1 of the field work was carried out from January to April 2012 at the four study sites (Saltpond hospital, Mankessim, Anomabo and Biriwa health centres). Five field assistants were assigned to the Saltpond hospital, whilst one field assistant each was assigned to the three health centres. One Supervisor and the Principal Investigator supervised the research assistants.

Facility Facility Estimated Number of Response rate Sample Size questionnaires (%) administered Saltpond hospital 990 968 97.8 Mankessim health 330 289 87.6 centre

327

330

1,914

99.1

100.0

96.7

330

330

1,980

 Table 5: Number of Instruments Administered and Response Rates by Health

 Facility

Source: Field data (2012)

Anomabo health

Biriwa health centre

centre

Total

A total of 1,914 questionnaires were completed within the period. This was 3.4 percent short of the 1,980 questionnaires estimated based on the sample size to be administered. The questionnaires administered by facility and response rates are as shown in Table 5.

The high response rates observed in Phase 1 were due to the following reasons:

- 1. The survey was facility based and made it possible and easier to meet many potential respondents during their scheduled antenatal visits;
- 2. A positive publicity about the survey was generated by the Municipal Health Directorate and the staff of the hospital and the health centres where the study was conducted. This generated interest amongst the pregnant women about the issues contained in the questionnaire;
- 3. The research assistants made administration of the questionnaires interesting, attractive and easy to complete because they understood and found the issues interesting and meaningful. This also made them tailor and personalize the communications between them and the respondents; and
- Respect and positive regard was shown by the research assistants to the respondents and this motivated them to respond and complete the survey questions.

The relatively lower response rate observed at the Mankessim health centre (Table 5) was because one of the research assistants assigned to the health centre was taken ill towards the end of the survey period and could not complete the required number of questionnaires assigned to her. Before Phase 2 of the survey started, periodic phone calls and occasional visits were made to respondents between 2012 and 2014 to be sure they would be available when Phase 2 began, and if they still wanted to be interviewed. Through the telephone contacts 540 respondents were traced. For the 819 who could not be contacted by phone, 481 of them were traced to their homes based on the descriptions of houses and localities provided during Phase 1. The remaining 338 could not be contacted either through the telephone calls or by means of the descriptions of houses provided.

Phase 2 was carried out from January to May 2014. For the survey, two research assistants were assigned to the communities in and around Saltpond, whilst one research assistant each was assigned to the communities in and around Mankessim, Anomabo and Biriwa. Of the 1,359 who agreed to be followed up, 1,004 could be identified, giving a response rate of 74 percent. Two main methods of follow-up were used in this phase. These included tracing the respondents to their homes and by telephone calls. In Phase 1, respondents who accepted to be followed up were made to describe directions to their homes and also provide telephone numbers on the completed questionnaires. Two of the field assistants were assigned to the communities in and around Saltpond, whilst one field assistant each was assigned to the communities in and around Mankessim, Anomabo and Biriwa.

Home visits and telephone calls were the two main approaches used to trace respondents for interviews in Phase 2. Based on descriptions of houses and communities and telephone numbers provided by respondents in Phase 1, attempts were first made to trace the houses of the respondents. When respondents were traced to their houses, they were interviewed face-to-face. In some circumstances respondents could not be traced to their houses. These respondents fell into three categories:

- a. Those who agreed only to be interviewed via telephone.
- b. Those with no or poor descriptions of houses and localities but were accessible through telephone numbers provided.
- c. Those with no or poor descriptions of houses and localities and were not accessible through telephone.

For those in categories (a) and (b) telephone interviews were conducted once they agreed. However, it was difficult to interview those in category (c) because contacts could not be made with them. After these processes, 1,004 respondents were interviewed, 340 by telephone and 664 by face –to-face interviews. These constituted about 53 percent of the 1,914 respondents interviewed in Phase 1 and 74 percent of the 1,359 respondents who agreed in Phase 1 to be followed up (Table 7). The high retention rate (90%) in Mankessim was due to the fact that descriptions of houses and localities provided by the respondents were very accurate and the telephone numbers provided were easy to reach. The relatively lower retention rates in Saltpond (45%), Biriwa (41%) and Anomabo (55%) (Table 7) were due to poor or inaccurate descriptions of houses and localities, inaccurate or inaccessible telephone numbers, non–provision of descriptions of houses and telephone numbers and refusal of some respondents to be interviewed. Table 6: Number and Percentage of Women Targeted and Interviewed per Health Facility Attended in

Phases 1 And 2

	d	hase 1	Phase 2		
Health facility	Number of respondents targeted	Number of respondents interviewed (a)	Number of respondents targeted	Number of respondents interviewed (b)	Percentage of Phase1 respondents in Phase 2 (b/a)
Saltpond hospital	066	968	654	432	44.6
Mankessim health	330	289	283	259	89.6
Anomabo health	330	327	242	179	54.7
Biriwa health centre	330	330	180	134	40.6
Total	1,980	1,914	1,359	1,004	52.5

Source: Field data (2012 and 2014)

67

Two hundred and twelve (212) women could not be reached at all, whilst 144 declined to be interviewed claiming, we were either bothering them, were no more interested or partners were not available.

Fieldwork Challenges

The study was initially conceived as a cross-sectional, quantitative study to determine factors associated with intention of pregnant women to use postpartum family planning and was to be completed within three years. However, it was decided that many benefits would be derived if the study was changed from a cross-sectional to a prospective panel (longitudinal) study. The change in design implied a waiting period of two years within which the pregnant women who were at different stages of their pregnancy, would have to deliver and be exposed adequately to some factors which would determine whether their intentions to use or not to use family planning or not would be translated into actual behaviour after delivery. This long waiting time interval greatly affected the completion time of the study.

In the first phase of the study, the work was made easy because the women were identified at health facilities. Phase 2 involved tracing women to their homes in various communities within the municipality with poor house numbering. This created difficulties for research assistants and supervisors. The research assistants and supervisors had to travel over wide areas to locate the houses before questionnaires were administered. This made it time consuming and, therefore, affected the speed with which questionnaires were planned to be completed. These challenges made the research assistants agitate for increase in fees charged. It was explained to them that there was no funding attached to the study and therefore they had to try their best to support us. Three of them abandoned the study and new research assistants had to be recruited to replace them.

Recalls of dates and durations of events in relation to the time of delivery was a challenge for some of the women. There were inconsistent reporting of periods of postpartum behaviours such as breastfeeding, resumption of sexual intercourse and amenorrhoea in relation to date of delivery and therefore research assistants spent time helping respondents to remember.

Some of the women complained that the questionnaire was long and therefore in some cases the interviews had to be terminated and rescheduled for another time. Others complained that it distracted their household activities such as cooking, cleaning and feeding their children. Some respondents especially in Saltpond, Anomabo and Biriwa complained that they were tired of being interviewed in several studies without any financial rewards or any improvements in their lives and refused outright to be interviewed.

Some funding from the USAID Small Grants programme was available for the study, but only for Phase 1. This is because the follow-up phase was not envisioned as part of the proposal presented to USAID for funding. This meant that Phase 2 had to be funded entirely by the Principal Investigator. Altogether, 3,273 questionnaires (minimum 3-pages each) were printed. This came at a huge financial cost especially in Phase 2. The recruitment and training of research assistants in Phase 2 also came with some logistical and financial demands. Between January and May 2014, several travels to and from the houses of respondents and telephone calls had to be made by the research assistants and supervisors to avoid high attrition. These brought about huge transportation and telephone costs.

Despite the training, pretesting and motivations offered to the research assistants and supervisors to ensure quality of data collection, some of the data that were collected showed inconsistencies. For Phase 1, if these inconsistencies were detected whilst the pregnant woman was still at the health facility, she was reinterviewed on those responses that showed inconsistencies. However, if the pregnant woman had left the facility attempts were made to contact her by phone to correct the inconsistencies. In cases where the respondents could not be traced and the inconsistencies were few, those responses were treated as missing. Otherwise, the questionnaires were discarded if the inconsistencies were unacceptable. For Phase 2, wrong and inconsistent responses were corrected via telephone calls or re-visiting the respondents at their homes.

Despite the challenges, the study offered opportunities for learning the conduct of research, applying theoretical research principles and appreciating problems related to longitudinal studies. Diverse socio-cultural procedures, especially pertaining to community entry were encountered and these led to strengthening of the community engagement skills of the principal investigator.

Ethical Issues

Ethical and administrative approvals were obtained from the Ethics Review Committee of the Ghana Health Service (GHS), and the Municipal Health

SAM JONAH LIBRARY UNIVERSITY OF CAPE COAST

100

Directorate (MHD). Written informed consent was obtained from each participant with a witness. Before each interview, the aims, objectives and benefits of the study were explained to respondents; they were then allowed to ask any questions for clarifications. Once they consented, they were made to sign with a witness of their choice. Interviews were conducted in places that guaranteed maximum privacy. For individuals who could have emotional problems from interviews, the research assistants were taught to reassurance clients and immediately report to the Principal Investigator (who is a Physician) to manage those concerned. A Clinical Psychologist from the School of Medical Sciences, University of cape Coast was recruited on stand-by to handle cases that were beyond the control of the Principal Investigator.

Data Processing and Analyses

At the end of each interview, field assistants and/or supervisors checked the questionnaires for completeness and consistency before the clients were allowed to leave. Each of the selected health facilities was assigned peculiar codes to allow for easy identification and tracking if errors and inconsistencies were detected. For data entry, a coding manual was developed in order to ensure consistency.

The data were double-entered using EPI-DATA, verified and cleaned. The clean data was exported into STATA (version 11) for analysis. Descriptive analyses were carried out to describe the socio-demographic and socio-economic characteristics, intendedness of pregnancy, knowledge and past use of contraceptives, intention to use and use of PPFP among respondents. Chi square and logistic regression analyses were performed to explore relationships and the influence of socio-demographic and socio-economic characteristics on unintended pregnancy, intention to use and use of family planning postpartum. The influence of socio-cultural factors (sex composition of children) on pregnancy intendedness, intention to use and use of family planning and the relationship between intentions to use postpartum family planning were also explored using logistic regression. Pvalue was set at 0.05 unless otherwise stated.

Two models were used in a bivariate logistic regression analyses to test the association between independent variables and an outcome variable (unintended pregnancy). The significance level threshold was initially set at 0.05 (Model I). However, given so many independent variables in the study, p<0.05 may not be appropriately robust enough to determine which associations were real and which were by chance and so a second model (model II) was introduced. In model II, the significance level threshold was set higher to 0.003 by conducting a Bonferroni correction. Factors found to be significantly associated with the main outcome of interest, were included in a multivariate logistic regression model (Model III), to identify significant independent predictors of unintended pregnancy. Tests of covariance were conducted among all the significant variables and those found to show covariance were dropped from model III. The outcome variable (unintended pregnancy) was defined as any pregnancy that was not wanted at all at the time it occurred or in the future, or mistimed i.e. wanted at a later time but not at the time

it occurred. An intended pregnancy was defined as any pregnancy that was wanted at the time it occurred or wanted at an earlier time but occurred later.

To assess the factors influencing the intention of pregnant women to use postpartum family planning, three logistic regression models were used (Models I, II and III): Models I and II are bivariate and multivariate models respectively that assessed the influence of knowledge and past use of contraceptives and acceptability of family planning by close relations. The overall model (Model III) is a multivariate model which includes all knowledge, past use and acceptability variables which emerged significant (p<0.05) from model III. The factors which emerged significant from Model III after running the regression analysis were considered the predictors of pregnant women's intention to use PPFP.

In order to test the association between sex composition of children ever born and reproductive health outcomes such as unintended pregnancy, intention to use and use of PPFP, sex composition of children was categorized into five groups. Based on the number of children ever born and the sexes of these children the categorization was as follows: zero sons, one or more daughters; zero daughters, one or more sons, daughters and sons , but more daughters than sons; daughters and sons but more sons than daughters; and equal numbers of daughters and sons.

For those who used postpartum contraception, average timing of contraceptive use was calculated with average timing categorized into the following: less than three months, 3-6 months and 7-12 months and over 12 months. The relationship between timing of PPFP use and postpartum behaviours such as breastfeeding, sexual abstinence and amenorrhoea were further assessed using descriptive statistics

Study Limitations, their Mitigation and Quality Assurance

Attrition rate of 47 percent is high and may affect the generalizability of the conclusions drawn from Phase 2 of the study. The attrition could have been worse but for the periodic visits and phone calls made to respondents in the period between Phases 1 and 2.

A possible Hawthorne effect may have occurred in Phase 1. Despite the low rate of past use of contraceptives (<30%), 70 percent expressed intention to use PPFP. To minimize the occurrence of this effect, especially in Phase 2, research assistants and supervisors were trained adequately to ask questions in local languages in order to elicit the right responses. They were also taught to critique the answers provided and to seek further clarifications from respondents to ensure that they had understood the questions posed.

To ensure that selection bias was reduced to the minimum, the research assistants were requested to explain the study objectives and their implications very well to the respondents. Also, the data collectors were made to tally and calculate the proportion of eligible respondents who consented to be recruited into the study. The result of these actions was that a high percentage of respondents agreed to be recruited (97.7%). The total number of eligible respondents who declined recruitment into the study were fifty-two (52). Overall, inferences drawn from Phase 1 may be generalisable to the entire population, but the same cannot be said of Phase 2. Nonetheless, the approach has provided unique data for the study of postpartum family planning intentions.

Summary

A prospective study design was adopted to track the translation of intentions to actual behaviour as opposed to the retrospective approach which relies on recall of past events. Prospective studies have inherent strengths and weaknesses. First, ability to demonstrate clear temporal sequence between exposure and outcome in these studies, make it easier to demonstrate causal associations. Second, a direct computation of incidence rates is permitted for both exposed and unexposed. In this study, incidence of family planning adoption postpartum among pregnant women with intention and without intention to adopt postpartum family planning could be computed. Third, prospective studies provide opportunity for assessing multiple outcomes. Fourth they can be used to study exposures that are relatively uncommon. Weaknesses of prospective studies include potential large sample size requirements, long follow-up periods, and a need to reassess exposure on a frequent basis

Recall bias and reliance on administrative data (which most often have missing and some poorly captured information) are greatly minimized in prospective studies. Data for hypotheses testing about the effects of a wide range of individual, family and community variables was also provided by this study design (Taplin, 2005). Phase 1 involved sampling of study participants from antenatal clinics in health facilities. This made it possible to interview a high percentage of women. Although all the women were given equal chances to be part of the study, losses to follow up experienced in this study could lead to sampling bias, reduced generalizability of study results and increased variance of study estimates. Evidence exist that different factors influence dropout rate. These include low educational status, unemployment, not being married (de Graaf, Bijl, Smit, Ravelli, & Vollebergh, 2000; Bjerkeset, Nordahl, Larsson, Dahl, & Linaker, 2008; Nilsen et al., 2009; Tambs et al., 2009; Torvik, Rognmo, & Tambs, 2012), smoking, high alcohol consumption and physical inactivity (Van Loon, Tijhuis, Picavet, Surtees, & Ormel, 2003; Nilsen et al., 2009; Thygesen, Johansen, Keiding, Giovannucci, & Grønba ek, 2008; Torvik, Rognmo, & Tambs, 2012). Social factors such as support from spouse or friends and characteristics of children, may also affect drop out(Gustavson, von Soest, Karevold, & Røysamb, 2012).

Several strategies to reduce non-response and attrition and improve retention rates have been developed over the years and documented in several studies (Hunt & White, 1997; Brown-Peterside et al., 2001; Coday et al., 2005; Robinson, Dennison, Wayman, Pronovost, & Needham, 2007; Booker, Harding, & Benzeval, 2011). Some of the strategies employed in this study to improve retention rates included the following: first, giving detailed explanations about the study objectives and its possible impacts on the individual, family and society and allaying any anxieties and fears about participating in the study, whilst ensuring that emotional support was on hand to deal with extreme cases; second, obtaining detailed personal information including names, telephone numbers, house addresses (where available) and detailed descriptions of directions to houses of respondents; third, providing adequate motivation for participation by ensuring that the research assistants were friendly, showed respect and courtesy to the respondents and provided adequate privacy at the environment of the interview; fourth, improving rapport between research team and respondents by making periodic contacts with the respondents who agreed to be followed up through telephone calls, home visits and personal contacts; fifth, providing learning opportunities to the research assistants by the comprehensive training given. This motivated the research assistants to engage the respondents in ways that improved rapport and encouraged participation; sixth, providing research assistants in Phase 2 additional incentives for transportation to help them access all the respondents assigned to them, especially to the remotest parts of the municipality and seventh, regular sensitization about the study carried out by the Municipal Health Directorate through their health centres, outreach points, CHPS centres and home visits.

CHAPTER FOUR

SOCIO-DEMOGRAPHIC CHARACTERISTICS AND DETERMINANTS OF UNINTENDED PREGNANCIES

Introduction

Sexual and reproductive health of individuals are influenced by a complex set of factors ranging from sexual and reproductive behaviours, attitudes and societal factors, to biological and genetic factors. Understanding sexual and reproductive health issues also requires deeper appreciation of the underlying social, demographic, economic and cultural characteristics of individuals that make them vulnerable to risks (WHO 2006) or influence reproductive health outcomes. Socio-demographic characteristics have been shown in several studies to influence family planning and reproductive health behaviours and outcomes. One such outcome that has resulted from low family planning use is unintended pregnancy.

Sub-Saharan Africa is known to experience high rates of unintended pregnancies (Goicolea & San Sebastian, 2010). Among the reasons for the situation are contraceptive failure, lack of access to contraception, religious beliefs, and poor knowledge about fertility and pregnancy, a history of previous unintended pregnancy, insufficient reproductive health education, desire for at least two children, parity of five, lack of communication or support within relationships, husband's reluctance to limit family size, and sexual violence

108

(Rosenfeld & Everett, 1996; Glasier et al, , 2006; Amin Shokravi et al, 2009; Goicolea & San Sebastian, 2010).

In Ghana, Omane-Adjepong et al. (2012) identified age, marital status, place of residence, educational status, occupation, gravidity and parity as predictors of unintended pregnancies. Given that one of the goals of family planning is the prevention of unintended pregnancies, understanding the predictors of unintended pregnancies can help to develop strategies to improve family planning uptake (World Health Organization, 2013).This chapter describes the background characteristics of the respondents who were initially recruited into the study, those who expressed intention to be followed up and those who were actually followed up in Phase 2. It also discusses the determinants of unintended pregnancies among the pregnant women.

Socio-Demographic Characteristics of Respondents in Phase 1

Of the 1,914 pregnant women aged 15 - 49 years who were interviewed in Phase 1, mean age was 25 ± 6.5 years, with the highest proportion (29. 7%) in the 20-24 year group. Overall, the number of respondents in each age group decreased with increasing age, reflecting the young age structure of the country. Nearly three-quarters (72.8%) of the respondents interviewed in Phase1 were below 30 years old. All the women were either married, had never married or had ever married. Table 7 shows that 57.2 percent of them were married with 32.4 percent more married under the traditional than the ordinance system.

	Phase 1 respondents	Respondents who Intended to be followed up	Respondents actually followed up in Phase 2
Characteristic	Percent	Percent	Percent
Age			
15-19	17.8	13.2	11.3
20-24	29.7	29.4	25.2
25-29	25.3	28.2	28.0
30-34	15.3	16.3	20.7
35-39	9.0	9.4	11.2
40+	2.9	3.5	3.6
Education			
None	21.6	17.7	8.3
Primary	22.4	22.4	21.4
Middle/JSS	44.1	46.2	54.1
SSS/SHS/Voc	8.7	9.6	11.9
Tertiary	3.2	4.1	4.3
Religion			
Christian	93.2	92.7	92.2
Muslim	4.6	5.1	5.5
Traditionalist	0.5	0.4	0.4
Other	1.7	1.8	1.9

Table 7: Socio-Demographic Characteristics of Respondents

Parity			
0	35.2	33.2	
1 -2	40.3	42.5	59.3
3-4	18.4	19.1	33.0
5+	6.1	5.2	7.7
Ethnicity			
Fante	91.0	91.0	88.8
Other	9.0	9.0	11.2
!Marital status			
Married by ordinance	12.4	11.9	11.2
Married (Traditional)	44.8	46.6	43.2
Engaged	14.7	17.2	20.9
Cohabitation	14.0	13.9	13.3
Divorced/separated	0.4	0.4	0.5
Single	13.7	10.0	10.9
Occupation			
Fishmonger	16.6	13.3	7.0
Farmer	3.5	3.5	3.3
Petty trader	47.8	47.4	56.3
Civil/Public Servant	4.4	5.7	6.1
Student	6.5	5.9	3.2
Other	21.2	24.2	24.1

Table 7 continued

Table 7 continued

!Area of residence			<u> </u>
Saltpond	22.2	27.8	31.6
Biriwa	12.1	9.6	12.5
Anomabo	17.1	12.7	16.8
Mankessim	29.8	29.8	27.8
Other	18.8	20.1	11.3
Total (%)	100.0	100.0	100.0
Total Number	1,914	1,359	1,004

! Difference between total and 1,914 = Missing values; education level=highest level reached irrespective of whether respondent completed level or not.

Source: Field data (2012 and 2014)

About a fifth of the women had attained primary school education, whilst a little over a tenth (11.9%) of them had attained secondary school education or higher. Forty-four percent of the women had attained middle or junior secondary school education and 22% had no formal education.

From Table 7, ninety-one percent belonged to the Fante ethnic group, 93.2 percent were Christians and Muslims were 4.6 percent of the respondents. The main occupations of the respondents were petty trading (47.8%) and fish mongering (16.6%). The average number of children ever born amongst the respondents was 2 ± 2 with the highest proportion of them having up to two children.

Socio-Demographic Characteristics of Follow-up (Phase 2) Respondents

Columns 3 and 4 of Table 7 show the background characteristics of women who expressed intention to be followed up and those who were actually followed up in Phase 2. The total number of women interviewed in Phase 2 was 1,004. These constituted nearly 53 percent of Phase 1 respondents and 74 percent of those who expressed intention to be followed up.

The mean age was 27 ± 6.5 years, with the highest proportion (28%) among the 25-29 year age category. Most (54.1%) of these women had attained Middle/JSS education, were mostly Christians (90.3%) and Fantes (88.8%) respectively. Four out of ten women were married under the traditional system (32% more than those married under the ordinance system) and 51.3 percent were petty traders. The mean number of children ever born among these women was 2.7±2 with a greater proportion having between one to two children.

Wantedness of Current Pregnancy

Pregnant women were asked if the pregnancies they were carrying at the time of the survey were wanted at the time they occurred and at the right time (wanted); wanted but not at the time they occurred (mistimed) or never wanted at the time they occurred or anytime in the future (unwanted). As shown in Table 8, 70 percent of the women indicated that the pregnancies they were carrying were unintended, with 39 percent being mistimed and 31 percent unwanted. Unintended pregnancies were highest in the extreme ages (91% and 80.4%) among those 15-19 and 40 years or older. Reporting of unwanted pregnancy showed a similar



Figure 14: Age and Unintended Pregnancy Source: Field data (2012)

pattern. However, the reporting of mistimed pregnancy was highest among those aged 20 - 24years and lowest among those 40 years and older (Figure 14). Unintended pregnancies were 2.5 times among women with no formal education compared to those with tertiary education (77% versus 31%).



Figure 15: Education and Unintended Pregnancy Source: Field data (2012)

Unwanted pregnancies among women without formal education were 10 times those among women with tertiary education (38% versus 3.2%). Mistimed pregnancy also showed irregular but decreasing trend with increasing level of education (Figure 15).

Among all religious groups, unintended pregnancies were highest among the traditionalists (82%), followed by Muslims (43%) and Catholics (36%) (p<0.001) (Table 8). Expectant mothers with five or more children reported higher levels of unwanted pregnancies (61%) compared to those with up to four or less children. Compared to all other marital arrangements, women who were married under the ordinance had the least unintended (50.4%) and unwanted (17.8%) pregnancies. Of the 256 respondents who were never married, only a tenth of the pregnancies were intended. Similarly, only a third of women who were married under the traditional system, engaged or cohabiting had intended pregnancies.

Unintended pregnancies among students (n=124) was high (90%) compared to women employed in the formal sector as civil/public servants (32% of the 84 respondents). Three out of every four pregnancies among women in the informal sector (petty traders, fishmongers and farmers), were unintended. Intended pregnancies amongst those living in the two semi-urban settlements (Mankessim and Saltpond) were higher than those in the rural areas (Biriwa and Anomabo) (35% versus 20%), (p<0.001) (Table 8).

	0 0		0	verail Pregn	ancy Status	(%)
Demographic Characteristics	Number	Percent of total sample size	Intended (%)	Mistimed (%)	Unwanted (%)	*Unintended (%)
Age**			-		<u></u>	
15-19	340	17.8	9.2	32.3	58.5	90.8
20-24	569	29.7	31.7	45.0	23.4	68.3
25-29	483	25.3	37.7	42.0	20.3	62.3
30-34	291	15.3	38.1	35.1	26.8	61.9
35-39	172	9.0	28.5	36.1	35.5	71.5
40+	56	2.9	19.6	25.0	55.4	80.4
Education level**			÷			
None	414	21.6	23.0	39.0	38.0	77.0
Primary	429	22.4	24.8	39.3	36.0	75. 2
Middle/JSS	843	44.1	30.0	40.4	29.6	70.0
SSS/SHS/VOC	166	8.7	41.2	36.4	22.4	58.8
Tertiary	62	3.2	69.4	27.4	3.2	30.7
Religion**						
Christian	1,783	93.2	29.3	39.0	31.7	70.7
Muslim	88	4.6	39.8	36.4	23.9	60.2
Traditionalist	10	0.5	20.0	60.0	20.0	80.0
Other	33	1.7	22.6	41.9	35.5	77.4

Table 8: Socio-Demographic Characteristics of Women by Pregnancy Status

Gravidity**				<u></u>		
1 -2	1,025	53.8	31.8	36.7	31.5	68.2
3 -4	531	27.9	30.5	46.9	22.6	69.5
5+	348	18.3	21.6	34.2	44.3	78.5
Parity**						
0	673	35.2	30.9	31.1	38.0	69.1
1 -2	772	40.3	34.1	46.3	19.6	65.9
3 -4	353	18.4	23.1	42.2	34.8	76.9
5+	116	6.1	11.2	27.6	61.2	88.8
^Marital Status**						
Married by Ordinance	236	12.4	49.6	32.6	17.8	50.4
Married (Traditional)	857	44.8	29.3	42.1	28.6	70.7
Engaged	282	14.7	38.1	40.2	21.7	61.9
Cohabitation	267	14.0	24.2	45.3	30.6	75.9
Divorced/Separated	8	0.4	0.0	62.5	37.5	100.0
Single	262	13.4	9.6	26.7	63.8	90.4
^Occupation**						
Fishmonger	318	16.6	21.5	38.6	39.9	78.5
Farmer	67	3.5	19.4	43.3	37.3	80.6
Petty trader	913	47.8	28.1	42.3	29.6	71.9
Civil/Public Servant	84	4.4	67.9	23.8	8.3	32.1
Student	124	6.5	9.8	30.9	59.4	90.2
Other	406	21.2	39.0	37.0	24.0	61.0

Table 8 continued

117

.

Table 8 continued

^Area of residence**						<u>_</u>
Saltpond	422	22.2	36.3	38.9	24.9	63.7
Biriwa	231	12.1	21.2	35.9	42.9	78.8
Anomabo	324	17.0	20.1	41.1	38.9	79.9
Mankessim	567	29.8	34.0	36.5	29.5	66.0
Other	358	18.8	28.8	43.6	27. 7	71.2
!Religious Denomination**						
Catholic	199	11.2	35.7	32.16	32.16	64.3
Protestant/Charis/pent	1,361	76.9	30.1	39.75	30.2	70.0
Muslim	81	4.6	43.2	39.51	17.28	56.8
Traditionalist	74	4.2	17.6	48.65	33.78	82.4
No/other religion	54	3.1	22.2	46.3	31.48	77.8
Overall Total	1914	100.0	29.6	39.1	31.4	70.4

*Unintended (mistimed + unwanted), (Pearson Chi2 Statistic - **p<0.001), ^N=1912(difference from total=missing) !N==1769(difference from total=missing), "N=1904(difference from total=missing)

Source: Field data (2012)

Factors Influencing Status of Pregnancy

Unintended pregnancy (outcome variable) was regressed on each of the identified independent variables relating to background characteristics of respondents and partners, characteristics which bother on seriousness of relationships such as living arrangements with partners, partners' relationships with other women and number of years spent in relationship with partners and others including knowledge and past use of family planning methods and previous experience with abortions and miscarriages(Model I). These characteristics have been shown in other studies to have influence on pregnancy status but have not been explored in the study setting. Only those independent variables that were found to be significantly correlated (p<0.05) with the outcome were subjected to Bonferroni's correction (p<0.003) (Model II) (Table 9).

The results of Bonferroni's correction (Model II) showed that women aged 20 years and older had significantly lower odds of having unintended pregnancy (OR 0.83, 95% CI 0.77-0.89) compared to those 15 -19 years. Factors found to be significantly associated with higher odds of unintended pregnancies included: not married under the ordinance system (OR 1.41, 95% CI 1.30-1.52); partner not living in the same house as the woman (OR 2.15, 95% CI 1.70 -2.72) and parity of one or more (OR 1.20, 95% CI 1.12-1.29). Respondents who were aware of modern and traditional family planning methods or had ever used traditional methods, showed significantly lower odds of carrying an unintended pregnancy [OR 0.40, 95%CI(0.25-0.62); OR 0.50, 95% CI(0.40-0.64); OR 0.68, 95%CI(0.55-0.82) respectively]. Education was not found to be a significant factor influencing unintended pregnancies in this study (Table 9).

The variables which emerged significant following bivariate analysis with Bonferroni's correction (p<0.003) were parity, marital status, partner living in same house as woman, awareness of modern and traditional methods of family planning and past use of traditional family planning. These were put in a multivariate logistic regression model and analysed (Table 9, Model III).

.

······································	MODEL		MODEL	MODEL
	I		II	111
Independent Variables	OR(95%CI)	Non- adjusted P-Value	Bonferroni Adjusted P-Value	Adjusted Odds Ratio (AOR)
Age				
Ref:15-19 years	1.00			
\geq 20 years	0.83(0.77-0.89)	<0.001	<0.001	NA
<i>Educational Status</i> Ref: None	1.00			
Primary or more	0.98(0.92-1.03)	0.45	0.450	NA
Ethnicity				
Ref: Fante	1.00			
Others	1.00(0.98-1.02)	0.89	0.890	NA
Religion				
Ref: Christian	1.00			
Non-Christian	1.04(0.99-1.09)	0.15	0.150	NA
Parity				
Ref: 0	1.00			1.00
≥1	1.20(1.12-1.29)	<0.001	<0.001	
1–2				1.48(1.14-1.93)**
3-4				2.64(1.88-3.71)***
5+				6.06(3.24-11.38)***

Table 9: Logistic Regression Analyses of Unintended Pregnancy and Selected Independent Variables: Models I and II (Bivariate) and Model III (Multivariate)

.
Table 9 continued

Marital status				
Ref: ordinance	1.00			1.00
Non -ordinance	1.41(1.30-1.52)	<0.001	<0.001	
Traditional rites				1.81(1.33-2.45)***
Engaged				1.58(1.10-2.28)*
Cohabitation				2.91(1.96-4.31)***
Single				7.32(4.21-12.75)***
Partner Age Ref:15-19 years	1.00			
\geq 20 years	0.97(0.95-0.98)	<0.001	<0.001	NA
Partner Religion				
Ref: Christian	1.00			
Non-Christian	1.25(1.06-1.47)	0.009	0.162	NA
Partner with Children From other women				
Ref: Yes	1.00			
No	1.33(1.08-1.64)	0.008	0.144	NA
Years of marriage				
Ref: <1Year	1.00			
≥1 year	1.02(1.00-1.05)	0.019	0.342	NA

Table 9 continued

Partner Lives in				
same house as				
(Ref: Yes)	1.00			1.00
No	2.15(1.70-2.72)	<0.001	<0.001	1.72(1.28-2.30)***
Partner has other spouses				
Ref: Yes	1.00			
No	1.47(1.06-2.23)	0.019	0.342	NA
Gravidity				
(Ref:1-2)	1.00			
≥2	1.08(1.12-1.14)	0.004	0.072	NA
Previous Abortion or miscarriage				
Ref: Yes	1.00			
No	1.00(1.0-1.0045)	0.043	0.774	NA
Awareness of Modern FP methods				
(Ref: No)	1.00			1.00
Yes	0.40(0.25-0.62)	<0.001	<0.001	0.70(0.42-1.17)
Awareness of Traditional FP methods				
(Ref: No)	1.00			1.00
Yes	0.50(0.40-0.64)	<0.001	<0.001	0.66(0.49-0.89)**

Table 9 continued

Ever use of modern FP method				
Ref: No	1.00			
Yes	0.78(0.64-0.95)	0.014	0.250	NA
Ever use of Traditional FP				
Ref: No	1.00			1.00
Yes	0.68(0.55-0.82)	<0.001	<0.001	0.95(0.75-1.21)
NB: Age of respond of covariance, ***p	dents and partner dro <0.001, **p<0.01, *	pped from property of the pped from provide the pped from	Model III be	cause it failed test

Source: Field data (2012)

Increasing parity was significantly associated with increasing odds of unintended pregnancy. The odd of carrying unintended pregnancy among women with five or more children was four times higher than those with 1-2 children [AOR 6.06, 95% CI (3.24-11.38) versus AOR 1.48, 95%CI (1.14-1.93)].

Women who were married under the traditional system and those unmarried showed significantly higher odds of carrying unintended pregnancy compared to those married by ordinance system (Table 9). Single women showed the highest odds of carrying unintended pregnancy [AOR 7.32, 95% CI (4.21-12.75]. Women not living with their partners exhibited increased odds of having unintended pregnancies compared to women who lived with their partners (AOR 1.72, 95% CI: 1.28 - 2.30). Awareness of traditional methods of family planning (withdrawal and rhythm) was associated with lower odds of having unintended pregnancy compared to non-awareness (AOR 0.66, 95%CI (0.49-0.89).

Summary

Generally, the background characteristics of Phases 1 and 2 respondents were comparable: They were all likely to be below 30 years, had attained middle school/JSS education, be Fante, and likely to be Christian. They were all mostly married under the traditional system and mostly petty traders. The implications of this observation are that the postpartum family planning decisions taken and behaviours expressed by the Phase 2 respondents may be a true reflection of the intentions expressed in Phase 1. Similarly, inferences drawn in Phase 2 may generally be representative of the entire population of respondents.

Studies have shown that reproductive health behaviours and outcomes are influenced by diverse background characteristics of individuals (DeRose, Dodoo, & Patil, 2002; Lehrer, 2004;Grown, Gupta, & Kes, 2005;Kumi-Kyereme, Awusabo-Asare, Biddlecom, & Tanle, 2007; Doctor, Phillips, & Sakeah, 2009).

The age distribution of respondents in this study showed that they were mostly younger and below 30 years. This largely reflected the relatively young age structure of the women who give birth in this country (GSS & Macro, 2009). Young age structure of women who give birth have been shown consistently in studies to be associated with negative reproductive health behaviours and outcomes (GSS & Macro, 2009; Duncan, Edwards, & Alexander, 2010; Jutte et al., 2010) mainly due to barriers to reproductive health services (Warenius et al., 2006; Regmi, Van Teijlingen, Simkhada, & Acharya, 2010)..

Education offers opportunity for people to acquire knowledge and skills and is one of the strongest predictors of reproductive health. More educated women are more likely to have greater knowledge of contraception, use modern methods of contraception, have desired family sizes and have improved spousal communication (Ashton, Giridhar, Holcombe, Madon, & Turner, 2015). This study revealed that women of reproductive age in the municipality had generally low educational status. Closely related to educational attainment is literacy (the ability to read and write) which is an important personal asset. Although not directly measured in this study, the low educational background gives a clue as to the possible low level of literacy in the area.

Evidence from demographic studies has highlighted that health and mortality outcomes for married persons are better than for unmarried persons because married people have more advantages in terms of support for healthy lifestyles, economic resources, and social and psychological support (marriage protection effects); and/or because healthier individuals are more likely to marry and to stay married (marriage selection effects) (Verbrugge, 1979;Waldron, Hughes, & Brooks, 1996;Waldron, Weiss, & Hughes, 1997;Schoenborn, 2004). The finding in this study that the women were mostly married than unmarried creates a window of opportunity for family planning programming. Whilst this opportunity exists, family planning programmme managers need to be mindful of the sociocultural influence on family planning, especially those married under the traditional system.

Petty trading was the main source of income for most of the women in this study. This picture generally reflected that of the 2008 Ghana Demographic and Health Survey which showed that women in the fertile age were mostly employed in the sales and services sector (51.4%) (GSS & Macro, 2009). Evidence exist that women who earn income are empowered, have decision making power in the household, and are more likely to use reproductive health services including family planning (Bloom, Wypij, & Gupta, 2001; Furuta & Salway, 2006; Gill, Pande, & Malhotra, 2007; Malarcher, 2010).

The women in this study were mostly Christians. Religion has been found to be a significant factor in maternal health service utilization. According to Gyimah, Takyi, and Addai (2006), Moslem and Traditional women are less likely to use such services compared with Christians. Designing of strategies in family planning needs to take this into consideration.

This study has attempted to identify the determinants and predictors of unintended pregnancies in the Mfantseman Municipality. Reported unintended pregnancy was high among the women in the municipality, nearly twice the national average of 37% (GSS & Macro, 2009). Although age was excluded from the multivariate analysis because of covariance, the descriptive analysis in this study nevertheless identified **a** trend towards increasing unintended pregnancy with increasing age from 35 years, and is consistent with studies in Nigeria (Okonofua, Odimegwu, Ajabor, Daru, & Johnson, 1999), Iran(Abbasi-Shavazi & Hosseini-Chavoshi, 2004), Nepal(Adhikari, Soonthorndhada, & Prasartkul, 2009) and Tanzania (Exavery et al., 2014). Perceived decreased risk of pregnancy, cessation of contraceptive use because of perceived intolerable side effects and unstable partnerships which negatively influence contraceptive use with increasing age (Godfrey, Chin, Fielding, Fiscella, & Dozier, 2011) have been proffered to explain this finding.

Mistimed pregnancies generally decreased as a woman's age increased. This finding is consistent with those of other studies in Nigeria (Sedgh et al., 2006) and Kenya (Ikamari, Izugbara, & Ochako, 2013), but contradicts studies in Tanzania (Takwimu & Macro, 2011). Reasons advanced to explain the contradiction are that younger women may have sexual intercourse for other reasons apart from childbearing or may have lower knowledge and skills surrounding pregnancy control mechanisms such as contraceptive use compared with their older counterparts (Exavery et al., 2012).

In the logistic regression analysis, education was not found to be a significant factor influencing unintended pregnancy in this study. However, the descriptive analysis revealed that as educational status increased to tertiary level, the proportion of women reporting unintended or unwanted pregnancies decreased. This is consistent with other studies (Bennett, Culhane, McCollum, & Elo, 2006; Akalework, 2008; Omane-Adjepong et al., 2012; Calvert et al., 2013) and emphasizes the point that education empowers women with knowledge about reproductive health issues and enables them to make informed decisions

concerning pregnancy, childbirth and contraception (Yonas, 2005; Malik & Courtney, 2011).

Factors which were identified in the multivariate logistic regression analysis to be significantly associated with unintended the pregnancy included parity, marital status, living arrangement with partner and awareness of traditional methods of contraception. The expectation was that the level of unintended pregnancy would be lower with increasing parity. However, this study showed that high parity was significantly associated with unintended pregnancy and is consistent with other studies (Hamdela, Tilahun, & others, 2012; Omane-Adjepong et al., 2012; Ikamari, Izugbara, & Ochako, 2013). In India, the desire for sons, or not having any son, was associated with an increase in parity (Chaudhuri, 2012). This supports prior research in South East Asia (Das, 1987; Bairagi, 2001; Sathar & Phillips, 2001; Das Gupta et al., 2003). In a southern coastal district such as the Mfantseman Municipality where fishing is one of the main sources of livelihood, the search for sons to help with the fishing business (Akinyoade, 2007) may also be a reason for the increase in unintended pregnancies associated with increasing parity.

This study found that women married under the traditional system and unmarried women had higher odds of unintended pregnancies compared to those married under the ordinance system. The finding of unmarried women having higher odds of unintended pregnancies than married women is consistent with other findings (Korenman, Kaestner, & Joyce, 2001; Bouchard, 2005;Thomson, 2005; Lachance-Grzela & Bouchard, 2009; Calvert et al., 2013; Achana et al., 2015). This observation could possibly be due to the use of pregnancy as a prelude to marriage or for solidifying relationships (Bouchard, 2005) which eventually do not materialize. Evidence exist that pregnancy among unmarried couples may increase commitment to and encourage transition to marital life prior to birth as couples idealize their future childbearing and family life (Kendall et al., 2005; Guzzo & Hayford, 2014). The finding that women who were married under the traditional system had higher odds of unintended pregnancies compared to those married under the ordinance system, presents an issue for further investigation within the Ghanaian context.

High knowledge of family planning methods has been shown to be associated with low likelihood of unintended pregnancies (Adhikari et al., 2009). In this study, women with knowledge of traditional (withdrawal and rhythm) methods of contraception were less likely to have unintended pregnancies compared to those who were not aware. Existing evidence that knowledge about contraceptive methods is a strong predictor of use (Frost, Lindberg, & Finer, 2012) could explain a possible use of these traditional methods among some groups of women in the municipality to prevent unintended pregnancies. Bivariate analysis in this study revealed that women who had ever used traditional methods had significantly lower odds of unintended pregnancy among them and may support this assertion.

This study revealed reporting of high unintended pregnancies among women in the Mfantseman Municipality and increasing unintended pregnancies with increasing age and lower educational status. Predictors of unintended pregnancy found in this study included high parity, marital status (marriage under the traditional system and being unmarried), partners not living together and nonawareness of traditional (rhythm and withdrawal) methods of contraception. As part of family life education, family planning programmes need to consider the promotion of ordinance marriage and the advantages of partners living together. Best practices and opportunities associated with use of traditional family planning need to be explored and promoted among segments of the population that may not want to use modern family planning. Multi-sectoral collaboration, especially with the Gender and Education Ministries to promote female education and empowerment of women may need strengthening. The next Chapter discusses pregnant women's intention to adopt family planning following delivery and within the postpartum period.

CHAPTER FIVE

FACTORS INFLUENCING THE INTENTION OF WOMEN TO USE POSTPARTUM FAMILY PLANNING

Introduction

The periods of pregnancy and immediately after delivery are considered opportune for counseling women on the use of modern FP methods. This is because this period is often associated with a woman's frequent encounter with the health system (Warren et al., 2010). Despite the fact that intentions data in general have been used to predict actual behaviours (Fishbein & Cappella, 2006), very few studies have been carried out on factors influencing intention to use family planning in general and postpartum family planning in particular. Influencing factors have generally been socio-demographic, socio-cultural and socio-economic (Adegbola & Okunowo, 2009; Agha, 2010; Kariuki, 2011; Di Giacomo et al, 2013).

Apart from socio-demographic characteristics of the respondents, this survey explored other factors such as knowledge and past use of various methods and acceptability of PPFP to partners and close relations, This chapter details how women's intention to use PPFP is influenced by these factors.

Knowledge of Family Planning Methods by Socio-demographic Characteristics

Knowledge of available family planning methods is important in influencing individuals' intentions to adopt and eventually use them. Women were asked about their awareness of available contraceptive methods in this study and presented (Table 11). The most known methods among the women were male condom (86.4%), injectables (80.5%), female condom (77.6%) and the pill (77.2%). The least known methods were the foam (20.7%), diaphragm (24.4%) and intrauterine device (IUD) (37.9%).

As revealed in Table 10, women who were aged 15-19 years, had no formal education, had never had children, were single, divorced or separated and were students, were less likely to know about family planning methods. On the other hand, women aged 25-29 years, had tertiary education, had at least a child, were married under the ordinance and were civil/public servants, generally knew about family planning methods.

Past Use of Family Planning Methods

Women's past experiences with family planning methods could influence their subsequent or future choices. In this study women were also asked about their past use of family planning methods and responses summarised in Table 11. Withdrawal was the method most used in the past by women (29.4%), followed by the male condom (26.9%), rhythm (23.7%) and injectables (18%). The IUD (1%), diaphragm (1%), foam (1.2%) and implants (1.6%) were the least used methods in the past. Table 10: Knowledge of Family Planning Methods by Socio-Demographic Characteristics (%)

	PILL	GDI	ĨŊĨ	IMPL	MCon	FCon	Diaph	FOM	Ryth	With	Emeg	Total No. of women
IAge												
15-19	53.2	17.3	51.0	34.0	75.6	63.1	11.9	11.2	43.9	42.6	31.4	312
20-24	77.5	33.7	82.3	61.9	87.9	77.5	21.4	18.1	58.9	64.7	50.3	569
25-29	87.2	48.6	91.1	75.0	93.0	86.6	32.0	28.9	69.8	74.4	57.4	484
30-34	85.3	48.1	89.4	75.8	89.8	82.9	31.7	25.6	61.4	6.99	45.7	293
35-39	86.6	43.0	91.3	69.2	88.4	80.2	23.8	19.8	54.7	61.6	44.2	172
40+	83.9	50.0	91.1	67.9	87.5	80.4	32.1	16.1	50.0	64.3	35.7	56
Education												
None	67.0	27.2	74.8	59.0	78.4	65.8	18.7	14.8	43.0	49.8	37.1	412
Primary	6.69	33.1	76.5	55.9	80.7	6.69	19.6	16.6	48.7	55.7	38.2	429
Middle/JSS	82.0	37.8	83.0	64.1	91.2	82.8	23.3	20.5	63.4	66.3	48.2	843
SSS/SHS/VOC	89.8	60.8	88.0	73.5	93.4	93.4	40.4	28.9	81.9	84.9	6.69	166
Tertiary	96.8	50.0	93.6	91.9	96.8	96.8	69.4	71.0	95.2	91.9	87.1	64

Table 10 continued						ļ						
Parity												
0	66.6	29.60	64.8	47.9	80.4	72.1	21.6	18.7	55.3	56.8	42.1	673
1-2	82.5	43	88.2	71.1	90.4	82.1	26.2	24.0	63.9	68.9	53.5	772
3-4	83.0	45.0	91.2	71.4	89.5	78.2	26.9	19.6	55.5	62.6	45.0	353
5+	85.3	34	87.9	69.0	85.3	77.6	21.6	14.7	47.4	57.8	33.6	116
Religion												
Christian	77.4	38.0	80.9	63.3	86.3	77.3	24.5	20.7	58.1	62.5	46.7	1,783
Muslim	73.9	37.5	73.9	53.4	87.5	81.8	21.6	21.6	63.6	67.1	48.9	88
Traditionalist	90.06	40.0	100.0	100.0	100.0	100.0	40.0	20.0	60.0	70.0	30.0	10
Other	71.0	41.1	71.0	51.6	70.2	77.4	22.6	22.6	58.1	64.5	48.4	33
Marital Status												
Married (Ordinance)	91.1	56.4	87.7	72.9	92.0	90.3	36.9	38.6	67.4	72.9	61.4	236
Married (Traditional)	81.2	38.7	86.8	68.5	87.9	77.4	23.8	19.5	56.5	63.6	44.8	857
Engaged	81.6	46.8	84.8	68.4	92.2	86.2	33.0	24.1	6.69	71.6	56.4	282
Cohabitation	71.5	24.7	79.0	56.2	87.3	73.4	13.5	11.2	57.7	58.1	44.6	267
Divorced/Separated	37.5	12.5	37.5	12.5	100.0	87.5	12.5	12.5	37.5	50.0	12.5	8
Single	53.9	22.3	52.0	37.5	69.5	61.7	16.8	14.5	44.1	46.5	32.0	264

Table 10 continued												-
Occupation												
Fishmonger	68.9	34.9	76.7	63.21	76.7	67.3	23.3	20.4	49.1	55.7	38.4	318
Farmer	85.1	44.8	82.1	64.18	82.1	74.6	29.9	20.9	49.3	49.3	34.3	67
Petty trader	81.5	36.5	85.8	66.16	90.06	80.3	22.8	19.6	58.6	63.5	50.0	913
Civil/Public Servant	97.6	81.0	97.6	91.67	98.8	98.8	70.2	6.7.9	92.9	91.7	84.5	84
Student	54.0	17.7	45.2	32.26	75.0	66.1	16.9	14.5	43.6	48.4	28.2	124
Other	74.4	40.7	70.1	55.9	78.2	70.8	15.2	11.7	59.6	62.5	47.3	406
Area of residence												
Saltpond	77.1	41.0	82.1	60.4	86.8	7.67	25.47	22.2	61.3	71.9	46.0	424
Biriwa	65.7	28.8	71.6	63.6	75.4	63.6	16.53	14.4	40.3	48.3	37.3	236
Anomabo	75.8	28.8	83.7	70.3	90.5	74.5	19.94	15.6	55.5	54.0	50.0	326
Mankessim	82.5	44.6	81.8	61.2	90.06	83.1	29.28	25.8	64.7	69.1	51.5	567
Other	71.2	41.0	75.1	59.9	75.4	78.1	17.7	13.7	57.4	59.6	43.5	361
Total	77.2	37.9	80.5	62.9	86.4	71.6	24.4	20.7	58.3	62.3	46.7	1,914
! N=1886- observed di	fference fr	om 1,914	is due to	missing	values							

Source: Field data (2012)

Contraceptive method used in the past	Percentage	Number
Oral contraceptive Pills	16.1	308
Intrauterine device (IUD)	1.0	20
Injectables	18.0	344
Implants	1.6	30
Male Condom	26.9	516
Female Condom	1.9	36
Diaphragm	1.0	20
Foam	1.2	22
Rhythm	23.7	454
Withdrawal	29.4	564
Emergency contraceptives (EC)	9.2	176

Table 11: Past Use of Family Planning Methods

NB: responses for each method were multiple; percentages are row percentages, Each row total is 1,914

Source: Field data (2012)

Intention to Adopt Postpartum Family Planning

The respondents were asked whether they intended to adopt family planning after delivery. Of those who responded to the question, 70 percent of them (1,326) said they intended to adopt postpartum family planning. Women who were aged 15-19 years, were single, had never had any children, had tertiary education and were civil/public servants expressed significantly the least intention to adopt postpartum family planning (Table 12).

Background	Intention to adopt PPFP (%)	Total No. of respondents n=1902
!Age**		
15-19	59.4	310
20-24	72.0	564
25-29	71.6	482
30-34	74.6	291
35-39	72.1	172
40+	78.2	55
Education**		
None	70.1	408
Primary	74.4	426
Middle/JSS	70.9	839
SSS/SHS/Voc	57.6	165
Tertiary	51.6	63
Parity***		
0	58.3	667
1 -2	74.7	770
3 -4	76.9	350
5+	80.9	115

Table 12: Intention to Use PPFP by Socio-Demographic Characteristics

_

....

Table 12 continued

70.1	1,771
61.4	88
80.0	10
67.7	33
61.9	236
75.5	853
61.9	281
77.4	266
100.0	8
55.7	258
67.2	311
77.6	67
75.2	910
54.8	84
57.7	123
59.5	407
	 70.1 61.4 80.0 67.7 61.9 75.5 61.9 77.4 100.0 55.7 67.2 77.6 75.2 54.8 57.7 59.5

Table 12 continued

Area of residence		
Saltpond	74.4	422
Biriwa	67.5	231
Anomabo	80.5	323
Mankessim	63.3	567
Other	62.3	359
All Total (%)	69.7	100
All Total (N1)	1,326	1,902

*** p< 0.001 N1 =1,902 !N =1,874 (difference from 1902=missing), percentages are row percentages

Source: Field data (2012)

On the other hand, women who were aged forty years and above, had five or more children, had primary education and were divorced or separated, significantly expressed the most intention to adopt postpartum family planning (Table 12).

Preferred Postpartum Family Planning Methods Intended to be Used

The women who intended to use postpartum family planning were asked to rank the family planning methods they would prefer to use after delivering. They were to rank them as first, second or third most preferred methods. Those who were not sure of their choices were also given opportunity to state so. Their responses were presented in Table 13. Injectable was the most preferred, requested by 41% of the women, followed by the pill (25.4%) and the male condom (13.7%). The least preferred methods were the IUD, Emergency contraceptives and sterilization. For these three methods, over 90% of women were not sure of usage.

		Prefer	ed choice	
Intended PPFP method	First	Second	Third	Not sure
	(%)	(%)	(%)	(%)
Pill	17.3	25.4	13.4	43.9
IUD	1.1	4.0	2.2	92.7
Injectables	40.6	22.0	8.3	29.1
Implants	12.4	8.4	9.7	69.5
Male Condom	3.8	8.3	13.7	74.2
Rhythm	2.9	6.9	12.7	77.5
Sterilization	5.1	1.6	2.9	90.4
Lactional Amenorrhoea Method	13.9	8.5	8.4	69.2
Emergency Contraceptive	1.1	1.4	3.4	94.1

Table 13: PPFP Methods Women Intended to Use

Percentages are row percentages; N=1326

Source: Field data (2012)

Knowledge of Family Planning Methods and Intention to Use PPFP

Bivariate descriptive analyses of knowledge of family planning methods and intention to use PPFP was carried out and presented in Table 14. Among the respondents, those who knew most about oral contraceptive pills (56.2%), injectables (59.1%), implants (46.3%) male condoms (62.3%) and female condoms (56.3%) were also the more likely to have intentions to adopt PPFP (p<0.001). However, those who had not heard of the diaphragm (53.9) were significantly more likely to have intentions to adopt PPFP compared to those who had heard. Similarly, majority of the respondents (36.6%) did not know that exclusive breastfeeding is a means of contraception but had intentions to adopt PPFP (p<0.05) (Table 14).

To determine the strength of the associations and the factors that influenced intentions to use PPFP, logistic regression models (Models I and II) on knowledge of contraceptives and intention to use PPFP were fitted (Table 15). Model I (Bivariate analysis) revealed that respondents who had heard about the pill (OR 1.76, 95% CI 1.40-2.20), injectables (OR 2.20, 95% CI 1.73-2.79), implants (OR 1.60, 95%CI 1.30-1.95), male condoms (OR 1.97, 95% CI 1.50-2.59) and female condoms (OR 1.62, 95% CI1.29-2.03) were about two times more likely than those who had not heard about any of those methods to have intentions to use PPFP. Similarly, women who were aware that exclusive breastfeeding (OR 1.83, 95% CI 1.34-2.50) could be a method of contraception were nearly twice more likely to have an intention to use PPFP than those who were not aware.

		Intention	to use PPFP	
		Yes	No	Total
Awareness of method		(%)	(%)	Number
Oral contraceptive	Yes (%)	56.2	21.3	1474
pills***	No (%)	13.5	9.0	428
Intrauterine	Yes (%)	25.8	12.2	722
contraceptive device	No (%)	44.0	18.1	1180
Injectables***	Yes (%)	59.1	21.7	1537
	No (%)	10.6	8.6	365
Implants***	Yes (%)	46.3	16.8	1200
	No (%)	23.4	13.5	702
Male condom***	Yes (%)	62.3	24.5	1650
	No (%)	7.5	5.8	252
Female condom***	Yes (%)	56.2	21.8	1482
	No (%)	13.6	8.5	420
Diaphragm *	Yes (%)	15.8	8.6	464
	No (%)	53.9	21.7	1438
Foaming tablets	Yes (%)	13.6	7.2	395
	No (%)	56.2	23.1	1507

Table 14: Knowledge and Intention to Use PPFP

Table	14	contin	ued
-------	----	--------	-----

Rhythm method	Yes (%)	39.8	18.7	1113
	No (%)	29.9	11.6	789
Withdrawal method	Yes (%)	44.9	18.1	1199
	No (%)	24.8	12.1	703
Emergency	Yes (%)	33.1	13.8	892
contraception	No (%)	36.6	16.5	1010
Exclusive	Yes (%)	29.8	10.8	772
breastfeeding*	No (%)	39.9	19.5	1130
Total Number		1326	576	1902

Significance = *p<0.05, **p<0.01, ***p<0.001

Source: Field data (2012)

On the other hand, those who had heard of the diaphragm (OR 0.74, 95% CI 0.59-0.92) and foaming tablets (OR 0.77, 95%CI 0.61-0.98) were less likely than those who had not heard of those methods to have intentions of using PPFP.

Knowledge of the pill, injectables, implants, male and female condoms, diaphragm and exclusive breastfeeding and foaming tablets were significantly associated with intention to use PPFP in Model I and were put in a multivariate model (Model II) to determine which of them had the most influence on intention to use PPFP.

Independent var	-iable: –		
Knowledge of Contraceptive		Model I	Model II
methods		(Bivariate)	(Multivariate)
Pills	Yes	1.76(1.40-2.20)***	1.16(0.70-1.91)
	No	1.0	1.0
IUD	Yes	0.87 (0.71-1.06)	NA
	No	1.0	
Injectables	Yes	2.20 (1.73-2.79)***	1.69 (1.21-2.36)**
	No	1.0	1.0
Implants	Yes	1.60 (1.30-1.95)***	1.64 (1.11-2.42)*
	No	1.0	1.0
Male condom	Yes	1.97 (1.50-2.59)***	1.18 (0.58-2.41)
	No	1.0	1.0
Female	Yes	1.62 (1.29-2.03)***	1.03 (0.60-1.75)
condom	No	1.0	1.0
Diaphragm	Yes	0.74 (0.59-0.92)***	0.59 (0.38-0.93)**
	No	1.0	1.0
Foaming	Yes	0.77 (0.61-0.98)*	0.84 (0.53-1.32)
Tablets	No	1.0	1.0

Table 15: Logistic Regression of Intention to Use PPFP and Knowledge of FP

Dependent variable: Intention to use PPFP

Rhythm	Yes	0.82 (0.67-1.01)	NA
method	No	1.0	
Withdrawal	Yes	1.21 (0.99-1.48)	NA
method	No	1.0	
Emergency	Yes	1.09(0.89-1.32)	NA
contraception	No	1.0	
Exclusive	Yes	1.83 (1.34-2.50)***	1.56(1.12-2.18)**
breastfeeding	No	1.0	1.0

If a woman had Knowledge about injectables (AOR 1.69, 95%CI 1.21-2.36), implants (AOR 1.64, 95%CI 0.11-2.41), and exclusive breastfeeding (AOR 1.56, 95%CI 1.12-2.18), she was more likely to have an intention to use PPFP than if she did not. On the other hand, if a woman had knowledge of the diaphragm (AOR 0.59, 95%CI 0.38-0.93) she was less likely to have an intention to use PPFP.

Past Use of Family Planning Methods and Intention to Use PPFP

Table 16 shows the descriptive bivariate analysis of past use of family planning methods and intention to use PPFP. Among women who had used any contraceptive methods in the past, those who had used oral contraceptive pills (13.6%) and injectables (15.8%) in the past respectively were about 5 and 7 times more than those who had not used those methods and were significantly more likely to have intentions to use PPFP. Logistic regression models (Models I and II) were constructed to determine the strengths of associations and the factors that influenced intention to use PPFP (Table 17).

In model I (bivariate analysis) women who had used the pill (OR 2.53, 95% CI 1.83-3.49), injectables (OR 3.83, 95%CI 2.70-5.43) and emergency contraceptive pills (ECPs) (OR 1.52, 95%CI 1.05-2.20) in the past had significantly higher odds of having intentions to use PPFP than those who had never used those methods in the past. On the other hand, women with prior experience with the use of the IUDs (OR 0.39, 95%CI 0.16-0.96) had significantly lower odds than those who had never used them, to have intentions to use PPFP.

All the significant factors from model I (past use of the pill, injectables, ECPs and IUDs) were put in a multivariate logistic regression model (Model II). Women who had used injectables (AOR 3.72, 95% CI 2.61-5.30) and oral contraceptive pills (AOR 2.22, 95%CI 1.59-3.11) in the past were more likely to have the intention to use PPFP than those who had never used them in the past. In contrast, women who had used IUDs in the past (AOR 0.13, 95%CI 0.05-0.38) had significantly lower odd of having intentions to use PPFP than those who had never used the intention to use PPFP than those to use PPFP than those who had never used IUDs (Table17)

			· · · · ·	
Past use of contraceptive me	thod	No. (9/)		Total
		Y es (%)	NO (%)	No.
Oral contraceptive pills***	Yes (%)	13.5	2.6	307
	No (%)	56.2	27.7	1595
Intrauterine contraceptive device*	Yes (%)	0.5	0.5	19
	No (%)	69.2	29.8	1883
Injectables***	Yes (%)	15.8	2.2	342
	No (%)	53.9	28.1	1560
Implants	Yes (%)	1.0	0.5	29
	No (%)	68.7	29.8	1873
Male condom	Yes (%)	18.8	8.2	513
	No (%)	50.9	22.1	1389
Female condom	Yes (%)	1.2	0.6	35
	No (%)	68.6	29.6	1867
Diaphragm	Yes (%)	0.6	0.4	19
	No (%)	69.1	29.9	1883
Foaming tablets	Yes (%)	0.7	0.4	21
	No (%)	69.0	29.9	1881
Rhythm (calendar) method	Yes (%)	16.1	7.7	453
	No (%)	53.6	22.6	1449

Table 16: Past Use of Family Planning Methods and Intention to Use PPFP

Intention to use PPFP

Table 16 continued

Withdrawal method	Yes (%)	21.4	8.1	562
	No (%)	48.3	22.1	1340
Emergency contraception	Yes (%)	7.1	2.1	175
	No (%)	62.6	28.2	1727
Total Number		1326	576	1902

Significance = *p<0.05, **p<0.01, ***p<0.001

Source: Field data (2012)

Table 17: Logistic Regression of Intention to Use PPFP and Past Use of FP

		Dependent variable: I	intention to use PPFP		
Independent variables:					
Past use of Contraceptive meth	nods				
		Model I	Model II		
Oral contraceptive pills	Yes	2.53 (1.83-3.49)***	2.22 (1.59-3.11)***		
	No	1.0	1.0		
Intrauterine contraceptive	Yes	0.39 (0.16-0.96)*	0.13 (0.05-0.38)**		
device	No	1.0	1.0		
Injectables	Yes	3.83 (2.70-5.43)***	3.72 (2.61-5.30)***		
	No	1.0	1.0		
Implants	Yes	0.82 (0.38-1.78)	NA		
	No	1.0			

Table 17 continued

Male condom	Yes	1.00 (0.81-1.25)	NA
	No	1.0	
Female condom	Yes	0.73 (0.37-1.46)	NA
	No	1.0	
Diaphragm	Yes	0.74 (0.29-1.71)	NA
	No	1.0	
Foaming tablets	Yes	0.70 (0.29-1.71)	NA
	No	1.0	
Rhythm (calendar) method	Yes	0.87 (0.70-1.10)	NA
	No	1.0	
Withdrawal method	Yes	1.20 (0.97-1.50)	NA
	No	1.0	
Emergency contraception	Yes	1.52 (1.05-2.20)*	1.26 (0.86-1.85)
	No	1.0	1.0
Significance = ***p<0.001,	**p<0.01,	*p<0.05, Ref=No	

Source: Field data (2012)

Acceptability of Family Planning and Intention to Use Postpartum Family Planning

Sixty five percent of the women reported acceptability of family planning and had intention to use PPFP (p<0.001). Higher proportions of them who had intention to use PPFP reported partner's acceptability of family planning (50%) and also reported that they would seek partner's permission before using family planning (59.7%) or will not use family planning without the awareness of their partners (37.4%). Although greater proportions of the women reported that their mothers (37%), mothers-in-law (48%), fathers-in-law, and their religious leaders (45%) did not accept family planning, they still expressed intentions to use postpartum family planning (p<0.001) (Table 18).

To determine the acceptability factors that influenced intention to use PPFP, logistic regression models (Models I and II) were constructed (Table 19). Model I (bivariate analysis) showed that acceptability of family planning by the woman herself (OR 8.55, 95% CI 6.29-11.63), her mother (OR3.30, 95%CI 2.50-4.33), the partner (OR 7.78, 95%CI5.81-10.40), mother -in-law (OR 3.89, 95%CI 2.84 – 5.33), father-in-law (OR 4.19, 95%CI 3.01-5.83) and religious leaders (OR 2.75, 95%CI 2.07 – 3.63) were associated with higher odds of having intention to use PPFP. Furthermore, if a woman would seek partner's permission before using PPFP (OR 2.23, 95% CI 1.75 – 2.85) or would be willing to use PPFP without awareness of partner (OR 3.04, 95%CI 2.41-3.83) she has higher odds of having an intention to use PPFP. When all these significant factors were introduced into Model II and further analysed (Multivariate analysis), woman's acceptability (AOR 3.21, 95%CI 1.64 – 6.26) and partner's acceptability (AOR 3.20, 95%CI 1.94 – 5.48) of family planning were the most significant predictors of intention to use PPFP (Table 19).

The factors which emerged very significant from the multivariate analyses of knowledge of family planning, past use of family planning and acceptability of

		Inte	ntion to us	e PPFP
		Yes	No	Total
Acceptability of family plannir	ng	(%)	(%)	Number
Woman's acceptability***	Yes (%)	65.4	19.6	1616
	No (%)	4.3	10.7	286
Partner's acceptability***	Yes (%)	50.0	10.1	1143
	No (%)	19.8	20.1	759
Acceptability by mother***	Yes (%)	32.9	9.2	800
	No (%)	36.9	21.1	1102
Acceptability by mother-in-	Yes (%)	22.2	5.8	533
law	No (%)	47.5	24.5	1369
Acceptability by father-in-	Yes (%)	19.8	5.2	476
law	No (%)	49.9	25.1	1426
Acceptability by religious	Yes (%)	24.9	8.1	627
leaders	No (%)	44.9	22.2	1275
Partner's permission before	Yes (%)	59.7	22.0	1555
using PP***	No (%)	10.0	8.3	347
Will use FP without partner's	Yes (%)	32.3	6.7	741
awareness	No (%)	37.4	23.6	1161
Total number		1326	576	1902

Table 18: Acceptability of Family Planning and Intention to Use PPFP

Significance = ***p<0.001 **p<0.01 p<0.05

Source: Field data (2012)

		Dependent variable: Intention to use PPFP		
Independent variables:		Model I	Model II	
Acceptability of PPFP		(Bivariate)	(Multivariate)	
woman's acceptability	Yes	8.55 (6.29-11.63)***	3.21 (1.64-6.26)***	
	No	1.0	1.0	
Partner's acceptability	Yes	7.78 (5.81-10.40)***	3.2 (1.94 -5.48)***	
	No	1.0	1.0	
Partner's permission	Yes	2.23 (1.75-2.85)***	1.60 (0.78-3.27)	
before using PPFP	No	1.0	1.0	
Willing to use without	Yes	3.04 (2.41-3.83)***	3.06 (0.71-13.27)	
awareness of partner	No	1.0	1.0	
Acceptability by	Yes	3.30 (2.50-4.33)***	0.56 (0.12-2.65)	
mother	No	1.0	1.0	
Acceptability by	Yes	3.89 (2.84-5.33)***	0.90 (0.43-1.87)	
mother-in-law	No	1.0		
Acceptability by	Yes	4.19 (3.01-5.83)***	0.94 (0.54-1.61)	
father-in-law	No	1.0	1.0	
Acceptability of by	Yes	2.75 (2.07-3.63)***	1.37 (0.90-2.11)	
religious leaders	No	1.0	1.0	

Table 19: Logistic Regression of Intention to Use PPFP and Acceptability ofFamily Planning

Significance = ***p<0.001, **p<0.01, *p<0.05), Ref = No

Source: Field data 2012

PPFP on intention to use PPFP were further analysed using a logistic regression model (Model III) controlling for socio-demographic and socio-economic factors (Table 20)

Following the analysis, the intention of pregnant women to use PPFP were influenced by the partner's acceptability of family planning (AOR 4.42, 95%CI 3.25-6.01), personal conviction of acceptability (AOR 4.69, 95%CI 3.29-6.69) and prior experience with using injectables (AOR 2.74, 95%CI 1.78-4.20) and pills (AOR 1.74, 95%CI 1.21-2.50). Women who had heard of the diaphragm (AOR 0.59, 95%CI 0.44-0.80) as a contraceptive method or had used IUDs in the past (AOR 0.22, 95%CI 0.06-0.76) were significantly less likely to want to use PPFP. Women who were cohabiting at the time of the survey exhibited higher odds of intending to use PPFP compared to those who were married by ordinance (AOR 1.96, 95% CI 1.13-3.39) (Table 20).

Summary

Seventy percent (70%) of women covered in this study expressed the intention to adopt PPFP. This is higher than reported intention among married family planning non-users in 2008 in Ghana of up to 53 percent (GSS & Macro, 2009). The contrast may be explained by the fact that the respondents in this study were pregnant women who may reasonably be expected to want to delay a subsequent pregnancy. Nevertheless, pregnant women need to be made targets for PPFP interventions.

.

Knowledge, Past Use and Acceptability Variables				
MODEL III				
Dependent variable: Inter	ntion to use PPFP			
Independent variables:				
Most significant variables	AOR (95%C.I)			
Age				
Ref: 15-19	1.00			
20-24	1.24(0.83 -1.85)			
25-29	1.08(0.67-1.71)			
30-34	0.93(0.52-1.65)			
35-39	0.72(0.38-1.39)			
40+	0.99(0.40-2.46)			
Parity				
Ref: 0	1.00			
1 -2	1.29(0.93-1.80)			
3 -4	1.22(0.73-2.01)			
5+	2.17(0.99-4.73)			
Education				
Ref: None	1.00			
Primary	1.18(0.81-1.71)			
Middle/ JSS	0.90(0.63-1.28)			
SSS/SHS/Vocational	0.67(0.39-1.16			
Tertiary	0.71(0.29-1.70)			

Table 20: Logistic Regression of Intention to Use PPFP and Most Significant

.

Table 20 continued

Occupation		
Ref: Fishmonger		1.00
Farmer		1.40(0.66-2.98)
Petty trader		1.35(0.93-1.95)
Civil/Public Servant		0.78(0.34-1.80)
Student		0.97(0.54-1.74)
Other		0.54(0.30-0.96)*
Marital status		
Ref: Married –Ordinance		1.00
Married(Traditional)		1.23(0.78-1.93)
Engaged		0.91(0.56-1.50)
Cohabitation		1.96(1.13-3.39)*
Single		1.71(0.98-2.99)
Knowledge of contraceptives		
Injectables	Ref: No	1.00
	Yes	1.21(0.84-1.75)
Implants	Ref: No	1.00
	Yes	1.30(0.95-1.77)
Diaphragm	Ref: No	1.00
	Yes	0.59(0.44-0.80)**
Exclusive Breastfeeding	Ref: No	1.00
	Yes	1.15(0.78-1.68)

Table 20 continued

Past use of contraceptives		
Oral Contraceptive Pills	Ref: No	1.00
	Yes	1.74(1.21-2.50)**
IUD	Ref: No	1.00
	Yes	0.22(0.06-0.76)*
Injectable	Ref: No	1.00
	Yes	2.74(1.78-4.20)***
Acceptability of FP		
By woman	Ref: No	1.00
	Yes	4.69(3.29-6.69)***
By Partner	Ref: No	1.00
	Yes	4.42(3.25-6.01)***

Significance = ***p<0.001, **p<0.01, *p<0.05

Source: Field data (2012)

Although less than 30 percent of the women had used contraceptives in the past, 85 percent of them considered family planning acceptable and 65 percent had intentions to use postpartum family planning. This picture may point to a favourable basis for promotion of PPFP. However, the existence of the gap between intention and actual use of family planning (Ross and Winfrey, 2001, DeRose et al, 2004; Callahan and Becker, 2014) needs to be born in mind. On this basis, it appears there is a disparity between stated intentions and PPFP use in this study. The low past use of contraceptives (Table 12) despite high levels of
contraceptive knowledge (Table 11) and the fact that 24 percent of the women considered family planning acceptable but did not have any intention to adopt PPFP (Table 18) affirm the disparity. Phase 2 of this study was influenced by these concerns and would bring out the gaps if any between FP intentions and actual uptake.

Women in this study reported withdrawal method as the most used in the past, followed by the male condom, rhythm and the injectable. The fact that withdrawal and rhythm together were the most used methods in the past may be explained by the existence of some perceived benefits offered by their use, some perceived side effects of modern contraceptives, entrenched traditional and sociocultural influences on the people and poor penetration of family planning education on modern methods in the Municipal area (Annual report, Mfantseman Municipal Health Directorate, 2010). The finding that the injectable was the preferred postpartum contraceptive method among the women who intended to use PPFP may have been influenced significantly by past use of injectables (Tables 14 and 20). This is consistent with a recent review of the uptake of various FP methods in Sub-Saharan Africa and the 2008 Ghana Demographic and Health Survey (Ross & Winfrey, 2001; GSS & Macro, 2009). The preference for injectables may also be explained by fact that some women are able to conceal their family planning status when they use them (Biddlecom & Fapohunda, 1998; Tensou, Hailemariam & Reniers, 2008.) Factors that make women in Ghana prefer injectables would most likely provide indications of the interventions that would be required to increase the uptake of PPFP in general.

The finding that pregnant cohabiting women showed significantly higher odds of intention to use PPFP than women who were married by ordinance, could be explained by the fact that their cohabiting status does not guarantee them security compared to that offered by ordinance marriage. This is in contrast with the finding by Kariuki (2011), in which being married was a significant predictor of intention to use postpartum contraceptives. There is also the other argument that cohabiting women may not have intention to use FP so that a pregnancy could be a reason for marriage(Cuff & Francis, 1978).

The study identified women's acceptability of family planning (personal conviction) as an important factor influencing pregnant women's intention to use PPFP. Further evidence from the study, however, suggests that personal conviction is insufficient to ensure actual uptake of PPFP by the women. This is because, the women in the study made the point about their need for partner approval before they could use a method of PPFP. This points to the role played by male partners in FP decision-making processes among women (Bawah, 2002; Gebreselassie & Mishra, 2007; Crissman, Adanu, & Harlow, 2012).

Improving spousal communication could be considered an integral component of interventions to increase male involvement in family planning (Hartmann, Gilles, Shattuck, Kerner, & Guest, 2012). For instance, studies in Uganda and South Africa have shown that a simple intervention such as written letter of invitation to a male partner can lead to significant increase in male attendance at antenatal clinics and opportunities for couple counseling (Rujumba

et al., 2012; Bhalla et al., 2012). The next chapter discusses women's translation of their intention to adopt PPFP into actual use and the associated factors.

CHAPTER SIX

POSTPARTUM CONTRACEPTIVE USE: TRANSLATING INTENTION INTO BEHAVIOUR

Introduction

Empirical evidence over the past three decades has led to the recognition that specific behaviours can be predicted with considerable accuracy by assessing intentions to engage in the behaviours under consideration (Fishbein, 2008). According to Piotrow et al.,(1997), individuals and groups progress from knowledge to sustained behaviour change and advocacy.

Some other empirical evidence suggests that intentions do not always lead to behaviour as postulated by the Theory of Planned Behaviour (Gollwitzer, 1999). Despite the strong intention-behaviour relationship observed in some longitudinal studies (Callahan & Becker, 2014), the relationship is weak for some experimental studies, in some studies of intentions measured further away in time from behaviours, risky behaviours performed within a social context and behaviours influenced by habits (Webb & Sheeran, 2006).

This chapter discusses the extent to which prenatal intentions of pregnant women were translated into actual behavior (contraceptive use) in the postpartum period, the timing of postpartum contraceptive use, the preferred methods and determinants of contraceptive use.

Translation of Prenatal Intentions into Actual PPFP Use

Seventy percent of the 1,914 women in Phase 1 expressed intention to use PPFP. These women were followed up in Phase 2. Of the 1,004 women who were eventually followed up in Phase 2, 75 percent expressed prenatal intention to use PPFP. Among the women in Phase 1, those who expressed prenatal intentions to use PPFP were mostly 40 years or more, had primary school education, had five or more children, were traditionalist or Christian, were cohabiting, divorced or separated, were farmers and resident in Anomabo. Apart from being mostly resident in Biriwa, and having 3-4 children, the Phase 2 women who expressed prenatal intentions to adopt PPFP were similar to those in Phase 1 (Table 21, Columns 2 and 3).

Fifty seven percent from Phase 1 who expressed intention to use FP also expressed intention to use in Phase 2. Over 80 percent were among those 30 years and above, a third among those 20-24 years. Furthermore, 85 percent had tertiary education and a quarter with no education. They were also mostly Muslim (98%), single (73%) and with parity of three or more (98%).

Of the women in Phase 2 who said they would use FP after delivering, 56.5 percent reported using PPFP (Table 21). These women were more likely to be thirty years or more (over 64%), attained middle school/JSS education (63.2%), be Muslim (60.4%), married under the ordinance system (80.2%), be civil/public servants and from Anomabo (76.5%). Although a lower percentage was recorded for those who reported using PPFP among Phase 1 women who expressed

intention to use FP (32.1%), their characteristics were generally similar to those using PPFP from Phase 2.

As shown in Table 22, 75 percent of the 1,003 respondents in Phase 2 expressed prenatal intention to use PPFP. Of these, 42 percent reported using PPFP whilst the remaining 33 percent had not used it. Furthermore, 25 percent of the women in Phase 2 expressed no prenatal intention to use family planning after delivering but 8 percent reported using it, whilst 17% reported not using FP. These findings were significant ($\chi^2 = 45.7$; P<0.001) implying that the prenatal intentions of some of the women were inconsistent with their postpartum behaviours. This observation notwithstanding, the odds of using PPFP given that a woman had a prenatal intention to do so, was nearly three times (2.78) that of not using PPFP given that a woman did not have a prenatal intention to do so (Table 22).

The characteristics of the women in Phase 2 were further analysed based on their prenatal FP intentions and PPFP outcomes. A third (33%) of the women had prenatal family planning intentions but had not used PPFP after delivering. Higher proportions were between 25 and 29 years (38.4%), had attained up to primary education (44%) and were engaged, divorced or separated. These were likely to be farmers (70%), have one to two children and be neither Christian, Muslim nor Traditionalists (Table 23). Very few women (8%) had no intention to use PPFP when they were pregnant but went on to use FP after delivering. These were likely to be in the 35-39 year group, be Muslim, have attained Middle school/JSS education, and be single and to have had 3-4 children.

Background	Phase 1	Phase 2	Proportion	Use of	Use of	Use of	Use of
)	respondents	women	of Phase 1	PPFP	PPFP	PPFP	PPFP
	who	orhw	who	(Irrespective	(from	from	(from
	intended to	intended	expressed	of intention	intention	intention	intention
	use PPFP	to use	intention	status)	expressed	expressed	expressed
		PPFP	in Phase 2		in Phase	in Phase	in Phase
	(a)	(q)	(b/a)		2)	1	2)
	No. (%)	No. (%)	%	No.	No.	%	%
Age							
15-19	184 (59.4)	74(82.2)	40.2	48	41	22.2	55.4
20-24	406 (72.0)	133(69.3)	32.7	75	66	16.3	49.6
25-29	345 (71.6)	204 (72.6)	59.1	121	96	27.8	1.71
30-34	217 (74.6)	203 (75.6)	93.5	147	130	59.9	54.0
35-39	124 (72.1)	100 (75.8)	80.6	87	99	53.2 6	6.0
40+	43 (78.2)	38 (95.0)	88.4	27	26 (60.5 6	8.4

Table 21: Prenatal Intention and Actual PPFP Use

None	286 (70.1)	68 (86.5)	23.8	36	33	11.5	48.5
Primary	317(74.4)	185 (86.5)	58.4	102	92	29.0	49.7
Middle/JSS	595(70.9)	399(73.5)	67.1	306	252	42.4	63.2
SSS/SHS/Voc	95(57.6)	72 (60.5)	75.8	44	34	35.8	47.2
Tertiary	33(51.6)	28 (63.6)	84.8	17	14	42.4	50.0
Parity							
0	389 (58.3)	i			,	,	ų,
1-2	575 (74.7)	398 (72.5)	69.2	205	175	30.4	44.0
3-4	269 (76.9)	264 (79.8)	98.1	193	150	55.8	56.8
5+	93(80.9)	90 (75.6)	96.8	107	100	107.5	111.0
Religion							
Christian	1,242(70.1)	680 (75.1)	54.8	459	386	31.1	56.8
Muslim	54 (61.4)	53 (69.7)	98.1	39	32	59.3	60.4
Traditionalist	8 (80.0)	2 (50.0)	25.0	1	1	12.5	50.0
Other	22 (67.7)	17 (100.0)	77.2	6	6	27.3	35.3

164

Education

Married(Ordinance)	146 (61.9)	101 (39.9)	69.2	66	81	55.5	80.2
Married(Traditional)	644 (75.5)	367 (84.5)	57.0	218	181	28.1	49.3
Engaged	174 (61.9)	103(42.9)	59.2	94	75	43.1	72.8
Cohabitation	206 (77.4)	68 (93.2)	33.0	46	44	21.4	64.7
Divorced/Sep	8 (100.0)	5 (100)	62.5	1	1	12.5	20.0
Single	148 (55.7)	108(71.1)	73.0	47	43	29.1	39.8
Oocupation							
Fishmonger	209(67.2)	53 (76.8)	25.4	36	28	13.4	52.8
Farmer	52(77.6)	30 (90.9)	57.7	7	7	13.5	23.3
Petty trader	685(75.2)	431(83.7)	62.9	307	266	38.8	61.7
Civil/Public Servant	46 (54.8)	43 (38.7)	93.5	40	31	67.4	72.1
Student	71(57.7)	20 (62.5)	28.2	18	14	19.7	70.0
Other	263 (59.5)	175 (72.0)	66.5	16	79	30.0	45.1

Marital Status

Area of residence							
Saltpond	314 (74.4)	221 (69.9)	70.4	157	124	39.5	56.1
Biriwa	156 (67.5)	121 (89.6)	77.6	60	56	35.9	46.3
Anomabo	260 (80.5)	153 (85.5)	58.9	130	117	45.0	76.5
Mankessim	359 (63.3)	181(69.9)	50.4	107	84	23.4	46.4
Other	237 (62.3)	76 (66.7)	32.1	51	44	18.6	57.9
Total	1,326 (69.7)	752 (75.0)	56.7	505	425	32.1	56.5
Overall Total	1,902 *	1,003!					

*1914-1902=Missing values; ! 1004-1003=Missing value; Percentages in columns a and b are row percentages, NB: Total for Age not equal to 1326 because of missing values.

Source: Field data (2012 & 2014)

	· · · · · · · · · · · · · · · · · · ·	Actual us	e of PPFP		Odds	95%CI
					Ratio	
		Yes (%)	No (%)	Total	2.78	(2.07-3.74)***
Intention	Yes	42.4	32.6	752		
use PPFP	No	8.0	17.0	251		
	Total	505	498	1,003		

Table 22: Association of Prenatal Intention and PPFP Use

***p<0.001, Pearson Chi Square= 45.7 at p<0.001

Source: Field data (2014)

Seventeen percent of the women had no intention to use PPFP whilst pregnant and still had not used it. These were mostly younger (20-24 years), had attained tertiary education and were married under the traditional system. They were also likely to belong to the traditional religion and had had 1-2 children (Table 23).

	Had intention and used PPFP	Had intention and did not use PPFP	Had no intention but used PPFP	Had no intention and did not use PPFP	Total
Age	$\chi^2 = 63.4^{**}$	**			
15-19	45.6	36.7	7.7	10.0	90
20-24	34.4	34.9	4.7	26.0	271
25-29	34.2	38.4	8.9	18.5	281
30-34	48.5	27.2	6.3	17.9	210
35-39	50.0	25.8	15.9	8.3	111
40+	65.0	30.0	2.5	2.5	40
Education	$\chi^2 = 58.2*$	**			
none	39.8	42.2	3.6	14.5	133
Primary	43.0	43.5	4.7	8.9	214
Middle/JSS	46.4	27.1	9.9	16.6	543
SSS/SHS/Voc.	28.6	31.9	8.4	31.1	83
Tertiary	31.8	31.8	6.8	29.6	30
Religion	$\chi^2 = 14.5!$,			
Christian	42.6	32.5	8.1	16.9	934
Muslim	42.1	27.6	9.2	21.1	54
Traditionalist	25.0	25.0	0.0	50.0	4
Other	35.3	64.7	0.0	0.0	11

Table 23: Prenatal FP Intentions and PPFP Outcomes of Women

Occupation	$\chi^2 = 69.4^{***}$	*			
Fishmonger	40.6	36.2	11.6	11.6	136
Farmer	21.2	69.7	0.0	9.1	33
Petty trader	51.7	25.1	8.0	15.3	5 15
Civil/Public servant	27.9	43.2	8.1	20.7	44
Student	43.8	18.8	12.5	25.0	32
Other	32.5	39.5	7.4	20.6	243
Parity	$\chi^2 = 171.4^{\circ}$	***			
0	-	-	-	-	-
12	31.9	36.8	5.3	26.1	553
34	45.3	34.4	13.0	7.3	331
5+	84.0	6.7	5.9	3.4	119
Marital status	$\chi^2 = 57.8$				
Married(Ordinance)	52.1	20.7	8.5	18.8	132
Married	41.7	30.4	8.5	19.4	434
(Traditional)					
Engaged	31.3	45.4	7.9	15.4	140
Cohabitation	60.3	32.9	2.7	4.1	173
Divorced/Sep	20.0	80.0	0.0	0.0	5
Single	34.2	36.8	10.5	18.4	119

Table	23	continu	led
-------	----	---------	-----

Residence	$\chi^2 = 86.7$	***			
Saltpond	39.2	30.7	10.4	19.6	296
Biwira	41.5	48.2	3.0	7.4	135
Anomabo	65.4	20.1	7.3	7.3	199
Mankessim	32.4	37.5	8.9	21.2	259
Other	38.6	28.1	6.1	27.2	114
Total (%)	42.4	32.6	8.0	17.1	100
Total (Number)	425	327	80	171	1,003

***P<0.001, !P>0.05; Percentages are row percentages

Source: Field data (2014)

Timing and Preferred Postpartum Contraceptive Methods

The mean time of adoption was 3.5 ± 2.7 months after delivery. Among the women who had used PPFP, sixty-six percent had used a method within the first three months after birth, with the frequency of use decreasing towards the end of the extended postpartum period and beyond (Table 24).

The most used PPFP methods were the male condoms (33.7%), followed by the injectables (30.3%) and the pills (20.6%) as reported in Table 25. This was largely consistent with their prenatal PPFP preference where the most preferred methods were injectables, pills and Lactational amenorrhoea method (LAM).

Time of PPFP use (Months)	Frequency	Percent	
0-3	276	65.4	
4-6	96	22.8	
7-9	35	8.3	
10-12	12	2.8	
>12	3	0.7	
Total*	422	100	
*======================================			

Table 24: Timing of PPFP Use Following Delivery

*missing values (505-422)

Source: Field data (2014)

Table 25: Postpartum FP Methods Used by Women Following Birth

Methods used	Frequency	Percent	
Pills	109	20.6	
Injectables	160	30.3	
Implants	32	6.1	
Male condom	179	33.7	
Female sterilization	14	2.7	
LAM	11	2.1	
Periodic abstinence	13	2.4	
Withdrawal	11	2.1	
Total*	529	100	

*Deviation from total who used PPFP methods due to some respondents using a combination of some methods Source: Field data (2014) Suitability of method to client (83%) was the main driving force behind choices of particular PPFP methods by the women (Table 26).

Reasons for choice of method	Frequency	Percent	
It is suitable for me	435	82.8	
It has fewer side			
effects	19	3.6	
My partner endorsed it	25	4.8	
provider chose it for me	46	8.8	
Total*	525	100	

and and a set of the s	Table 26: Reasons for Method Choice
--	-------------------------------------

*Deviation from total who used PPFP due to multiple responses Source: Field data (2014)

Predictors of Actual Postpartum Family Planning Use

Women's uptake of postpartum family planning may be influenced by facility of delivery, pregnancy outcomes, postpartum contraceptive behaviours (breastfeeding, amenorrhoea and sexual abstinence), fertility preferences, postpartum spousal discussion, advocacy about family planning, PPFP intention and background characteristics. Information about these variables were ascertained and analysed using logistic regression models to determine which of them strongly predicted women's postpartum family planning uptake. Three multiple logistic regression models were constructed using PPFP use as the dependent variable in all cases. Model 1 used PPFP intention and background characteristics as independent variables, whilst Model 2 used PPFP intention, facility of delivery, pregnancy outcomes, postpartum contraceptive behaviours (breastfeeding, amenorrhoea and sexual abstinence), postpartum spousal discussion and advocacy about family planning as independent variables. The most significant factors that emerged (p<0.05) from models 1 and 2 were selected and used as independent variables in Model 3 to determine the predictors of PPFP.

In Model 1, prenatal (postpartum family planning) intention was a strong predictor of postpartum family planning use. A pregnant woman with an intention to use PPFP, was 3.34 times more likely to use family planning in the postpartum period compared to one who did not have an intention to [AOR¹ 3.34] 95%C.I(2.09-5.35)]. Education, occupation, parity and preferred number of children also influenced PPFP use: Women with tertiary education were 5 times more likely than those without any formal education to use PPFP [AOR¹ 5.39, 95%C.I (1.20-24.25)]. For occupation, farmers were less likely [AOR¹ 0.04, 95%C.I (0.004-0.28)] whilst petty traders were more likely [AOR¹ 2.31, 95% C.I (1.01-5.25)] than fishmongers respectively to use PPFP. Women with two children or more had higher odds of using PPFP compared to those who had two children. The odds was about 13-times higher for grand-multiparous (5+children) women [AOR¹ 53.48, 95% C.I (20.48-139.64)] compared to those with 3-4 children [AOR¹ 4.18, 95% C.I (2.45-7.11)]. Also with preferred number of children, women who preferred five children or more were 13 times more likely to adopt those who preferred two or children (Table 27) less PPFP than

ndependent variables	Depend	dent variable: Postpartum contracej	ptive use
	MODEL 1	MODEL 2	MODEL 3
	OR(95%CI)	OR (95%CI)	OR(95%CI)
oppp intention			
kef: No	1.00	1.00	1.00
Yes	3.34(2.09-5.35)***	2.85(2.06-3.96)	2.98(1.86-4.76)**
Age (years)		NA	
Ref: 15-19	1.00		
20-24	1.09(0.31-3.81)		
25-29	1.06(0.30-3.73)		
30-34	0.83(0.23-3.05)		
35-39	1.67(0.41-6.78)		
40+	1.08(0.26-4.55)		

Education		NA	•
Ref: None	1.00		
Primary	1.61(0.65-4.01)		1.74(0.72-4.25)
Middle/JSS	1.85(0.79-4.31)		2.11(0.93-4.79)
SSS/SHS/Vocation	1.97(0.72-5.39)		2.28(0.85-6.07)
Tertiary	5.39(1.20-24.25)*		5.17(1.24-21.50)*
Religion		NA	
Ref: Christian	1.00		
Muslim	0.77(0.38-1.57)		
Traditionalist	ł		
Other	0.26(0.03-2.11)		

Decupation NA tef Fishmonger 1.00 tef Fishmonger 0.04(0.004-0.28)*** "armer 1.11(0.35-3.49) Civil/Public servant 1.11(0.35-3.49) Student 4.33(0.96-19.51) Student 1.11(0.35-3.49) Student 1.11(0.35-3.49) Student 1.17(0.74-4.21) Other 1.77(0.74-4.21) Refibrond 1.00 Ref. Saltpond 1.00 Biritwa 1.04(0.45-2.39) Anomabo 1.54(0.80-2.97) Mankessim 0.71(0.42-1.18)				
lef: Fishmonger 1.00 armer 0.04(0.004-0.28)*** 0.05(0.01-0.28)*** etty trader 2.31(1.01-5.25) 0.05(0.01-0.28)*** etty trader 2.31(1.01-5.25) 2.28(1.01-5.25) SivilPublic servant 1.11(0.35-3.49) 2.28(1.01-5.25) SivilPublic servant 1.11(0.35-3.49) 2.28(1.01-5.25) SivilPublic servant 1.11(0.35-3.49) 2.28(1.01-5.25) Sindent 4.33(0.96-19.51) 3.30(0.77-1.3.24) Other 1.77(0.74-4.21) 3.30(0.77-1.3.24) Other 1.77(0.74-4.21) 3.30(0.77-1.2.24) Other 1.77(0.74-2.39) NA Ref Saltpond 1.00 1.00 Biriwa 1.04(0.45-2.39) NA Anomabo 1.54(0.80-2.97) Markessim 0.71(0.42-1.18)	ocupation		NA	•
tarmer0.04(0.004-0.28)***0.05(0.01-0.etty trader2.31(1.01-5.25)2.28(1.01-5Civil/Public servant1.11(0.35-3.49)2.28(1.01-5Student4.33(0.96-19.51)1.24(0.42-3Student4.33(0.96-19.51)3.30(0.77-1-Other1.77(0.74-4.21)NA2.01(0.85-4ResidenceNANA2.01(0.85-4Ref: Saltpond1.00NA2.01(0.85-4Biriwa1.04(0.45-2.39)NA-Anomabo1.54(0.80-2.97)0.71(0.42-1.18)Markessim0.71(0.42-1.18)0.71(0.42-1.18)	tef: Fishmonger	1.00		
etty trader 2.31(1.01-5.25) 2.28(1.01-5 Civil/Public servant 1.11(0.35-3.49) 1.24(0.42-3 Student 4.33(0.96-19.51) 3.30(0.77-1) Student 1.77(0.74-4.21) 3.30(0.77-1) Other 1.77(0.74-4.21) NA Residence NA 2.01(0.85-4 Ref. Saltpond 1.00 1.00 Biriwa 1.04(0.45-2.39) - Anomabo 0.71(0.42-1.18) -	armer	0.04(0.004-0.28)***		0.05(0.01-0.41)***
Civil/Public servant 1.11(0.35-3.49) 1.24(0.42-3 Student 4.33(0.96-19.51) 3.30(0.77-1- Student 1.77(0.74-4.21) 3.30(0.77-1- Other 1.77(0.74-4.21) 3.30(0.77-1- Other 1.77(0.74-4.21) 3.30(0.77-1- Residence 1.77(0.74-4.21) 2.01(0.85-4) Ref: Saltpond 1.00 - Biriwa 1.00 - Mankessim 0.71(0.45-2.39) -	Petty trader	2.31(1.01-5.25)		2.28(1.01-5.18)*
Student 4.33(0.96-19.51) 3.30(0.77-1- Other 1.77(0.74-4.21) 3.30(0.77-1- Other 1.77(0.74-4.21) 2.01(0.85-4 Residence NA 2.01(0.85-4 Ref: Saltpond 1.00 NA Biriwa 1.04(0.45-2.39) - Anomabo 1.54(0.80-2.97) 0.71(0.42-1.18)	Civil/Public servant	1.11(0.35-3.49)		1.24(0.42-3.72)
Other 1.77(0.74-4.21) 2.01(0.85-4 Residence NA 2.01(0.85-4 Ref: Saltpond 1.00 - Ref: Saltpond 1.00 - Biriwa 1.04(0.45-2.39) - Anomabo 1.54(0.80-2.97) Mankessim 0.71(0.42-1.18)	Student	4.33(0.96-19.51)		3.30(0.77-14.09)
Residence NA Ref: Saltpond 1.00 Biriwa 1.04(0.45-2.39) Anomabo 1.54(0.80-2.97) Mankessim 0.71(0.42-1.18)	Other	1.77(0.74-4.21)		2.01(0.85-4.79)
Ref: Saltpond 1.00 Biriwa 1.04(0.45-2.39) Anomabo 1.54(0.80-2.97) Mankessim 0.71(0.42-1.18)	Residence		NA	
Biriwa 1.04(0.45-2.39) Anomabo 1.54(0.80-2.97) Mankessim 0.71(0.42-1.18)	Ref. Saltpond	1.00		•
Anomabo 1.54(0.80-2.97) Mankessim 0.71(0.42-1.18)	Biriwa	1.04(0.45-2.39)		
Mankessim 0.71(0.42-1.18)	Anomabo	1.54(0.80-2.97)		
	Mankessim	0.71(0.42-1.18)		

Table 27 continued

farital status		NA	
tef: Married (Ordinance)	1.00		
Married (Traditional)	1.19(0.68-2.09)		
Engaged	0.85(0.43-1.68)		
Co-habitation	0.91(0.39-2.15)		
Divorced/Separated			
Single	0.38(0.09-1.61)		
Parity		NA	
Ref:0-2	1.00		1.00
3-4	4.18(2.45-7.11)***		4.26(2.67-6.80)***
5+	53.48(20.48-139.64)***		52.25(22.08123.64)***

** ** 1.00 1.00 1.00 0.65(0.39-1.10)
* * *

Return of menses		
Ref: Yes	1.00	
No	0.99(0.59-1.69)	
Pregnancy outcome		
Ref: Live	1.00	
Stillbirth	1	
Miscarriage	1.16(0.38-3.57)	
Spousal discussion		
Ref: Yes	1.00	1.00
No	0.46(0.32-0.66)***	0.76(0.45-1.28)
Sort PPFP information		
Ref:Yes	1.00	
No	1.44(1.01-2.07)*	0.91(0.54-1.55)

Advocate about PPFP		
Ref. Yes	1.00	
No	0.55(0.38 - 0.81)***	0.38(0.22-0.65)***
Facility of delivery Ref: Home	1.00	
Public Health Facility	0.97(0.43-2.19)	
Private Health Facility		
Maternity Home	4	

***p<0.001, ** p<0.01, *p<0.05

Source: Field data (2014)

Model 2 indicates that women who had prenatal PPFP intentions were 2.85 times more likely to use PPFP compared to those who did not have [AOR² 2.85, 95% C.I (2.06-3.96)]. The reduction of the odds ratio from 3.34 in Model 1 to 2.85 in Model 2 was due to the presence of behavioural variables (breastfeeding, sexual abstinence, spousal discussion advocacy for family planning and information seeking on FP) which had strong negative influence on intention than background characteristics in Model 1. This also indicates that intentions alone may not predict PPFP use and that other factors are necessary in women's FP decisions. Women who had had spousal discussions, sort information or advocated about PPFP respectively, had significant relationship with PPFP use. Those who did not have any spousal discussion [AOR² 0.46, 95%C.I (0.32-0.66)] or advocate PPFP to other women [AOR² 0.55, 95%C.I (0.38 - 0.81] had lower odds of using PPFP compared to those who had. Interestingly, women who had not sort information about PPFP showed higher odds of using PPFP [AOR² 1.44, 95% C.I (1.01-2.07)]. Resumption of sex, breastfeeding, postpartum amenorrhoea, delivery outcome and place of delivery did not have any significant relationship with PPFP use (Table 27).

Model 3 deals with significant factors in models 1 and 2. The results indicate that women who had prenatal PPFP intentions still had significantly higher odds of using PPFP [AOR³ 2.98, 95% C.I (1.86-4.76)]. Women with tertiary education were 5.17 times more likely to use PPFP compared to those without any formal education [AOR³ 5.17, 95% C.I (1.24-21.50)]. Also, farmers were less likely [AOR³ 0.05, 95% C.I (0.01-0.41)] whilst petty traders were more likely [AOR³ 2.28, 95% C.I (1.01-5.18)] to use PPFP compared to fishmongers.

Women who did not advocate PPFP to other women were less likely to use PPFP [AOR³ 0.38, 95% C.I (0.22-0.65)]. Parity and preferred number of children also showed significant relationships with PPFP use (Table 27).

Reasons for use of PPFP	Frequency	Percent	
Partner approval	133	33.2	-
Self-approval	265	66.1	
Other	3	0.7	
Total	401	100.0	
Reasons for not using PPFP			
Got pregnant too soon	70	11.9	
Partner resistance	56	9.5	
Wanting to get pregnant as soon as possil	ole 88	14.9	
Child died soon after birth	19	3.2	
Lack of interest in FP	171	29.1	
Indecision	11 7	19.9	
Other	68	11.5	
Total	589	100.0	

Table 28: Reasons for Use and Non-Use of PPFP

Source: Field data (2014)

In the prenatal period, the strongest determinants of women's intention to use PPFP were self-approval, partner approval and past use of injectables. The main reasons for use of PPFP were not very different from these. Two out of every three women gave self-approval as the main reason for using PPFP. Lack of interest and indecision were the two main reasons given for not using PPFP (Table 28).

Summary

This study revealed that if women had prenatal intention to use family planning within the extended postpartum period, they were more likely to translate intention into action. In addition, if they preferred certain methods, during the prenatal period, they were more likely to use those methods in the postpartum period. These findings are consistent with prevailing theories of health behaviour which consider an individual's intention to perform a behaviour as one of the most important precursors to actually carrying out the behaviour (Azjen & Fishbein, 1980; Azjen, 1985 & 1991; Piotrow et al., 1997; Fishbein, 2000) and also consistent with findings of Callahan and Becker (2012) in their study on contraceptive intentions and use in Bangladesh. It was also revealed that inconsistencies existed between stated intentions and subsequent use of contraceptives. Some women said they will not, but eventually did, whilst some who said they will, never did. The lesson is that intention alone may not be enough to predict behaviours. There is the need to identify women with intention and follow and encourage them to do so during the postnatal period. Behavioural change communication needs to be targeted at women whose stated intentions are inconsistent with subsequent behaviour.

In this study, reasons given by women for not using PPFP included not interested in family planning, too early after delivery to decide on family planning, wanted to get pregnant as soon as possible, got pregnant too soon, partner disapproved of family planning and the loss of their child (Table 28) and reflect the dynamism of contraceptive use intention. For those who used PPFP, some influencing factors that were found in this study were women's self -approval and motivation, partner approval, higher educational status of women (tertiary level), high parity (5+ children), motivation to advocate for family planning and preference for large families. Self-approval and motivation by women is emerging as a window of opportunity that postpartum family planning programmes need to tap into. It indicated that there are some women who do not need consent from their partners to use family planning in the country. This notwithstanding, spousal engagement in family planning will help reinforce the self-motivation to improve contraceptive uptake.

CHAPTER SEVEN

SEX COMPOSITION OF CHILDREN EVER BORN AND INTENDEDNESS FOR POSTPARTUM FAMILY PLANNING AND POSTPARTUM CONTRACEPTIVE USE AND FERTILITY BEHAVIOURS

Introduction

This chapter presents and discusses two topical postpartum issues: (1) The implications of sex composition and the intention to use and use of postpartum family planning within the context of sex preference; and

(2) Postpartum behaviours of breastfeeding and sexual abstinence in relation to postpartum contraceptive behaviour and how critical this information is in identifying women who are susceptible to unintended pregnancy

Sex composition of children has been observed to influence use of family planning within patriarchal societies (Calhoun et al., 2013). In the southern sector of the country where the matrilineal Akans account for nearly half of the population (Assimeng, 1981; Oppong, 1981; Nukunya, 2003; Ghana Statistical Service, 2013), sex composition could similarly influence use of family planning. Fuse (2010) has observed that in Ghana sex preference is predominantly for balance (47%), followed by daughter preference (21%) and then son preference (19%). This study carries the debate a step further, by examining sex composition and its influence on classification of a pregnancy as intended or not and the

•

intention to use and use of postpartum family planning among predominantly matrilineal women in the Mfantseman Municipality.

Among the matrilineal Akans, there are statements to the effect, you are doomed if you do not have a sister, implying that will be the end of the lineage. Although matrilineal, it has elements of patriarchy. For instance, in most cases the heads of matriclans are males. Under the matrilineal system, relations consist of uterine siblings (Awusabo-Asare, 1990) and that although, a woman, her husband and their children are distinguished as a linked group (nuclear and conjugal), their functioning as a social group is less clearly defined. As a rule, the conjugal family is not the most effective social unit (Bleek & others, 1987) because a number of activities are organized around the lineage as well as at the conjugal level (Bleek et al., 1987; Caldwell & Caldwell, 1987). For instance, issues such as marriage and inheritance are organized with the view of the woman contributing to the continuity of the matriclan (Oppong, 1981; Lockwood, 1995); hence, a woman who gives birth to ten children, was expected to give a ram to the husband, for assisting her to add ten children to the matriclan. Among the patrilineal Ga-Adangbe, the ram is from the man to the woman for increasing the patriclan by ten children.

Women are defined as insusceptible to pregnancy following a birth if they are not at risk of conception because they are amenorrhoeic or abstaining from sex. The period of insusceptibility, which is influenced by sexual abstinence and breastfeeding, lengthens the time until the next birth. Use of contraception within this period offers extra protection against the risk of pregnancy (Gebreselassie et al., 2008). The length and intensity of breastfeeding and the length of amenorrhea and sexual abstinence vary among women and within societies. In this respect postpartum behaviours need to be understood in relation to the contraception behaviours of women especially in a setting such as the Mfantseman Municipal area.

In Ghana, the duration of breastfeeding can vary from 6 weeks to over one year and abstinence for up to two years. It is observed that in developed countries some women do not breastfeed at all, and for those who breastfeed, duration of breastfeeding is short. In contrast, among traditional societies in developing countries, duration of breastfeeding is longer; sometimes lasting till the next pregnancy occurs (Bongaart, 1978). Haggerty and Rutstein (1999), observed that in traditional societies, median duration of postpartum amenorrhea for mothers who did not breastfeed their children was 3 months whilst that of those who exclusively breastfeed their children was 14 months. In Egypt, Darwish and Sahn (1991) found that 31 percent of breastfeeding mothers were still amenorrheic at 6 months Postpartum.

Sex Composition of Children

The association between sex composition and reproductive health behaviours and outcomes such as FP use and pregnancy intendedness has been the commonly used approach to studying son preference (Calhoun et al., 2013). Sex composition was defined by each woman's number of living children and the sex of the children. Based on these considerations, sex composition was conceptualized into the following: daughters only, sons only, more daughters than sons, more sons than daughters and Equal number of daughters and sons. The understanding of fertility motivations are influenced by the value of children in demographic research. According to Williamson (1976), this value of children scheme is important when considering issues about sex preferences. Williamson outlines conditions that influence preference for sons, equal numbers of sons and daughters, a predominance of daughters and a lack of sex preference. A strong son preference is characterised by economic, social and psychological conditions (Gray & Evans, 2004). Parents will have a strong son preference if, economically, sons are more productive than daughters. This includes providing for parents in old age, bringing a dowry to the family once married, or if boys have more opportunities for advancement. The social conditions for son preference include: a system of patrilineal families and patrilocal residences; where continuity of the family line or name is important; where there is inter-group conflict; where social or religious customs require sons; and, in rural communities. The psychological reasons given for boy preference are to provide companionship for fathers, to provide status, security, and influence (Gray & Evans, 2004).

The conditions influencing girl preference are also economic, social and psychological(Andersson & Woldemicael, 2001). In economic terms, daughters are preferred if women are considered more productive than men, if girls bring a bride price, or if daughters support parents in old age. Socially, a matrilineal family system supports daughter preference, whereas psychologically, daughters are preferred if they are more rewarding companions or if there is social

competition between fathers and sons (Gray & Evans, 2004). Preference for sex balance is also associated with the assumption that boys and girls will have different traits, strengths, leisure activities, and interests (Williamson, 1976).

As traditional gender-role attitudes shift to expectations of shared roles there is a weakened effect of sex of existing children because boys and girls are substitutable. Evidence exists that as societies modernize, sex preference declines (Bongaarts, 2001; Pollard & Morgan, 2002), thus, making mixed sex composition preferable. A preference for both a son and a daughter suggests that boys and girls are viewed as providing different benefits to parents (Gray & Evans, 2004).

Results of this study showed that 35 percent of the women were pregnant for the first time. Thirty five percent (35%) of the pregnant women did not have any child. Of those who had at least one child (consisting of 861 males and 792 females and sex ratio of 109), the average number of children per woman was 2.0 \pm 2. A third (31.2%) of these respondents had only sons, whilst a quarter (24.5%) had only daughters. Just over a tenth had children of both sexes but more daughters than sons (11.6%) and more sons than daughters (11.3%) respectively. A fifth (21.4%) of them had equal number of sons and daughters (Table 29).

Sex Composition, Pregnancy Intendedness and Intention to Use PPFP

Bivariate descriptive analysis of sex composition and unintended pregnancy, PPFP intention and PPFP use were carried out as presented in Table 29. Women with only sons had the lowest levels of unintended pregnancy (58%) and PPFP intention and use (33.3%). Unintended pregnancy was common among women with more sons than daughters (82.1%) whilst intention to adopt PPFP was common among those with more daughters than sons (83.3%). For PPFP use, women with equal numbers of males and females had the highest prevalence (56.5%) (Table 29).

.

Demographic Characteristics	Sample Size (N)	Percent of total sample	Overall Pregnancy intendedness (Unintended)	PPFP Intention (Yes)	PPFP use (Yes)
		(%)	(%)** [']	(%)*	(%)!
Sex Composition			·		
Only daughters	254	24.5	64.2	77.6	45.1
Only sons	323	31.2	57.9	74.6	33.3
More daughters	120	11.6	76.7	83.3	33.9
than sons					
More sons than	117	11.3	82.1	76.9	48.4
daughters					
Equal number of	222	21.4	72.5	76.6	56.5
sons and daughters					
TOTAL	1,036#	100	67.47	77.0	44.9

PPFP Intentions	Table	29: Family PPFP I	Sex Comp	osition, P	regnancy	Intendedness	and
-----------------	-------	----------------------	----------	------------	----------	--------------	-----

.**p<0.001 for pregnancy Intendedness and *p>0.05 for PPFP intention. #Differences observed between these and original total of 1,914 are due to missing values which arose due to poor and incomplete responses !815 respondents out of 1004 in Phase 2 provided responses for computation of sex composition. Percentages in columns 4, 5 and 6 are row percentages

Source: Field data (2014)

To examine the influence of sex composition on reproductive health outcomes and behaviours (unintended pregnancy and PPFP intention and use) three logistic regression models were fitted: Model A (sex composition and pregnancy intendedness); Model B (sex composition and PPFP intention and Model C (sex composition and PPFP use) (Table 30)

There was no significant relationship between sex composition of children and intention to adopt PPFP, but with unintended pregnancy and PPFP use. Women with only sons had lower odds of classifying their pregnancy as unintended (OR_a 0.52, 95%CI 0.36-0.75) and higher odds of using PPFP (OR_b 2.60, 95%CI 1.07-6.33) compared to those with equal number of sons and daughters. Also, women with more daughters than sons were 2.5 times more likely to use PPFP than those with equal number of sons and daughters (OR_c 2.53, 95%CI1.75 – 3.65).

Concept of Breastfeeding and Sexual Abstinence as Elements for Spacing

In the second Phase of the survey, the women were asked whether they had breastfed their babies after birth and the duration; and also whether they had resumed sex and for how long after birth they had abstained from sex following the last birth. The responses are summarized in Tables 31 and 32. Seventy-eight percent of the follow up women reported that they had breastfed their babies within the first year after their last birth, whilst 88 percent had reported resuming sex within the first year since their last birth.

.

	Model A: Sex composition and unintended pregnancy	Model B: Sex composition and PPFP intention	Model C: Sex composition and PPFP use
	OR _a (95%CI)	OR _b (95%CI)	OR ₆ (95%CI)
Sex Composition			
	1.00	1.00	1.00
Equal number of sons and daugments (KEr) Only daughters	0.68(0.46 -1.00)	1.05(0.69 -1.62)	1.58(0.86-2.91)
Only sons	0.52(0.36 - 0.75)***	0.90(0.60 -1.34)	2.60(1.07 - 6.33)*
More daughters than sons	1.24 (0.74 – 2.09)	1.53(0.86 -2.71)	2.53(1.75-3.65)***
More sons than daughters	1.73 (0.99 -3.02)	1.02(0.60 -1.73)	1.39(0.96 – 2.00)

Table 30: Sex Composition and Women's Reproductive Health Outcomes

Source: Field data (2014)
The lowest proportion of breastfeeding was recorded among women forty years and above (56.7%) and those married under the ordinance system (59.2%) whilst the highest was among women aged 20-24 years (85.4%) and those cohabiting (85%).

Although breastfeeding amongst the women was generally high by education, parity, religion, facility of delivery and intention to adopt PPFP categories, they were highest among women with tertiary education (89%), fishmongers, (94.2%); those who delivered in private facilities and maternity homes (100%) and those who had no intentions to adopt PPFP (88.1%). Breastfeeding was lowest among women with primary education (70.6%), petty traders (70.1%), those who delivered in public health facilities (76.2%) and those who intended to adopt PPFP (Table 31).

Over 80 percent of the women by age groups had resumed sex at the time of the survey. There were no significant differences in the observations made ($\chi^2 =$ 4.7, p<0.50). As expected, the lowest proportion of women who had resumed sexual relationships were among those separated or divorced (20%), whilst the highest proportions were observed among those who were cohabiting (99%) ($\chi^2 =$ 90.4, p<0.001). The proportion of women who had resumed sexual relationships was lowest among farmers (52%) and women who delivered in private facilities (2.4%) (Table 31).

-	Breastfed within a	Resumption of sex	
	year after last birth	within a year since	
		last birth	
-	Yes	Yes	
	% (Number)	%(Number)	Total Number
Age (Years)	$\chi^2 = 19.2, p < 0.001$	$\chi^2 = 4.7, p < 0.50$	
15-19	84.9(28)	90.9 (30)	33
20-24	85.4(164)	86.5 (166)	192
25-29	83.6(235)	90.4 (254)	281
30-34	72.8(195)	86.2 (231)	268
35-39	78.8(104)	88.6 (117)	132
40+	56.7(55)	83.5 (81)	97
Marital Status	$\chi^2 = 55.2, p < 0.001$	$\chi^2 = 90.4, p < 0.001$	
Married (Ordinance)	59.2 (126)	94.8 (202)	213
Married (Traditional)	82.5 (358)	91.2 (396)	434
Engaged	83.3(200)	73.3 (176)	240
Cohabitation	84.9 (52)	98.6 (72)	73
Divorced/Separated	80.0 (4)	20.0 (1)	5
Single	81.6 (31)	84.2 (32)	38

Table 31: Breastfeeding and Sexual Abstinence by BackgroundCharacteristics

- L.

.

Table 31 continued			
Education	$\chi^2 = 21.6, p < 0.001$	$\chi^2 = 18.4, p=0.001$	
None	88.0 (73)	73.5 (61)	83
Primary	70.6 (151)	86.5 (185)	214
Middle/JSS	76.2 (414)	89.8 (488)	543
SSS/SHS/Voc	87.4 (104)	89.1 (106)	119
Tertiary	88.6 (39)	88.6 (39)	44
Parity	$\chi^2 = 199.0; p < 0.001$	$\chi^2 = 25.0, p < 0.001$	
0	100.0 (4)	100.0 (4)	4
1-2	87.7 (435)	89.3 (442)	495
3-4	84.0 (278)	81.0 (268)	331
5+	87.6 (65)	95.4 (165)	173
Religion	$\chi^2 = 2.4, p < 0.50$	$\chi^2 = 20.9, p < 0.001$	
Christian	77.5 (702)	87.9 (796)	906
Muslim	84.2 (64)	92.1 (70)	76
Traditionalist	75.0 (3)	100 (4)	4
Other	70.6 (12)	52.9 (9)	17
Occupation	$\chi^2 = 50.9, p < 0.001$	$\chi^2 = 49.9, p < 0.001$	
Fishmonger	94.2 (65)	78.3 (54)	69
Farmer	87.9 (29)	51.5 (17)	33
Petty trader	70.1 (361)	90.3 (465)	515
Civil/Public Servant	75.7 (84)	89.2 (99)	111
Student	75.0 (24)	84.4 (27)	32
Other	89.7 (218)	89.3 (217)	243

Table 31 continued

Facility of delivery	$\chi^2 = 19.2, p < 0.001$	$\chi = 318.5, p < 0.001$	
Home	93.9 (31)	63.4 (21)	33
Public hospital/	76.2 (705)	92.3 (854)	925
Private hospital	100 (42)	2.4 (1)	42
Maternity home	100 (3)	100 (3)	3
Intends to adopt FP	$\chi^2 = 20.1, p < 0.001$	$\chi^2 = 2.42, p {<} 0.12$	
Yes	74.5 (560)	86.7 (652)	752
No	88.1 (221)	90.4 (227)	251
All Total	77.9 (781)	87.6 (879)	1,003

Percentages are row percentages

Source: Field data (2014)

Duration of Breastfeeding, Postpartum Amenorrhoea and Sexual Abstinence

Table 32 depicts the duration of breastfeeding, sexual abstinence and amenorrhoea in the postpartum period by background characteristics.

Duration of Breastfeeding

The overall median duration of breastfeeding among the women who breastfed was 6.6 months, with no substantial variation by mother's age. Youngest women (15-19 years) had the longest median duration of breastfeeding (7.2 months), whilst the shortest of 6.2 months was found among those 25-29 years.

	Breastfeeding	Sexual abstinence	Amenorrhoea
Overall Median duration	6.6	4.4	7.8
Age			
15-19	7.2	6.8	7.1
20-24	6.5	4.7	7.6
25-29	6.2	4.1	7.8
30-34	6.7	4.5	7.7
35-39	6.6	4.1	8.5
40+	7.0	4.5	8.4
Marital status			
Married (Ordinance)	5.5	4.4	6.8
Married (Traditional)	6.8	4.5	7.6
Engaged, yet to marry	7.5	4.3	9.0
Co-habitation	6.1	3.0	8.5
Divorced/Separated	2.3	12.0	2.8
Single	6.0	7.3	7.4
Education			
None	7.5	4.2	9.0
Primary	7.7	4.8	8.0
Middle/JSS	6.1	4.3	7.6
SSS/SHS/Vocation/Tertiary	6.5	4.4	8.0

Table 32: Median Duration (Months) of Breastfeeding, Sexual Abstinence and Postpartum Amenorrhoea

.

Table 32 continued

_

Religion		"	
Christian	6.5	4.5	7.9
Muslim	6.2	4.3	6.7
Traditionalist	4.7	2.3	6.5
Other	9.5	3.5	9.9
Parity			
1-2	6.3	4.5	7.6
3-4	7.4	4.2	8.7
5+	5.7	5.0	6.7
Occupation			
Fishmonger	7.6	4.6	9.2
Farmer	10.2	5.1	11.1
Petty trader	6.4	4.2	7.4
Civil/Public servant	6.5	4.7	7.3
Student	6.5	6.0	7.9
Other	6.2	4.3	8.1
Facility of delivery			
Home	7.0	4.6	8.2
Public hospital/	6.2	4.4	7.6
Private hospital	13.8	4.0	12.9
Maternity home	6.0	3.0	12.0

.

Table 32 continued

Family planning intent	ion			
Yes	6.8	4.5	7.8	
No	5.9	4.3	7.9	

Source: Field data (2014)

Differences existed by marital arrangements. The shortest median duration of breastfeeding (2.3months) was observed among women who were divorced or separated, whilst the longest was among those who were engaged but yet to marry (7.5months). The difference in median duration of breastfeeding among women who were married by ordinance and those married traditionally was 1.3 months (5.5 versus 6.8 months respectively)

There is a general trend towards declining median duration of breastfeeding as educational level increased. The women who had no formal education and those with primary education breastfed for at least a month longer than those with higher levels of education (Table 32). Traditionalists and grandmultiparous women (5+) breastfed for shorter durations (4.7 and 5.6 months) among the major religions and parity groups respectively,

Women who delivered at private health facilities breastfed for at least 5 months longer than those who delivered at public facilities and maternity homes. Furthermore, women who had prenatal intentions to use family planning breastfed nearly a month longer than those who did not have any prenatal intentions

Duration of Postpartum Amenorrhoea

The overall median duration of amenorrhoea among the women was 7.8 months. Although no major differences were observed among age groups, there was an increasing trend towards increasing median duration of amenorrhoea as age increased. Among the different marital arrangements, women who were divorced or separated had the shortest duration of amenorrhoea (2.8 months). Those who were engaged and yet to marry had the longest duration of amenorrhoea (9 months). As educational level increased, median duration of amenorrhoea tended to decrease. Those with no education had duration of amenorrhoea at least a month longer than those with some education.

Traditionalists and grand-multiparous women had the shortest durations (6.7 months) of amenorrhoea amongst the major religious and parity groups. Women who delivered at private facilities had durations of amenorrhoea at least 5 months longer than those who delivered in public health facilities. No major differences in durations of amenorrhoea were observed among women who had or had no prenatal intention to use family planning.

Duration of Postpartum Sexual Abstinence

The overall median duration of sexual abstinence was 4.4 months amongst the women. As age increased, the median duration of sexual abstinence generally decreased. Women aged less than 20 years had at least 2-month duration of sexual abstinence longer than those aged 20 years and above. The women who were

SAN CHARLISPARY

divorced / separated or single had relatively longer durations of sexual abstinence (12 - and 7.3 - months) respectively than those with other marital arrangements.

Traditionalists had the shortest median duration of sexual abstinence. The duration was about 2 months shorter than those of Christians and muslims. Grandmultiparous women had the longest duration of sexual abstinence amongst the different parity groups. There were no major differences in durations of sexual abstinence by educational status, facility of delivery and prenatal family planning intentions.

Postpartum Contraceptive Use Before and After Susceptibility

Women are considered insusceptible if they are not exposed to the risk of pregnancy because they are either amenorrheic or abstaining from sexual intercourse after a birth. The duration of postpartum insusceptibility is the longer of the two (Gebreselassie et al., 2008). Table 33 shows the median time of insusceptibility, time of postpartum contraceptive use and number of women who used contraceptives before and after susceptibility.

Postpartum women who used contraceptives during the period of insusceptibility to pregnancy are considered to be those who reported using a method while they were either amenorrheic or abstaining at the time of the survey. On the other hand, those who used contraception after they became susceptible to pregnancy are those who reported using a method while they were neither amenorrheic nor abstaining at the time of the interview (Gebreselassie et al., 2008). As shown in Table 33, women who used contraceptive methods in the postpartum period from 0-9 months did so before they became susceptible to pregnancy. These women constituted the majority (96.5%). Very few (3.5%) of the women used contraceptives after they had become susceptible to pregnancy. These women used contraceptives at least a month into their period of susceptibility.

Summary

The reported sex composition of children ever born among the women was skewed towards males. Sex ratio of children ever born for the group was 109. In the study, women with only sons were significantly less likely to be carrying unintended pregnancies at the time of the survey. The fact that these same group of women were found to be the more likely to use PPFP may explain the reported low unintended pregnancy among them. This group of women may have satisfied an underlying preference for sons and may not want any more children. This is in contrast to findings made by Calhoun and colleagues in which families with only sons were significantly less likely to want no more children and used FP less (Calhoun et al., 2013).

Despite underlying son preference observed in the study, such women may not have any overall preference for male children (Bhatia, 1984). Therefore, any additional pregnancy may be intended, to satisfy the sex composition of their children. This is because both girls and boys are valued in a system which is matrilineal but has patriarchal elements ((Rajaretnam & Deshpande, 1994; Jain, 1999; Edmeades et al., 2012; Calhoun et al., 2013; Edmeades et al., 2012).

Time of PPFP use	Median duration of	Median duration of amenorrhoea	Median duration of sexual abstinence	Median duration of	Number of	Percentage of women
(smuoimi)	(Months)	(sunnoni)	(SUNIONII)	insusceptionity (months)	MOLITELL	
0-3	6.5	8.7	3.7	8.7	276	65.4 ^a
4-6	6.3	7.7	6.1	7.7	96	22.8 ^a
-9°	6.6	7.5	5.6	7.5	35	8.3 ^a
10-12	6.0	6.8	9.6	9.6	12	2.8 ^b
>12	6.7	12.0	5.7	12.0	т	0.7^{b}

Table 33: PPFP Use Before and After Susceptibility to Pregnancy

8 and 9 months was 8- and 10- months respectively

Source: Field data (2014)

In this study, women with more daughters than sons reported high levels of unintended pregnancy, expressed the highest intention to use PPFP (Table 29) and were more likely to use PPFP. It is possible that these women may have satisfied the sex composition of their children and therefore any additional pregnancy may not be wanted. Calhoun and colleagues also found that women with more daughters than sons were less likely to have no more children and use FP less (Calhoun et al., 2013). Women with more sons than daughters also reported high levels of unintended pregnancy because they probably had satisfied the sex composition of their children. However, this group of women showed no significant association with PPFP use. Family planning programmes need to be mindful of these women for education and counseling.

Sex composition of children is not significantly associated with intention to use contraceptives postpartum and therefore does not influence it. This study has revealed that in the Mfantseman Municipality, despite the persistence of more sons than daughters ever born, the preference is for sex balance in a predominantly matrilineal inheritance system. Furthermore, sex composition of children was significantly associated with pregnancy intendedness and PPFP use but not with postpartum family planning intention. The next chapter discusses postpartum behaviours (breastfeeding, sexual abstinence and amenorrhoea) and their relationship with postpartum contraceptive use.

Background, Socio-cultural and environmental factors influence significantly a woman's decision to breastfeed(Brand, Kothari, & Stark, 2011). Contrary to findings of other studies which showed that breastfeeding was common among older women 30 years and over(Renfrew et al., 2012), this study

found that the proportion of women who breastfed was highest among younger women 20 -24 years and lowest among those forty years and over. The youthful population in the Mfantseman Municipal may partly explain the picture; however, further studies need to be carried out to identify the reasons for this observation.

Marital status also affects breastfeeding rates, initiation and duration (Brand et al., 2011). In this study, breastfeeding rates were higher among cohabiting than women married under the ordinance system. This contrasts other studies where married women had higher rates of breastfeeding compared to unmarried women (Chin, Myers, & Magnus, 2008; Thulier & Mercer, 2009). Factors within ordinance marriages that influence breastfeeding in this setting are not clear and need to be investigated especially because of available evidence that if a woman views breastfeeding positively, and has support from her partner, she will be more likely to breastfeed (Persad & Mensinger, 2008).

Health-care workers, especially nurses and doctors play important and integral roles during the immediate postpartum period in helping mothers through individualized, interactional techniques to initiate and sustain breastfeeding (Taveras et al., 2003; (Swanson & Power, 2005; Persad & Mensinger, 2008; McInnes & Chambers, 2008). The finding in this study that women who delivered in private hospitals and maternity homes have higher rates of breastfeeding than those who delivered in public health facilities may be explained by the fact that health care workers in private hospitals and maternity homes in the Mfantseman Municipal have adequate time to counsel and educate women at antenatal and immediate postpartum periods because of relatively less patient loads compared to the public health facilities which handle relatively high patient loads. This however needs to be confirmed by further inquiry. This finding is in contrast with those of other studies in Ghana and Nigeria which have found the performance of breastfeeding among mothers delivering in public facilities to be better than those who delivered in private health facilities (Aidam, Perez-Escamilla, Lartey, & Aidam, 2005;Ukegbu, Ukegbu, Onyeonoro, & Ubajaka, 2011). Higher education has been shown to influence positively health outcomes in several studies. The finding in this study that higher education favours high rates of breastfeeding is consistent with other studies (Chin et al., 2008;Chalmers et al., 2009)

When women fully or nearly fully breastfeed and remain amenorrheic then it is effective as a postpartum contraceptive method and offers over 98% protection against pregnancy in the first six months postpartum (Trusell, 2011). When the period of sexual abstinence exceeds the period of amenorrhoea, it offers additional protection against pregnancy. The median duration of amenorrhoea (7.8months) in this study exceeds that of breastfeeding (6.6months) by 1.2 months; indicating that on the average, women in the Mfantseman Municipal exclusively breastfed, remained amenorrheic and hence insusceptible to pregnancy for at least the six months period recommended by WHO for exclusive breastfeeding. This far exceeds the median duration of exclusive breastfeeding of 3-months in Ghana (GSS & Macro, 2009). This notwithstanding, women who are divorced/ separated need to be targeted for education on breastfeeding because their duration of breastfeeding and amenorrhoea were the shortest (2.3 months) thus putting them at risk of unintended pregnancies. Exposure to the risk of pregnancy in the postpartum period can be influenced by sexual abstinence. Postpartum sexual abstinence tends to have additional contraceptive benefits if the duration of abstinence exceeds that of postpartum amenorrhea (Gebreselassie et al., 2012). The median duration of postpartum sexual abstinence of 4.4 months although longer than in other cultural settings (3.4 months in Kenya, 2.0 months in Indonesia, 2.1 months in the Dominican Republic, and 2.9 months in Peru) (Gebreselassie et al., 2012), is shorter than observed in Ghana (7.5 months) (GSS &Macro, 2009) and shorter than the periods of insusceptibility. This relatively short duration of sexual abstinence in the Mfantseman Municipal, is inadequate to prevent unintended pregnancies among women who: never use contraceptives; use contraceptives late; have shorter periods of amenorrhoea; and exclusively breastfeed or who never breastfeed at all.

Among women who used contraceptives postpartum, over 95% of them did so when they were insusceptible to pregnancy. This indication of early uptake of contraceptives at a time when women are not at risk of pregnancy makes it even more imperative for integration of family planning services and childhood immunization services.

family planning in the postpartum period into actual use. Data from the study is based on a prospective panel study carried out in two Phases and involved interviewing women during the antenatal period and up to two years after delivery. As part of the process, several theories of behaviour change were reviewed, among them, the steps-to-behaviour change model, Bongaarts framework for analyzing determinants of fertility, theory of planned behaviour and the postpartum use of contraceptives & postpartum behaviours framework. This chapter highlights the main findings, conclusions, policy implications, policy recommendations and contributions to knowledge.

Key Findings of the Study

Phases 1 and 2 consisted of women of similar background characteristics. In both Phases, the women were mainly below 30 years, Christian, married under the traditional system, were petty traders, had mostly attained middle school/JSS education and had mostly up to two children. Seventy percent of the women reported that the pregnancies they were carrying were unintended, with 39 percent being mistimed and 31 percent unwanted. Unintended and unwanted pregnancies were higher among those less than 20 years and 40 years or more whilst mistimed pregnancy was higher among those aged 20-24 years. Unintended and unwanted pregnancies among women with no formal education were 2.5 and 10 times those among women with tertiary education respectively. Unintended pregnancy was high among all religious groups, but relatively higher among the traditionalists and relatively lower among Muslims and Catholics. Single or never married women had the highest unintended and unwanted pregnancies whilst those married under the ordinance system had the lowest. Unintended pregnancies were higher among students, those in the informal sector, rural dwellers and those with higher parity (5+) than civil/public servants, semi-urban dwellers and those with lower parity. Predictors of unintended pregnancy included higher parity, other forms of marital arrangements (apart from ordinance marriage), not living with partners, and nonawareness of traditional methods of contraception.

The most known methods of contraception were the male condoms, injectables, female condoms and the pill; whilst the least known were the foam, diaphragm and intrauterine device (IUD). Women aged 15-19 years, who had no formal education, had no children, were single, divorced, separated or students, were less likely to know about family planning methods compared to women aged 25-29 years, who had tertiary education, had at least a child, were married under the ordinance system and were civil/public servants. The withdrawal method was the most used whilst IUD was the least used in the past.

Seventy percent of the pregnant women expressed intention to use PPFP. Women who were aged 15-19 years, were single, had never had any children, had tertiary education and were civil/public servants were less likely to express intention to use postpartum family planning compared to women who were aged forty years and above, had five or more children, had primary education and were divorced/separated who were more likely to express intention to adopt postpartum family planning. For modern family planning methods that women intended to use postpartum, injectables were the most preferred, followed by the pill and male

condom. Determinants of women's intention to use PPFP included the perception of partner's acceptability of FP (AOR 4.42, 95%CI 3.25-6.01), personal conviction of acceptability (AOR 4.69, 95%CI 3.29-6.69), prior experience with using injectables (AOR 2.74, 95%CI 1.78-4.20) and pills (AOR 1.74, 95%CI 1.21-2.50), awareness of the diaphragm (AOR 0.59, 95%CI 0.44-0.80) as a contraceptive method or prior use of IUDs (AOR 0.22, 95%CI 0.06-0.76). Women who were cohabiting at the time of the survey exhibited higher odds of intending to use PPFP compared to those who were married under the ordinance system (AOR 1.96, 95% CI 1.13-3.39).

Of the 70 percent of women in Phase 1 who expressed intention to use family planning, over half (50.4%) had used PPFP at the time of the second Phase. This consisted of 42 percent of women who expressed prenatal intentions to use PPFP and 8 percent of those who expressed no prenatal intention to use PPFP. Thirty-three percent of women expressed prenatal intention to use PPFP but did not whilst 13 percent expressed intention not to and did not. Despite the observed inconsistencies between prenatal intentions and actual PPFP use, the odds of a woman using PPFP given that she had a prenatal intention was about three times the odds of not using PPFP (p<0.001). The women who had used PPFP at the time of the survey were more likely to be 35- 39 years (66%), attained mostly middle/JSS education (56%), had five or more children (88%), were Muslims (51%) and Christians (51%), were mostly cohabiting (63%), were petty traders(60%) and were likely to be from Anomabo (73%). Among the women who had used PPFP, the mean time of PPFP method use was $3.5\pm 2.7months$ after

delivery, with male condoms, injectables and pills being the most methods used. Suitability of method to client (83%) was the main driving force behind choices of PPFP methods by the women.

Prenatal intention [AOR 2.98, 95% C.I (1.86-4.76)], tertiary education [AOR 5.17, 95% C.I (1.24-21.50)], being a farmer [AOR 0.05, 95% C.I (0.01-0.41)] or petty trader [AOR 2.28, 95% C.I (1.01-5.18)] and Non- advocate of PPFP [AOR 0.38, 95% C.I (0.22-0.65)] predicted PPFP use. Parity and preferred number of children also showed significant relationships with PPFP use. Selfapproval and partner approval were the main reasons given for use of PPFP. For women who had not used PPFP, the main reasons for non-use were lack of interest in FP, indecision, wanting to get pregnant as soon as possible and getting pregnant too soon.

Sex composition of children ever born by the women was analyzed for its possible influence on wantedness of pregnancy intention to use and actual use of PPFP. Thirty five percent of the pregnant women did not have any child. Of those who had at least one child, a third of them had only sons, whilst a quarter had only daughters. A little over a tenth had children of both sexes but more daughters than sons (11.6%) and more sons than daughters (11.3%) respectively. A fifth of the women also had equal number of sons and daughters. There was significant relationship between sex composition and unintended pregnancy and PPFP use, but not with intention to adopt PPFP. Women with only sons were less likely to have unintended pregnancy (OR 0.52, 95%CI 0.36-0.75) and more likely to use PPFP (OR 2.60, 95%CI 1.07-6.33) than those with equal number of sons and

daughters. Also, women with more daughters than sons were 2.5 times more likely to use PPFP than those with equal number of sons and daughters (OR 2.53, 95%CI1.75 - 3.65).

Duration of breastfeeding among women who breastfed was 6.6 months, whilst, amenorrhoea was 1.2 months longer. Women who were divorced or separated had the shortest duration of breastfeeding (2.3months) and amenorrhoea (2.8 months). The duration of postpartum sexual abstinence among the women was 4.4 months. This was longer than in other cultural settings but shorter than observed in Ghana (7.5 months). Among women who used contraceptives postpartum, over 95% of them did so when they were insusceptible to pregnancy.

Conclusions

The level of reported unintended pregnancy was high (70%) and was influenced by higher parity, marriage (other than ordinance marriage), women not living with their partners, and non-awareness of traditional methods of contraception. Women who were aged forty years and above, had five or more children, had had primary education and were divorced or separated were more likely to express intention to adopt postpartum family planning. The positive predictors of women's intention to use postpartum family planning were the perception of partner's acceptability of FP (not background characteristics), personal conviction of acceptability, and prior experience with using injectables and pills. Negative predictors of intention to use postpartum family planning include past experience with use of intrauterine device (IUD) and awareness of the

diaphragm. Approval of PPFP (by the woman and Partner) remains one of the strongest drivers of women's use of postpartum family planning in the municipality.

There was a relationship between women's intention to use postpartum family planning and their subsequent use in the extended postpartum period. Women with prenatal intentions to use postpartum family planning were three times as likely to do so as those who did not express such intention. Determinants of PPFP use included intention to use PPFP, higher education (tertiary), high parity (5+), preference for five children or more and being non-advocate for FP.

Sex composition of children born had no influence on intention to use PPFP but influenced pregnancy intendedness and PPFP use. Average timing of PPFP use was 3.5 ± 2.7 months and among women who used family planning in the postpartum period, nearly all of them (95%) did so when they were insusceptible to pregnancy. This is an additional protection against unintended pregnancy apart from that offered by the relatively long periods of exclusive breastfeeding and lactational amenorrhoea observed among women in the municipality.

It can be concluded from the key findings that women's prenatal intentions to use family planning in the postpartum period, reasonably predict their actual use and that this is more likely to occur if family planning is acceptable and approved by the woman and supported by the partner. In addition, the likelihood of postpartum family planning use is influenced by socio-cultural considerations such as the sex composition of children ever born in a predominantly matrilineal inheritance system.

Evaluation of Conceptual Framework

The social structure and fertility framework developed by Davies and Blake (1956) and the steps to behavior change model were adopted for the study to examine the intention to use and use of contraceptives postpartum among pregnant women in the Mfantseman Municipality. The first framework is made up of the indirect and intermediate or proximate determinants of fertility. The second, the steps to behaviour change model, developed by Piotrow and colleagues in 1997 is made up of behaviour change constructs of knowledge, approval, intention, practice and advocacy.

In this study, the proximate determinants that were explored as a means to influence fertility were contraception and infecundity and voluntary abstinence. For infecundity, issues explored were breastfeeding and lactational amenorrhoea and intercourse factor explored was voluntary postpartum sexual abstinence.

Indirect factors influence fertility through the proximate determinants. The indirect factors explored in this study were mainly individual characteristics such as the demographic and socio-economic. These individual determinants provided the basis for exploring the background characteristics of the women and their sexual partners (where available) in relation to age, parity, area of residence, ethnicity, religion, highest completed level of education and occupation. It was found in the study that women's background characteristics influenced their knowledge, intention and use of family planning.

The results from the study can be situated in the steps to behaviour change framework (Table 34). The framework has knowledge, Approval, Intention to use

and Use of family planning (Practice) and Advocacy for family planning as its main constructs.

.

Construct	Description	Study G diago
Knowledge	Individuals recall family planning messages, understand what family planning means and can name family planning method(s) and/or source of supply.	The most known methods among the women were male condom (86.4%), injectables (80.5%), female condom (77.6%) and the pill (77.2%). The least known methods were the foam (20.7%), diaphragm (24.4%) and intrauterine device (IUD) (37.9%). Knowledge of implants, injectables and LAM predicted intention to use PPFP, whilst knowledge of the diaphragm was associated with lack of intention to use PPFP
Approval	Individuals approve of family planning and respond favourably to family planning messages. They further discuss family planning with personal networks (family, friends, community and any significant others) and have the perception that they approve of family planning.	Personal conviction about and approval of family planning, perceived partners acceptability and approval of family planning and spousal discussion about family planning influenced intention and actual use of FP
Intention	Individuals recognize that use of family planning can produce positive outcomes and can meet personal needs; therefore, they intend to consult a provider and also intend to practice family planning at some point.	70% of respondents expressed intention in Phase 1 to use PPFP. Intention predicted use of FP. If an individual expressed prenatal intention to use FP, she was 3 times more likely to use PPFP. Most of the women had prenatal intention to use the pill, injectable and male condoms

Table 34: Study Findings Related to Steps to Behavior Change Constructs

Table 34 continued

a intention to use PPFP, whilst use of IUD was associated a no intention to use FP. % of respondents had used P. In the postpartum period, 6 of women who had prenatal ntion had used PPFP, 33% 6 had prenatal intention had used, 8% who had no prenatal ntion had used, 17% who had intention had not used. The st used methods were pill, ctables and male condoms.
y 32% of the women had ken about FP to their friend or tive; individuals who had ken about FP to somebody e 2.6 times more likely to use FP
ct y ke tiv ke P

Source: Field data (2012 - 2014)

According to Piotrow et al (1997), individuals and groups progress from knowledge to sustained behaviour change and advocacy. These individuals may be at different stages at any point in time and constitute distinct audiences for programmatic targeting in FP. The results showed that postpartum contraception, which was the main focus of the study influences fertility through the Steps to Behaviour Change Model. Knowledge about contraception and exclusive breastfeeding are important to making informed decisions about contraception. The women in this study generally knew about contraceptives. They knew mostly the male condom, injectables, female condom and the pill and knew the least about the foam, diaphragm and IUD. With this knowledge, individuals, couples and significant others are able to approve and accept (or otherwise) to use or support the use of family planning. This study found women's personal conviction about family planning and their approval, supported by those of their partners to influence their intention to use PPFP. Individuals having recognized the positive outcomes of family planning at some point as was expressed by 70 percent of respondents in Phase 1 of this study. Practice of family planning may be influenced directly or indirectly by knowledge, approval or intention. Some individuals may have used family planning in the past and may have experienced positive outcomes.

These experiences may also influence current use of family planning. Knowledge of implants, injectables and LAM positively predicted intention to use PPFP whilst knowledge of the diaphragm was negatively associated with intention to use PPFP. Past use of injectables and pills were positively associated with intention to use PPFP. Most of the women in this study had prenatal intention to use the pill, injectables and male condoms and actually used same methods the most in the postpartum period.

To practice family planning, individuals go to a provider of information, supplies or services to seek information, choose a suitable method and begin use. Those who had used family planning in the past may also choose to continue based on past experience. Of all the women in Phase 2, over half (50.4%) had actually used family planning. Forty-three percent of them had prenatal intention to use and had used, 33 percent had prenatal intention to use and had not used, 8 percent had no prenatal intention and had used, whilst 17 percent had no intention to use and had still not used PPFP. This implies that 60 percent of those who had prenatal intentions to use or not to use PPFP carried out behaviours consistent with their intentions, whilst 40 percent carried out behaviours contrary to their intentions. When individuals have had positive experiences with use of family planning, they are motivated to advocate and recommend contraception to others. Although, few of the women (32%) had spoken about family planning to their friend or relative, individuals who had done so were three times as likely to use PPFP. The results test and practicalize the Steps to Behaviour Change model in its entirety from knowledge, through approval, intention, practice and advocacy.

Contribution to Knowledge

First, given that the results of this study can be situated in the Steps to Behaviour Change model makes it possible to examine practically the pathway through which contraceptive use influences fertility. This will benefit family planning education by helping to design appropriate messages that target individuals at different stages at any point in time;

Second, the study used a prospective design that enabled the statistical measure of the strength of intention as a predictor of actual family planning use.

This study adds to the scanty body of knowledge in family planning research that are able to predict actual or future FP use;

Third, although unintended pregnancies are high in Ghana and have been topical over the years, not many studies have been done to search for predictors. So far only one study (Omane-Adjepong et al., 2012) has detailed the determinants of unintended pregnancy in Ghana. This study is the second to detail the determinants of unintended pregnancies in Ghana. Some new predictors included living arrangements with partner, marriage by ordinance and awareness of traditional, non-pharmacological contraceptive methods;

Fourth, family sex composition in the context of son preference has been studied extensively in South and South-East Asia. Not many of such studies have been carried out in Africa and in Ghana, especially in recent times, to ascertain the influence of sex composition on reproductive health outcomes and behaviours such as pregnancy intendedness, intention to use and actual use of PPFP. Previous studies have also been carried out in more patriarchal settings. This study explored a new dimension to such studies by studying sex composition within a matrilineal inheritance system. Findings revealed that that sex composition had influence on pregnancy intendedness and PPFP use, but not on intention to use PPFP in a predominantly matrilineal inheritance system; and

Finally, determinants of intention to use and actual use of family planning in several studies have been generally socioeconomic and demographic. This study has examined the influence of self-approval and acceptability of family planning in adoption of family planning. This study revealed a strong influence of these factors in the Mfantseman Municipality and opens a window of opportunity to family planning managers to encourage and empower women to improve their decision making with regards to contraceptive choices.

Policy Implications

This study revealed that a woman's prenatal intention to adopt family planning is a strong predictor of use of family planning in the extended postpartum period. If a woman had a prenatal intention to adopt family planning in the postpartum period, she is nearly three times likely to do so despite the existence of other influences. This evidence has the opportunity to improve prenatal targeting and improve postpartum family planning uptake. For example, implementation of the family planning programme need to put in place strategies to follow up those who express intention and assist them to carry them through and also have strategies for those who may not express any intention to do so.

The fact that women mostly used postpartum family planning at the time that they were not susceptible to pregnancy offers additional protection against unintended pregnancy, in addition to the positive postpartum behaviours of prolonged durations of exclusive breastfeeding and lactational amenorrhoea. Strategies need to be strengthened to promote LAM and target women to use PPFP when they are least susceptible to pregnancy.

Recommendations

Based on the findings and conclusions drawn from this study, the following recommendations have been put forward:

- Considering the low attention given to PPFP in the current national family planning programme in general and in the MNCH policies, guidelines and service protocols in particular, it is recommended that the Family Health Division of the Ghana Health Service advocates strengthening integration of PPFP into the existing MCH/FP programmes.
- 2. Designing postpartum family planning programmes and interventions to reach women at one or more specific contact points within the health system is important to improving family planning uptake. These contact points include antenatal care, labour and delivery, pre-discharge, postnatal care, immunization and child health care clinics. During antenatal visits pregnant women should be targeted for discussion about their postpartum reproductive intentions relating to spacing or limiting and use of PPFP. This is based on the knowledge from this study that intentions reasonably predict subsequent behaviours. Other interventions that can be instituted include provision of PPFP and exclusive breastfeeding information and counselling at the facility and community levels. Such counselling by health workers should target couples and address their reproductive desires. Furthermore, information, education and communication materials on PPFP should be made available to women, partners and other significant family members.

- 3. During labour, delivery and pre-discharge, there is the need for health workers to ensure that women who choose a postpartum method receive high-quality PPFP services prior to discharge. Counseling on exclusive breastfeeding and lactational amenorrhoea are offered and women encouraged starting breastfeeding immediately before discharge. Furthermore, during immunization and child welfare period community health nurses need to mobilize mothers for immunization days, assist with and participate in PPFP group education sessions, and follow up mothers in the household for PPFP.
- 4. The Ghana Health Service's Family Planning Programmes may need to consider promotion of traditional, non-pharmacological methods alongside the modern methods, especially in rural settings, to improve overall contraceptive prevalence rates. For clients who do not want to adopt modern contraceptives despite all reassurances, the option of traditional methods should be offered them. This implies that health workers may need to be trained adequately to provide such services. Commitment from the Ghana Health Service Family Planning Programme would be required if this is to succeed.

Areas for Further Research

This study has revealed useful information about postpartum family planning; and since it is the only comprehensive study so far on PPFP in Ghana, it

would be useful to commission a national study on PPFP in order to capture context –specific issues that will help enrich a PPFP programme.

A more detailed study would be required to elucidate the reasons why women who delivered in private health facilities and maternity homes had high rates of breastfeeding than those who delivered at public health facilities. To obtain a more comprehensive picture about sex preference in the country, a study on family sex composition within a patrilineal system in Ghana needs to be undertaken since socio-cultural contexts differ across the country.

An investigation of the factors that make women in Ghana prefer injectables would most likely provide indications of the interventions that would be required to increase the uptake of PPFP in general. The possibility that injectables make it possible for women to use FP discretely needs to be considered.

BIBLIOGRAPHY

- Abbasi-Shavazi, M. J., & Hosseini-Chavoshi, M. (2004). Unintended pregnancies in the Islamic Republic of Iran: levels and correlates. *Asia-Pacific Population Journal*, 19(1), 27–38.
- Achana, F. S., Bawah, A. A., Jackson, E. F., Welaga, P., Awine, T., Asuo-Mante,
 E., & Phillips, J. F. (2015). Spatial and socio-demographic determinants of contraceptive use in the Upper East region of Ghana. *Reproductive Health*, 12:29.
- Adegbola, O., & Okunowo, A. (2009). Intended postpartum contraceptive use among pregnant and puerperal women at a university teaching hospital. Archives of Gynecology and Obstetrics, 280(6), 987–992.
- Adeyemi, A. B., Ijadunola, K. T., Orji, E. O., Kuti, O., & Alabi, M. M. (2005). The unmet need for contraception among Nigerian women in the first year post-partum. European J. of Contraception and Reproductive Healthcare, 10(4), 229-234.
- Adhikari, R., Soonthorndhada, K., & Prasartkul, P. (2009). Correlates of unintended pregnancy among currently pregnant married women in Nepal. BMC International Health and Human Rights, 9(1), 17.
- Agha, S. (2010). Intentions to use contraceptives in Pakistan: Implications for behaviour change campaigns. BMC Public Health, 10(1), 450.
- Aidam, B. A., Perez-Escamilla, R., Lartey, A., & Aidam, J. (2005). Factors associated with exclusive breastfeeding in Accra, Ghana. European Journal of Clinical Nutrition, 59(6), 789-796.

- Ajzen, I. (1989). Attitude structure and behaviour. Attitude Structure and Function, 241-274.
- Ajzen, I. (1991). The theory of planned behaviour. Organizational Behavior and Human Decision Processes, 50(2), 179–211.
- Ajzen, I., & Fishbein, M. (1980). Understanding attitudes and predicting social behaviour. Englewood Cliffs, NJ: Prentice-Hall.
- Akalework, B. (2008). Determinants and Magnitude of Unintended Pregnancy among Selected Married Women in Hawassa town. SNNPR, AAU, Addis Ababa.
- Akinyoade, A. (2007). Dynamics of reproductive behaviour in rural coastal communities of Southern Ghana. ISS Ph.D Theses. Maastricht: Shaker Publishing BV. retrieved from http://hdl.handle.net/1765/32623
- Amin Shokravi, F., Howden Chapman, P., & Peyman, N. (2009). A comparison study: Risk factors of unplanned pregnancies in a group of Iranian and New Zealander women. European Journal of Scientific Research, 26(1), 108-121.
- Andersson, G., & Woldemicael, G. (2001). Sex composition of children as a determinant of marriage disruption and marriage formation: Evidence from Swedish register data. *Journal of Population Research*, 18(2), 143–153.
- Anyara, E. L., & Hinde, A. (2006). Fertility transition in Kenya: Regional analysis of the proximate determinants. Southampton UK: Southampton Statistical Sciences Research Institute. 1-55

- Arbab, A. A., Bener, A., & Abdulmalik, M. (2011). Prevalence, awareness and determinants of contraceptive use in Qatari women. *EMHJ*, 17(1), 1-20
- Armitage, C. J., & Conner, M. (2000). Social cognition models and health behaviour: A structured review. *Psychology and Health*, 15(2), 173–189.
- Ashton, L., Giridhar, N., Holcombe, S. J., Madon, T., & Turner, E. (2015). A review of behavioural economics in reproductive health. Retrieved 20th January 2015, from http://www.beri-research.org/wpcontent/uploads/
- Assembly, U. G. (1948). Universal declaration of human rights. UN General Assembly. Retrieved 15th October 2013, from http://www.tandfonline.com/
- Assimeng, M. (1981). Social structure of Ghana: A study in persistence and change. Accra: Ghana Publishing Corporation
- Awusabo-Asare, K. (1990). Matriliny and the new intestate succession law of Ghana. Canadian Journal of African Studies/La Revue Canadienne Des études Africaines, 24(1), 1–16.
- Azmat, S. K., Shaikh, B. T., Hameed, W., Mustafa, G., Hussain, W., Asghar, J., Bilgrami, M. (2013). Impact of social franchising on contraceptive use when complemented by vouchers: a quasi-experimental study in rural Pakistan. *PloS One*, 8(9), e74260.
- Bachrach, C. A., & Newcomer, S. (1999). Forum: Intended Pregnancies and Unintended Pregnancies: Distinct Categories or Opposite Ends of a Continuum? Family Planning Perspectives, 31(5), 251-256.

Bairagi, R. (2001). Effects of sex preference on contraceptive use, abortion and fertility in Matlab, Bangladesh. International Family Planning Perspectives, 137-143.

Bandura, A. (1997). Self-efficacy: The exercise of control. New York: Freeman.

- Bandura, A. (1998). Health promotion from the perspective of social cognitive theory. *Psychology and Health*, 13(4), 623-649.
- Bandura, A. (2004). Health promotion by social cognitive means. Health Education & Behavior, 31(2), 143-164.
- Bandura, A., & Bryant, J. (2002). Social cognitive theory of mass communication. Media Effects: Advances in Theory and Research, 2, 121–153.
- Bawah, A. A. (2002). Spousal communication and family planning behavior in Navrongo: a longitudinal assessment. Studies in Family Planning, 33(2), 185–194.
- Becker, G.S. (1960). An economic analysis of fertility. In NBER (ed.) Demographic and economic change in developed countries. New York: Columbia University Press.
- Bell, L. S., Nathan, A. J., & Peleg, I. (2001). Negotiating culture and human rights. New York: Columbia University Press
- Bennett, I. M., Culhane, J. F., McCollum, K. F., & Elo, I. T. (2006). Unintended rapid repeat pregnancy and low education status: any role for depression and contraceptive use? *American Journal of Obstetrics and Gynecology*, 194(3), 749-754.

- Bhalla, K., Adofo, K., Mock, C. N., Afukaar, F., Appiah, N., & Ebel, B. E. (2012). Non-traditional data sources for injury control: an agenda for action in Ghana. *Injury Prevention*, 18 (4), 277-277.
- Bhatia, J. C. (1984). Sex preference of children in rural Ghana (West Africa). Health and Population: Perspectives and Issues, 7(1), 32-47.
- Biddlecom, A. E., & Fapohunda, B. M. (1998). Covert contraceptive use: prevalence, motivations, and consequences. Studies in Family Planning, 360-372.
- Biraben, J.-N. (1979). Essai sur l'evolution du nombre des hommes. Population (french Edition), 13-25.
- Bjerkeset, O., Nordahl, H. M., Larsson, S., Dahl, A. A., & Linaker, O. (2008). A 4-year follow-up study of syndromal and sub-syndromal anxiety and depression symptoms in the general population. Social Psychiatry and Psychiatric Epidemiology, 43(3), 192–199.
- Bleek, W. (1987). Family and family planning in Southern Ghana. In Sex Roles, Population and Development in West Africa: Policy-related studies on work and demographic issues, ed. Christine Oppong, Portsmouth, New Hampshire: Heinemann. 138–153.
- Bloom, S. S., Wypij, D., & Gupta, M. D. (2001). Dimensions of women's autonomy and the influence on maternal health care utilization in a north Indian city. *Demography*, 38(1), 67-78.
- Bongaarts, J. (1978). A framework for analyzing the proximate determinants of fertility. *Population and Development Review*, 105-132.
- Bongaarts, J. (1993). The relative contributions of biological and behavioural factors in determining natural fertility: a demographers perspective. Oxford, England: Clarendon Press. 9-18
- Bongaarts, J. (2001). Fertility and reproductive preferences in post-transitional societies. *Population and Development Review*, 260–281.
- Bongaarts, J., Cleland, J., Townsend, J. W., Bertrand, J. T., & Das Gupta M. (2012). Family planning programs for the 21st century: Rationale and design. Retrieved from http://www.popline.org/node/552109
- Bongaarts, J., & Potter, R. G. (1983). Fertility, biology, and behavior: An analysis of the proximate determinants. Academic Press New York.
- Bongaarts, J., & Sinding, S. W. (2009). A response to critics of family planning programs. International Perspectives on Sexual and Reproductive Health, 39-44.
- Booker, C. L., Harding, S., & Benzeval, M. (2011). A systematic review of the effect of retention methods in population-based cohort studies. *BMC Public Health*, 11(1), 249.
- Borda, M. R., Winfrey, W., & McKaig, C. (2011). Return to sexual activity and modern family planning use in the extended postpartum period: an analysis of findings from seventeen countries. *African Journal of Reproductive Health*, 14(4),75-82
- Botha, N., & Atkins, K. (2005). An assessment of five different theoretical frameworks to study the uptake of innovations. Paper presented to the 2005

New Zealand Agricultural and Resource Economics Society, Nelson, New Zealand, August 26-27, 2005.

- Bouchard, G. (2005). Adult couples facing a planned or an unplanned pregnancy two realities. *Journal of Family Issues*, 26(5), 619-637.
- Brand, E., Kothari, C., & Stark, M. A. (2011). Factors related to breastfeeding discontinuation between hospital discharge and 2 weeks postpartum. The Journal of Perinatal Education, 20(1), 36.
- Brewer, B. K., & Rimer, B. (2008). Perspectives on health behavior theories that focus on individuals. In Health behavior and health education: theory, research, and practice. (K. Glanz, B. Rimer, & K. Viswanath, Eds.). San Francisco: John Wiley&Sons.
- Brown-Peterside, P., Rivera, E., Lucy, D., Slaughter, I., Ren, L., Chiasson, M. A., & Koblin, B. A. (2001). Retaining hard-to-reach women in HIV prevention and vaccine trials: Project ACHIEVE. *American Journal of Public Health*, 91(9), 1377–1379.
- BUMC (2013). The Transtheoretical Model (Stages of Change). Retrieved August 20, 2015, from http://sphweb.bumc.bu.edu/otlt/MPH-Modules/SB/SB721-Models/SB721-Models 6.
- Buyinza, F., & Hisali, E. (2014). Microeffects of women's education on contraceptive use and fertility: the case of Uganda. Journal of International Development, 26(6), 763-778
- Cain, M., & Mittman, R. (2002). Diffusion of innovation in health care. Oakland CA: California Healthcare Foundation

- Caldwell, J. C., & Caldwell, P. (1987). The cultural context of high fertility in sub-Saharan Africa. *Population and Development Review*, 409–437.
- Calhoun, L. M., Nanda, P., Speizer, I. S., & Jain, M. (2013). The effect of family sex composition on fertility desires and family planning behaviors in urban Uttar Pradesh, India. *Reproductive Health*, 10(1), 48
- Callahan, R., & Becker, S. (2014). Unmet need, intention to use contraceptives and unwanted pregnancy in rural Bangladesh. *International Perspectives* on Sexual and Reproductive Health, 40(1), 4–10.
- Calvert, C., Baisley, K., Doyle, A. M., Maganja, K., Changalucha, J., Watson-Jones, D., Ross, D. A. (2013). Risk factors for unplanned pregnancy among young women in Tanzania. *Journal of Family Planning and Reproductive Health Care*, 39(4):e2
- Cates, W., Karim, Q. A., El-Sadr, W. M., Haffner, D. W., Kalema-Zikusoka, G., Rogo, K., Averill, E. M. D. (2010). Family planning and the millennium development goals. *Science*, 329(5999), 1603.
- Census Bureau, United States. (2011). Population distribution and change: 2000 to 2010. US Department of Commerce, Economics and Statistics Administration, US Census Bureau.
- Centers for Disease Control. (2010). Unintended pregnancy prevention. Home: United States Department of Health and Human Services Retrieved Apr 30th 2010 from http://www.Cdc.gov/reproductivehealth/Unintended Pregnancy/index. Htm.

- Chalmers, B., Levitt, C., Heaman, M., O'Brien, B., Sauve, R., & Kaczorowski, J. (2009). Breastfeeding rates and hospital breastfeeding practices in Canada: a national survey of women. *Birth*, 36(2), 122–132.
- Champion V.L. & Skinner C.S. (2008). "The Health Belief Model" In Glanz K, Rimer BK, Viswanath K, eds. Health behavior and health education: theory, research, and practice. San Francisco: John Wiley & Sons.
- Chaudhuri, S. (2012). The desire for sons and excess fertility: a household-level analysis of parity progression in India. *International Perspectives on Sexual and Reproductive Health*, 178–186.
- Chin, A. C., Myers, L., & Magnus, J. H. (2008). Race, education, and breastfeeding initiation in Louisiana, 2000-2004. Journal of Human Lactation, 24(2), 175–185.
- Cleland, J., Bernstein, S., Ezeh, A., Faundes, A., Glasier, A., & Innis, J. (2006). Family planning: the unfinished agenda. *The Lancet*, 368(9549), 1810– 1827.
- Cleland, J., Phillips, J. F., Amin, S., Kamal, G. M., & others. (1994). The determinants of reproductive change in Bangladesh: success in a challenging environment. World Bank. Retrieved August 21st 2015 from http://www.cabdirect.org/abstracts/19941808810.html
- Cobbah, J. A. (1987). African values and the human rights debate: an African perspective. Human Rights Quarterly, 309-331.
- Coday, M., Boutin-Foster, C., Sher, T. G., Tennant, J., Greaney, M. L., Saunders, S. D., & Somes, G. W. (2005). Strategies for retaining study participants in

behavioral intervention trials: retention experiences of the NIH Behavior Change Consortium. Annals of Behavioral Medicine, 29(2), 55-65.

- Connelly, M. J. (2009). Fatal misconception: The struggle to control world population. Cambridge: Harvard University Press..
- Creanga, A. A., Gillespie, D., Karklins, S., & Tsui, A. O. (2011). Low use of contraception among poor women in Africa: an equity issue. Bulletin of the World Health Organization, 89(4), 258–266.
- Creative Research Systems.(2012). Creative Research Systems Sample Size Calculator.Retrieved August 21,2015,from https://www.conservationgatew ay.org/ExternalLinks/Pages/creative-research-systems.aspx
- Crissman, H. P., Adanu, R. M., & Harlow, S. D. (2012). Women's sexual empowerment and contraceptive use in Ghana. Studies in Family Planning, 43(3), 201–212.
- Crossette, B. (2005). Reproductive health and the millennium development goals: the missing link. Studies in Family Planning, 36(1), 71–79.
- Cuff, E. C., & Francis, D. W. (1978). Some features of "invited stories" about marriage breakdown. International Journal of the Sociology of Language, 1978(18), 111–133.
- Darwish, O., & Sahn, F. el. (1991). Breastfeeding and fertility. Part II: Breastfeeding practices. The Journal of the Egyptian Public Health Association, 67(3-4), 259–273.
- Das Gupta, M., Zhenghua, J., Bohua, L., Zhenming, X., Chung, W., & Hwa-OK, B. (2003). Why is Son Preference so Persistent in East and South Asia? A

Cross-Country Study of China, India and the Republic of Korea. The Journal of Development Studies, 40(2), 153-187

- Das, N. (1987). Sex preference and fertility behavior: A study of recent Indian data. Demography, 24(4), 517-530.
- DaVanzo, J., Hale, L., Razzaque, A., & Rahman, M. (2007). Effects of interpregnancy interval and outcome of the preceding pregnancy on pregnancy outcomes in Matlab, Bangladesh. BJOG: An International Journal of Obstetrics & Gynaecology, 114(9), 1079-1087.
- Davis, K., & Blake, J. (1956). Social structure and fertility: An analytic framework. *Economic Development and Cultural Change*, 211–235.
- De Bruijn, B. J. (2006). Fertility: theories, frameworks, models, concepts. In G. Caselli, J. Vallin, G. Wunsch (Eds). Demography: Analysis and Synthesis. Amsterdam: Elsevier. 549-569
- De Graaf, R., Bijl, R. V., Smit, F., Ravelli, A., & Vollebergh, W. A. (2000). Psychiatric and sociodemographic predictors of attrition in a longitudinal study The Netherlands Mental Health Survey and Incidence Study (NEMESIS). American Journal of Epidemiology, 152(11), 1039–1047.
- DeRose, L. F., Dodoo, F. N.-A., & Patil, V. (2002). Fertility desires and perceptions of power in reproductive conflict in Ghana. Gender & Society, 16(1), 53-73.
- Di Giacomo, P., Sbarlati, A., Bagnasco, A., & Sasso, L. (2013). Woman's contraceptive needs and preferences in the postpartum period: an Italian study. Journal of Clinical Nursing, 22(23-24), 3406-3417.

- Directory of MOH/GHS (n.d). Policies, Standards, Guidelines and Protocols. Retrieved August14,2015, from http://www.ghanaqhp.org/filea
- Doctor, H. V., Phillips, J. F., & Sakeah, E. (2009). The influence of changes in women's religious affiliation on contraceptive use and fertility among the Kassena-Nankana of Northern Ghana. Studies in Family Planning, 40(2), 113-122.
- Donnelly, J.(2013). Universal human rights in theory and practice. Ithaca: Cornell UniversityPress.
- Duncan, S., Edwards, R., & Alexander, C. (2010). Teenage parenthood: what's the problem? Tufnell. Retrieved from http://eprints.soton.ac.uk/189265/
- Duong, D. V., Lee, A. H., & Binns, C. W. (2005). Contraception within six-month postpartum in rural Vietnam: Implications on family planning and maternity services. *European J. of Contraception and Reproductive Healthcare*, 10(2), 111–118.
- Edmeades, J., Pande, R., Macquarrie, K., Falle, T., & Malhotra, A. (2012). Two sons and a daughter: Sex composition and women's reproductive behaviour in madhya pradesh, india. *Journal of Biosocial Science*, 44(06), 749–764.
- Ehrlich, P. R., & Ehrlich, A. H. (2009). The population bomb revisited. The Electronic Journal of Sustainable Development, 1(3), 63-71.
- Exavery, A., Kanté, A. M., Njozi, M., Tani, K., Doctor, H. V., Hingora, A., & Phillips, J. F. (2014). Predictors of mistimed, and unwanted pregnancies

among women of childbearing age in Rufiji, Kilombero, and Ulanga districts of Tanzania. Reproductive Health, 11(1), 63.

- Exavery, A., Mrema, S., Shamte, A., Bietsch, K., Mosha, D., Mbaruku, G., & Masanja, H. (2012). Levels and correlates of non-adherence to WHO recommended inter-birth intervals in Rufiji, Tanzania. BMC Pregnancy and Childbirth, 12(1), 152.
- Fishbein, M. (2008). A reasoned action approach to health promotion. *Medical Decision Making*, 28(6), 834-844.
- Fishbein, M., & Cappella, J. N. (2006). The role of theory in developing effective health communications. *Journal of Communication*, 56(s1), S1–S17.
- Frost, J. J. (2013). US Women's Use of Sexual and Reproductive Health Services: Trends, Sources of Care and Factors Associated with Use, 1995–2010.
 NewYork: Guttmacher Institute.Retrieved 15th August 2014 from http://w www.alanguttmacherinstitute.net/pubs/sources-of-care-2013.pdf
- Frost, J. J., Lindberg, L. D., & Finer, L. B. (2012). Young adults' contraceptive knowledge, norms and attitudes: associations with risk of unintended pregnancy. *Perspectives on Sexual and Reproductive Health*, 44(2), 107– 116.
- Furuta, M., & Salway, S. (2006). Women's position within the household as a determinant of maternal health care use in Nepal. International Family Planning Perspectives, 17–27.
- Fuse, K. (2010). Variations in attitudinal gender preferences for children across 50 less-developed countries. *Demographic Research*, 23(36), 1031–1048.

- Gebreselassie, T., Rutstein, S. O., & Mishra, V. (2008). Contraceptive use breastfeeding amenorrhea and abstinence during the postpartum period: An Analysis of Four Countries. DHS Analytical Studies No. 14. Calverton, Maryland, USA: Macro International Inc.
- Ghana Health Policy Project (2012). Ghana GAP Tool: Brief-Health Policy Initiative. Retrieved from http://www.healthpolicyproject.com/ps/111
- Ghana Health Service, Central Region (2011). Family Health Report 2010. Accra: Ghana Health Service
- Ghana Health Service, Family Health Division (2015). Family Planning Programme. Accra: Ghana Health Service.
- Ghana Health Service, Municipal Health Directorate, Mfantseman (2010). Annual Report 2010. Accra: Ghana Health Service
- Ghana: Market Segmentation Analysis (2011). Arlington Va: USAID|DELIVER PROJECT, Task Order 1, 2010.
- Ghana Statistical Service (2012). 2010 Population & Housing Census of Ghana: Summary Report of Final Results. Accra: Ghana Statistical Service.
- Ghana Statistical Service (2013). National Analytical Report. Retrieved 23rd December 2015, from http://www.statsghana.gov.gh/docfiles/2010phc/
- Gill, K., Pande, R., & Malhotra, A. (2007). Women deliver for development. The Lancet, 370(9595), 1347-1357.
- Glanz, K., Rimer, B. K., & Viswanath, K. (2008). Health behavior and health education: theory, research, and practice. San Francisco: John Wiley & Sons.

.

Glasier, A., Gülmezoglu, A. M., Schmid, G. P., Moreno, C. G., & Van Look, P. F. (2006). Sexual and reproductive health: a matter of life and death. *The Lancet*, 368(9547), 1595–1607.

.

- Godfrey, E. M., Chin, N. P., Fielding, S. L., Fiscella, K., & Dozier, A. (2011). Contraceptive methods and use by women aged 35 and over: A qualitative study of perspectives. *BMC Women's Health*, 11(1), 5.
- Goicolea, I., & San Sebastian, M. (2010). Research Unintended pregnancy in the amazon basin of Ecuador: a multilevel analysis. *International Journal For Equity in Health*, 9(1), 1.
- Gollwitzer, P. M. (1999). Implementation intentions: strong effects of simple plans. American Psychologist, 54(7), 493.
- Gray, E., & Evans, A. (2004). Sex composition of children as a determinant of parity progression. Australian National University, Canberra: Negotiating the Life Course Project.
- Gribble, J. (2010). Financing contraceptives: a new funding environment. Retrieved from http://www.popline.org/node/210220
- Grol, R. P., Bosch, M. C., Hulscher, M. E., Eccles, M. P., & Wensing, M. (2007). Planning and studying improvement in patient care: the use of theoretical perspectives. *Milbank Quarterly*, 85(1), 93-138.
- Grown, C., Gupta, G. R., & Kes, A. (2005). Taking Action: Achieving gender equality and empowering women. London, England, Earthscan 2005, 279.

- GSS, G., Macro, & I. C. F. (2009). Ghana demographic and health survey 2008. Accra, Ghana: Ghana Statistical Service, Ghana Health Service, and ICF Macro.
- GSS, G., Macro, & I. C. F. (2014). Ghana demographic and health survey 2014. Accra, Ghana: Ghana Statistical Service, Ghana Health Service, and ICF Macro.
- Gupta, M. Das, Zhenghua, J., Bohua, L., Zhenming, X., Chung, W., & Hwa-Ok,
 B. (2003). Why is son preference so persistent in East and South Asia? A cross-country study of China, India and the Republic of Korea. *The Journal of Development Studies*, 40(2), 153-187.
- Gustavson, K., von Soest, T., Karevold, E., & Røysamb, E. (2012). Attrition and generalizability in longitudinal studies: findings from a 15-year populationbased study and a Monte Carlo simulation study. BMC Public Health, 12(1), 918.
- Guzzo, K. B., & Hayford, S. R. (2014). Fertility and the Stability of Cohabiting Unions Variation by Intendedness. Journal of Family Issues, 35(4), 547– 576.
- Gwatkin, D. R., Rutstein, S., Johnson, K., Suliman, E., Wagstaff, A., & Amouzou,
 A. (2007). Socio-economic differences in health, nutrition, and population.
 Washington, DC: The World Bank.
- Gyimah, S. O., Takyi, B. K., & Addai, I. (2006). Challenges to the reproductivehealth needs of African women: On religion and maternal health utilization in Ghana. Social Science & Medicine, 62(12), 2930–2944.

- Hamdela, B., Tilahun, T., & others. (2012). Unwanted pregnancy and associated factors among pregnant married women in Hosanna Town, Southern Ethiopia. *PLOS ONE*, 7(6), e39074.
- Hartmann, M., Gilles, K., Shattuck, D., Kerner, B., & Guest, G. (2012). Changes in couples' communication as a result of a male-involvement family planning intervention. *Journal of Health Communication*, 17(7), 802–819.
- Hochbaum, G. M. (1958). Public participation in medical screening programs: a socio-psychological study. US Department of Health, Education, and Welfare, Public Health Service, Bureau of State Services, Division of Special Health Services, Tuberculosis Program.
- Hunt, J. R., & White, E. (1997). Retaining and tracking cohort study members. Epidemiologic Reviews, 20(1), 57-70.
- Iaquinta, D. L., & Drescher, A. W. (2000). Defining periurban: understanding rural-urban linkages and their connection to institutional contexts. Land Reform 2, 8-27.
- Ikamari, L., Izugbara, C., & Ochako, R. (2013). Prevalence and determinants of unintended pregnancy among women in Nairobi, Kenya. BMC Pregnancy and Childbirth, 13(1), 69.
- International Planned Parenthood Federation (2012). *IPPF_FactSheet-6_Human Rights*. Retrieved from http://www.countdown2015europe.org/wpcontent/ upLoads/2012/03/IPPF

- Jabeen, M., Gul, F., Wazir, F., & Javed, N. (2012). Knowledge, attitude and practices of contraception in women of reproductive age. Gomal Journal of MedicalSciences, 9(2), 113-116
- Jamison, D. T., Breman, J. G., Measham, A. R., Alleyne, G., Claeson, M., Evans, D. B., & Musgrove, P. (2006). Disease control priorities in developing countries. World Bank Publications. Retrieved 20th November 2014, from http://books.google. com/books?
 - Johnson, F. A., & Madise, N. J. (2011). Targeting women at risk of unintended pregnancy in Ghana: Should geography matter? Sexual & Reproductive Healthcare, 2(1), 29-35.
- Joyce, T., Kaestner, R., & Korenman, S. (2000). The stability of pregnancy intentions and pregnancy-related maternal behaviors. *Maternal and Child Health Journal*, 4(3), 171–178.
- Jutte, D. P., Roos, N. P., Brownell, M. D., Briggs, G., MacWilliam, L., & Roos, L. L. (2010). The ripples of adolescent motherhood: social, educational, and medical outcomes for children of teen and prior teen mothers. Academic Pediatrics, 10(5), 293-301.
- Kariuki, W. (2011). Factors determining the intention to use postpartum family planning among the first time mothers aged 15-29 years in Kenya. Retrieved from http://docs.mak.ac.ug/sites/
- Kendall, C., Afable-Munsuz, A., Speizer, I., Avery, A., Schmidt, N., & Santelli, J. (2005). Understanding pregnancy in a population of inner-city women in

New Orleans—results of qualitative research. Social Science & Medicine, 60(2), 297-311.

- Keogh, S. C., Urassa, M., Roura, M., Kumogola, Y., Kalongoji, S., Kimaro, D., ... Zaba, B. (2012). The impact of antenatal HIV diagnosis on postpartum childbearing desires in northern Tanzania: a mixed methods study. *Reproductive Health Matters*, 20(39), 39-49.
- Kohler, H.-P. (2012). Population Growth. PSC Working Paper PSC 12-03 and Copenhagen Consensus 2012 Challenge Paper. Population Studies Centre, University of Pennsylvania and Copenhagen Consensus Centre, Copenhagen.
- Kole, E. S. (2002). Connecting women from developing countries to the internet: searching for an appropriate paradigm. In 41st Annual Convention of the International Studies Association (14-17 March). (Vol. 20). Retrieved from http://globalsolidarity.antenna.nl/kole.html
- Kollodge, R. (2011). State of world population 2011: People and possibilities. New York: United Nations Population Fund (UNFPA), 92-99.
- Korenman, S., Kaestner, R., & Joyce, T. J. (2001). Unintended pregnancy and the consequences of nonmarital childbearing. *Out of Wedlock: Causes and Consequences of Nonmarital Fertility.* New York: Russell Sage Foundation.
- Krishnamurty, J. (2000). Population Policy Development and the ICPD Programme of Action. Low Fertility and Policy Responses to Issues of Ageing and Welfare, 24-53.

- Kumi-Kyereme, A., Awusabo-Asare, K., Biddlecom, A., & Tanle, A. (2007). Influence of social connectedness, communication and monitoring on adolescent sexual activity in Ghana. *African Journal of Reproductive Health*, 11(1), 133-147.
- Lehrer, E. L. (2004). Religion as a determinant of economic and demographic behavior in the United States. *Population and Development Review*, 30(4), 707-726.
- Lockwood, M. (1995). Development policy and the African demographic transition: Issues and questions. Journal of International Development, 7(1), 1-23.
- Lomborg, B. (2001). The Skeptical Environmentalist: measuring the state of the real world. Cambridge University Press, Cambridge, UK.
- Luker, K. C. (1999). Forum: A reminder that human behavior frequently refuses to conform to models created by researchers. Family Planning Perspectives, 31(5), 248-249.
- Makinwa-Adebusoye, P. (2007). Sociocultural factors affecting fertility in sub-Saharan Africa. United Nations (2007a). Prospects for Fertility Decline in High Fertility Countries, Population Bulletin of the United Nations, Special Issue, (46/47), 55-69.
- Malarcher, S. (2010). Social determinants of sexual and reproductive health: Informing future research and programme implementation. World Health Organization (WHO). Geneva, Switzerland.

- Malik, S., & Courtney, K. (2011). Higher education and women's empowerment in Pakistan. Gender and Education, 23(1), 29-45.
- Marshall, S. J., & Biddle, S. J. (2001). The transtheoretical model of behavior change: a meta-analysis of applications to physical activity and exercise. Annals of Behavioral Medicine, 23(4), 229-246.
- Mayer, A. E., & Mayor, A. E. (1991). Islam and human rights: Tradition and politics. Boulder, Colorado: Westview Press.
- McAlister, A. L., Perry, C. L., & Parcel, G. S. (2008). How individuals, environments, and health behaviors interact: Social cognitive theory. In K. Glanz, B.K. Rimer, K. Viswanath (Eds.), *Health behaviour and health education: Theory, research and practice (4th ed., pp. 169-188).* San Francisco, CA: Jossey-Bass.
- McGuire, W. J. (1983). A contextualist theory of knowledge: Its implications for innovation and reform in psychological research. Advances in Experimental Social Psychology, 16, 1–47.
- McInnes, R. J., & Chambers, J. A. (2008). Supporting breastfeeding mothers: qualitative synthesis. *Journal of Advanced Nursing*, 62(4), 407–427.
- McKaig, C., & Chase, R. (2007). Postpartum Family Planning Technical Consultation: Meeting report Washington DC 14 November 2006. Synthesizing lessons learned and building consensus on effective approaches for postpartum family planning. Retrieved January 18th 2013 from http://www.popline.org/node/188770.

- McLeroy, K. R., Bibeau, D., Steckler, A., & Glanz, K. (1988). An ecological perspective on health promotion programs. *Health Education & Behavior*, 15(4), 351-377.
- Meeting the Commodity Challenge: Ghana's National Reproductive Health Commodity Security Strategy 2011–2016 (n.d.). Retrieved Sept 20 2014 from http://deliver.jsi.com/dlvr_content/resources/
- Michie, S., Johnston, M., Abraham, C., Lawton, R., Parker, D., & Walker, A. (2005). Making psychological theory useful for implementing evidence based practice: a consensus approach. *Quality and Safety in Health Care*, 14(1), 26–33.
- Mwangi, J. (2012). Preliminary findings from the 2011 National MICs to the Health Summit: is Ghana reducing disparities and improving equity on key health outcomes. *Ghana Annual Health Summit*, 20–22.
- National Strategic Plan (2014-2020). Family Planning in the DRC. Retrieved August 19, 2015, from http://familyplanning-drc.net/plan-strategique.php
- Ndugwa, R. P., Cleland, J., Madise, N. J., Fotso, J.-C., & Zulu, E. M. (2011). Menstrual pattern, sexual behaviors, and contraceptive use among postpartum women in Nairobi urban slums. *Journal of Urban Health*, 88(2), 341-355.
- Nilsen, R. M., Vollset, S. E., Gjessing, H. akon K., Skj\a erven, R., Melve, K. K., Schreuder, P., Magnus, P. (2009). Self-selection and bias in a large prospective pregnancy cohort in Norway. *Paediatric and Perinatal Epidemiology*, 23(6), 597-608.

- Noar, S. M., & Zimmerman, R. S. (2005). Health Behavior Theory and cumulative knowledge regarding health behaviors: are we moving in the right direction? *Health Education Research*, 20(3), 275–290.
- Nukunya, G. K. (2003). Tradition and Change in Ghana. Accra: Universities Press.
- Okonofua, F. E., Odimegwu, C., Ajabor, H., Daru, P. H., & Johnson, A. (1999). Assessing the prevalence and determinants of unwanted pregnancy and induced abortion in Nigeria. *Studies in Family Planning*, 30(1), 67–77.
- Oldenburg, B., & Glanz, K. (2008). Diffusion of innovations. Health Behavior and Health Education Theory, Research and Practice, 313-333.
- Omane-Adjepong, M., Oduro, F.T., Annin, K. (2012). A multinomial regression analysis of unplanned pregnancies in Ahafo Ano South District, Ghana.. *American International Journal of Contemporary Research 12(2)*.
- Oppong, C. (1981). Middle Class African Marriage: A Family Study of Ghanalan Senior Civil Servants. London: Allen and Unwin,.
- Orji, R., Vassileva, J., & Mandryk, R. (2012). Towards an effective health interventions design: an extension of the health belief model. Online Journal of Public Health Informatics, 4(3). Retrieved 17 jun 2013 from http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3615835/
- Orlandi, M. A., Landers, C., Weston, R., & Haley, N. (1990). Diffusion of health promotioninnovations.Retrieved 16thfeb. 2013 from http://psycnet.apa.org/ psycinfo/1990-98174-012

- Pemberton, H. E. (1936). The curve of culture diffusion rate. American Sociological Review, 1(4), 547-556.
- Perrin, A. K. (2005). Human Rights and Cultural Relativism: The "Historical Development" Argument and Building a Universal Consensus. Retrieved 24 August 2014 from http://www.academia.edu/download/30331911/
- Persad, M. D., & Mensinger, J. L. (2008). Maternal breastfeeding attitudes: association with breastfeeding intent and socio-demographics among urban primiparas. *Journal of Community Health*, 33(2), 53-60.
- Piotrow, P. T., Kincaid, D. L., Rimon, J. G., Rinehart, W., Samson, K., & others. (1997). Health communication: lessons from family planning and reproductive health. Retrieved Sept 19th 2013 from http://www.popline.org /node/269405
- Pollard, I. (1994). A guide to reproduction: social issues and human concerns. Cambridge: Cambridge University Press.
- Pollard, M. S., & Morgan, S. P. (2002). Emerging parental gender indifference? Sex composition of children and the third birth. *American Sociological Review*, 67(4), 600.
- Prata, N., Vahidnia, F., Potts, M., & Dries-Daffner, I. (2005). Revisiting community-based distribution programs: are they still needed? *Contraception*, 72(6), 402–407.

Pritchett, L. H. (1994). Desired fertility and the impact of population policies. Population and Development Review, 1-55.

- Prochaska, J. O., DiClemente, C. C., & Norcross, J. C. (1992). In search of how people change: Applications to addictive behaviors. American Psychologist, 47(9), 1102.
- Pulley, L., Klerman, L. V., Tang, H., & Baker, B. A. (2002). The extent of pregnancy mistiming and its association with maternal characteristics and behaviors and pregnancy outcomes. *Perspectives on Sexual and Reproductive Health*, 206–211.
- Qian, N. (2008). Missing women and the price of tea in China: The effect of sexspecific earnings on sex imbalance. The Quarterly Journal of Economics, 123(3), 1251-1285.
- Rajaretnam, T., & Deshpande, R. V. (1994). The effect of sex preference on contraceptive use and fertility in rural South India. *International Family Planning Perspectives*, 88–95.
- Regmi, P. R., Van Teijlingen, E., Simkhada, P., & Acharya, D. R. (2010). Barriers to sexual health services for young people in Nepal. Journal of Health, Population, and Nutrition, 28(6), 619-627
- Renfrew, M. J., Pokhrel, S., Quigley, M., McCormick, F., Fox-Rushby, J., Dodds,
 R., Williams, A. (2012). Preventing disease and saving resources: the potential contribution of increasing breastfeeding rates in the UK.
 UNICEF. Retrieved from http://bura.brunel.ac.uk/handle/2438/10266
- Reproductive Health Supplies Coalition. (2009). Contraceptive Projections and the Donor Gap. Meeting the Challenge. RHSC Belgium.

- Research, N. I., (Bangladesh) (2005). Bangladesh Demographic and Health Survey. Bangladesh: National Institute of Population Research and Training.
- Ringheim, K. (2011). Integrating Family Planning and Maternal and Child Health Services: History Reveals a Winning Combination. Population Reference Bureau, July.
- Robinson, K. A., Dennison, C. R., Wayman, D. M., Pronovost, P. J., & Needham,
 D. M. (2007). Systematic review identifies number of strategies important
 for retaining study participants. *Journal of Clinical Epidemiology*, 60(8),
 757-e1.
- Rogers, E. (2008). M, (2003), Diffusion of innovations. New York: Free Press.
- Rosenfeld, J. A., & Everett, K. D. (1996). Factors related to planned and unplanned pregnancies. *The Journal of Family Practice*. Retrieved from http://psycnet.apa.org/psycinfo/1996-05954-002
- Ross, J. A., & Winfrey, W. L. (2001). Contraceptive use, intention to use and unmet need during the extended postpartum period. *International Family Planning Perspectives*, 20-27.
- Roudi-Fahimi, F. (2002). Iran's family planning program: responding to a nation's needs. Population Reference Bureau Washington DC. Retrieved from http://rezvanhealth.com/fa
- Rujumba, J., Neema, S., Byamugisha, R., Tylleskär, T., Tumwine, J. K., & Heggenhougen, H. K. (2012). "Telling my husband I have HIV is too heavy to come out of my mouth": pregnant women's disclosure

experiences and support needs following antenatal HIV testing in eastern Uganda. Journal of the International AIDS Society, 15(2). Retrieved from http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3494159/.

- Rutstein, S. O. (2008). Further evidence of the effects of preceding birth intervals on neonatal infant and under-five-years mortality and nutritional status in developing countries: Evidence from the Demographic and Health Surveys. Retrieved from http://www.popline.org/node/209750.
- Sachs, J. D., & McArthur, J. W. (2005). The millennium project: a plan for meeting the millennium development goals. *The Lancet*, 365(9456), 347– 353.
- Santelli, J., Rochat, R., Hatfield-Timajchy, K., Gilbert, B. C., Curtis, K., Cabral, R., & Schieve, L. (2003). The measurement and meaning of unintended pregnancy. *Perspectives on Sexual and Reproductive Health*, 35(2), 94– 101.
- Sathar, Z. A., & Phillips, J. F. (2001). Fertility Transition in South Asia. Oxford UniversityPress.Retrievedfromhttp://econpapers.repec.org/bookchap/oxpo books/9780199241859.html.
- Schoenborn, C. A. (2004). Marital status and health, United States 1999-2002. Citeseer. Retrieved from http://citeseerx.ist.psu.edu/viewdoc/

Sedgh, G. (2010). Abortion in Ghana. Issues in Brief (Alan Guttmacher Institute),

^{(2), 1-4.}

- Sedgh, G., Bankole, A., Oye-Adeniran, B., Adewole, I. F., Singh, S., & Hussain, R. (2006). Unwanted pregnancy and associated factors among Nigerian women. International Family Planning Perspectives, 175–184.
- Sharma, V., Mohan, U., Das, V., & Awasthi, S. (2012). Socio demographic determinants and knowledge, attitude, practice: Survey of family planning. Journal of Family Medicine and Primary Care, 1(1), 43.
- Sinding, S. W. (2007). Overview and Perspectives. In The global family planning revolution: three decades of population policies and programs. (W. C. Robinson & J. A. Ross, Eds.). World Bank Publications.
- Singh, S., Darroch, J. E., Ashford, L. S., & Vlassoff, M. (2009). Adding It Up-The Benefits of Investing in Family Planning and Newborn and Maternal Health. New York: Guttmacher Institute and United Nations Population Fund.
- Singh, S., Sedgh, G., & Hussain, R. (2010). Unintended pregnancy: worldwide levels, trends, and outcomes. *Studies in Family Planning*, 41(4), 241-250.
- Singh, S., Wulf, D., Hussain, R., Bankole, A., & Sedgh, G. (2009). Abortion worldwide: a decade of uneven progress. New York: Guttmacher Institute.
- Smith, R., Ashford, L., Gribble, J., & Clifton, D. (2009). Family planning saves lives. Washington D.C: Population Reference Bureau
- Speidel, J. J., Sinding, S., Gillespie, D., Maguire, E., & Neuse, M. (2009). Making the case for US international family planning assistance. Baltimore: Bill and Melenda Gates Institute for Population and Reproductive Health, Johns

Hopkins School of Public Health. Retrieved September, 30, 2010 from http://www.jhsph.edu/gatesinstitute/_pdf/policy_practice/papers

- Speizer, I. S., Fotso, J. C., Okigho, C., Faye, C. M., & Seck, C. (2013). Influence of integrated services on Postpartum Family Planning use: a cross-sectional survey from urban Senegal. *BMC Public Health* 13:752
- Stanford, J. B., Hobbs, R., Jameson, P., DeWitt, M. J., & Fischer, R. C. (2000). Defining dimensions of pregnancy intendedness. *Maternal and Child Health Journal*, 4(3), 183–189.
- Stover, J., & Ross, J. (2010). How increased contraceptive use has reduced maternal mortality. *Maternal and Child Health Journal*, 14(5), 687-695.
- Swanson, V., & Power, K. G. (2005). Initiation and continuation of breastfeeding: theory of planned behaviour. *Journal of Advanced Nursing*, 50(3), 272– 282.
- Takwimu, T. O. ya T. ya, & Macro, I. C. F. (2011). Tanzania Demographic and Health Survey, 2010. National Bureau of Statistics.
- Tambs, K., Rønning, T., Prescott, C. A., Kendler, K. S., Reichborn-Kjennerud, T., Torgersen, S., & Harris, J. R. (2009). The Norwegian Institute of Public Health twin study of mental health: examining recruitment and attrition bias. Twin Research and Human Genetics, 12(02), 158–168.
- Taplin, S. (2005). Methodological design issues in longitudinal studies of children and young people in out-of- home care. Retrieved from http://flosse.fahcsia .gov.au/fahcsiajspui/

- Taveras, E. M., Capra, A. M., Braveman, P. A., Jensvold, N. G., Escobar, G. J., & Lieu, T. A. (2003). Clinician support and psychosocial risk factors associated with breastfeeding discontinuation. *Pediatrics*, 112(1), 108-115.
- Taylor Jr, H. C., & Berelson, B. (1971). Comprehensive family planning based on maternal/child health services: a feasibility study for a world program. Studies in Family Planning, 2(2), 22-54.
- Tensou, B., Hailemariam, A., & Reniers, G. (2008). The Prevalence of Covert use of contraceptives in Nazareth/Adama town. retrieved from http://www.researchgate.net/
- Teson, F. R. (1984). Internatioal Human Rights and Cultural Relativism. Va. J. Int'l L., 25, 869.
- Thant, U. (1967). Declaration on population: The world leaders statement. Studies in Family Planning, 26, 1-3.
- Thomson, E. (2005). Partnerships and parenthood: a comparative view of cohabitation, marriage, and childbearing. Alan Booth and Ann C. Crouter (eds) The New Population Problem: Why Families in Developed Countries Are Shrinking and What It Means, Mahwah, NJ: Lawrence Erlbaum, 129–49.
- Thulier, D., & Mercer, J. (2009). Variables associated with breastfeeding duration. Journal of Obstetric, Gynecologic, & Neonatal Nursing, 38(3), 259-268.
- Thygesen, L. C., Johansen, C., Keiding, N., Giovannucci, E., & Grønb\a ek, M. (2008). Effects of sample attrition in a longitudinal study of the association

between alcohol intake and all-cause mortality. Addiction, 103(7), 1149-1159.

- Torvik, F. A., Rognmo, K., & Tambs, K. (2012). Alcohol use and mental distress as predictors of non-response in a general population health survey: the HUNT study. Social Psychiatry and Psychiatric Epidemiology, 47(5), 805– 816.
- Trussell, J., Vaughan, B., & Stanford, J. (1999). Are all contraceptive failures unintended pregnancies? Evidence from the 1995 National Survey of Family Growth. Family Planning Perspectives, 246–260.
- Ukegbu, A. U., Ukegbu, P. O., Onyeonoro, U. U., & Ubajaka, C. F. (2011).
 Determinants of breastfeeding patterns among mothers in Anambra State, Nigeria. South African Journal of Child Health, 5(4), 112-116.
- United Nations, General Assembly, (1948). Universal Declaration of human rights. United Nations.
- United Nations. (1968). Proclamation of Teheran, Final Act of the International Conference on Human Rights, Teheran, 22 April to 13 May 1968, U.N. Doc. A/CONF. 32/41 at 3 (1968). Retrieved from https://www1.umn.edu/ humanrts/instree/l2ptichr.htm
- United Nations. (2011). Preventing gender-biased sex selection: an interagency statement-OHCHR, UNFPA, UNICEF, UN Women and WHO. Retrieved from http://apps.who.int/iris/handle/10665/44577
- United Nations, General Assembly., (1993). Vienna Declaration and Programme of Action. United Nations.

- United for Human Rights. (n. d.). UN Universal Declaration of Human Rights: Retrieved from http://www.humanrights.com/what-are-human-rights/ universal-declaration-of-human-rights.html
- University of Cape Coast, GIS Unit. (2014).. Map of Mfantseman Municipal. Cape Coast: UCC
- USAID (2009). USAID Banking on Health Project Report. Retrieved from http://www.shopsproject.org/sites/default/files/resources/BoH_End_of_Pro ject_Report.
- Van Loon, A. J. M., Tijhuis, M., Picavet, H. S. J., Surtees, P. G., & Ormel, J. (2003). Survey non-response in the Netherlands: effects on prevalence estimates and associations. *Annals of Epidemiology*, 13(2), 105-110.
- Verbrugge, L. M. (1979). Marital status and health. Journal of Marriage and the Family, 267-285.
- Vlassoff, M., Singh, S., Darroch, J. E., Carbone, E., & Bernstein, S. (2004). Assessing costs and benefits of sexual and reproductive health interventions. Alan Guttmacher Institute New York.
- Waldron, I., Hughes, M. E., & Brooks, T. L. (1996). Marriage protection and marriage selection—prospective evidence for reciprocal effects of marital status and health. Social Science & Medicine, 43(1), 113-123.
- Waldron, I., Weiss, C. C., & Hughes, M. E. (1997). Marital status effects on health: Are there differences between never married women and divorced and separated women? Social Science & Medicine, 45(9), 1387-1397.

Warenius, L. U., Faxelid, E. A., Chishimba, P. N., Musandu, J. O., Ong'any, A. A., & Nissen, E. B. (2006). Nurse-midwives' attitudes towards adolescent sexual and reproductive health needs in Kenya and Zambia. *Reproductive Health Matters*, 14(27), 119–128.

- Warren, C., Mwangi, A., Oweya, E., Kamunya, R., & Koskei, N. (2010). Safeguarding maternal and newborn health: improving the quality of postnatal care in Kenya. *International Journal for Quality in Health Care*, 22(1), 24-30.
- Waterman, A.(2004). Commentary on Everett Rogers Diffusion of Innovations. Retrieved 12th July 2016 from www.stanford.edu/class/symbsys205/comm entary-Rogers DiffusionInnovations.html
- Webb, T. L., & Sheeran, P. (2006). Does changing behavioral intentions engender behavior change? A meta-analysis of the experimental evidence. *Psychological Bulletin*, 132(2), 249.
- Weinstein, N. D., & Rothman, A. J. (2005). Commentary: Revitalizing research on health behavior theories. *Health Education Research*, 20(3), 294–297.
- Williamson, N. E. (1976). Sons or daughters: a cross-cultural survey of parental preferences. Beverly Hills: Sage.
- World Health Organization. (2013). Programming strategies for postpartum family planning. Geneva, Switzerland: WHO Press
- World Health Organization (2010). Monitoring the building blocks of health systems: A handbook of indicators and their measurement strategies. Geneva, Switzerland: WHO Press

World Health Organization. (2006). Defining sexual health: rep ort of a technical consultation on sexual health. 28-31 January 2002, Geneva. Geneva, Switzerland: WHO Press

- World Health Organization. (2006). Report of a WHO Technical Consultation on Birth Spacing, Geneva, Switzerland, 13-15 June 2005. Geneva, Switzerland: WHO Press
- Wolfe, R. A. (1994). Organizational innovation: Review, critique and suggested research directions*. Journal of Management Studies, 31(3), 405-431.
- Yihunie Lakew, A. A., Tamene, H., Benedict, S., & Deribe, K. (2013). Geographical variation and factors influencing modern contraceptive use among married women in Ethiopia: evidence from a national population based survey. *Reproductive Health* 10:52
- Yonas, A. (2005). Correlates of contraceptive use among currently married women in Assosa town of Benishangul Gumuz Regional State Ethiopia. Unpublished Master Thesis, AAU, AA.
- Zehner, O. (2011). The Environmental Politics of Population and Overpopulation. Retrieved 15th July 2013 from http://www.academia.edu/download/302465 48/Zehner_2011_Environmental_Politics_of_Population_and_Overpopulat ion_proof.pdf
- Zimmermann, B. J. (2000). Attaining Self-Regulation: a social cognitive perspective. Handbook of Self-Regulation, 13-39.

APPENDICES

1: Ethical Clearance

GHANA HEALTH SERVICE ETHICAL REVIEW COMMITTEE

In cure of reply the number and saie of this Letter should be special

My Ref. :GHS-ERC: 3 Your Ref. No

Fax + 213-03/02-685424 Ha-Our Ecropea h.Frimpoog@zhxmail.org

December 23, 2011

Research & Development Division

Ohana Health Service P. O. Box MB 190 Acera

Tel +223-0392-681100

Dr. Joseph Awuku-Asmbre/Dr. Frank Balden Centre for Bealth Ruscarch and Implementation Support (CHRIS-GHANA) P. O. Bas KB 584, Korie Bu Access

ETHICAL CLEARANCE - ID NO: GHS-ERC: 14/09/11

The Ghuna Health Service Ethics Review Committee has reviewed and given approval for the implementation of your Study Protocol titled:

"Intended contraceptive use among pregnant women presenting at health facilities in the Mfantseman district of the Central Region of Ghana"

This approval requires that you submit periodic review of the protocol to the Committee and a final full review to the Ethical Review Committee (ERC) on completion of the study. The ERC may observe or cause to be observed procedures and records of the study during and after implementation.

Please note that any modification of the project must be submitted to the ERC for review and approval before its amplementation.

You are also required to report all scrious adverse events related to this study to the ERC within seven days verbally and fourteen days in writing.

You are requested to submit a final report on the study to assure the ERC that the project was implemented as per approved protocol. You are also to inform the ERC and your mother organization before any publication of the research findings.

Please aways quote the protocol identification number in all future correspondence in relation to this protocol

Ce: The Director, Research & Development Division, Ghana Health Service, Acera

0 SIGNED PROFESSOR FRED BINKA (GHS-ERC CHAIRMAN)



2: Phase 1 Questionnaire

QUESTIONNAIRE PHASE 1

Intendeded contraceptive use among pregnant women presenting at health facilities in the Mfantseman Municipal of the Central Region of Ghana FACILITY CODE: I_I INTERVIEWER INITIAL: I_I_I CODE OF RESPONDENT: I I I I

INTRODUCTION

The Centre for Health Research and Implementation Support (CHRIS-Ghana) in conjunction with the Municipal Health Directorate and Saltpond Government Hospital are conducting a study in the intention of pregnant women to use contraceptive methods after delivery in the Mfantseman municipality. The objective is to assess the intentions of women who deliver to adopt modern family planning methods. The findings of the study will help the Ghana Health Service and other health agencies to design post-delivery family planning services in the Mfantseman Municipality and other parts of Ghana. The study is not a part of routine health services.

You were interviewed a few months ago when you were pregnant. . We are inviting you to participate in this follow up study because you agreed to do so during the first study.

PROCEDURES

A trained research staff will use a questionnaire to ask questions related your use of various family planning methods after delivery. He/she will ask you questions related to your behaviours after delivery related to family planning use, breastfeeding and sexual abstinence. You will also be asked questions on health related issues that are known to influence the use of family planning by women who have just delivered.

As you respond to the questions that staff will ask, he/she will fill the questionnaire. The interview will take about 20mins to complete.

As part of the study, you may be invited to participate in a group discussion with other women. Such discussion will be audio taped for the sole purpose of enabling transcription. They will be destroyed as soon as transcription is completed.

RISK AND BENEFITS

By taking time off your schedule to answer the questions, we will be causing you inconvenience; we will however minimize this by going strictly according to the questionnaire guide and completing on time to enable you go about your duties.

Some of the questions may require information that may be quite personal. Should you feel uncomfortable with such questions, you may opt not to answer them.

At the end of the questions, you will be offered the opportunity to ask questions that you may have on family planning. The staff will answer as much as he/she is trained to do so. No money or other reward will be given to you as result of your participation in the study.

CONFIDENTIALITY

All the information that you provide to complete the questionnaire will be considered strictly confidential. It will be available to only persons directly connected with the study. The forms on which the information will be collected will be kept under lock and key/ no information (e.g. name or hospital ID) that identifies and matches you with the responses on the form will be collected.

VOLUNTARINESS AND WITHDRAWAL

Your participation in the study is completely voluntary and you reserve the right not to participate, or even when you have started, to stop answering the question. This is your right and the decision you take will not be disclosed to anyone. It will also not affect the care that will be offered you at this facility now or future.

WHO TO CONTACT

This study has been approved by the Ethics committee of Ghana Health Services. If you wish to ask questions, or need further explanations later, you may contact Dr. Frank Baiden (0244591181) of the Ghana Health Sevice, Dr. Sebastian Eliason (0200975567) of the School of Medical Sciences in Cape Coast, Dr. Derek Bonsu of the Saltpond Government Hospital or Madam Yvonne Graham-Hayfron of the Municipal Health Directorate. You may also contact the administrator of the ethical review committee of the Ghana Health Service on Tel 0302681109

Part 2: CONSENT DECLARATION

"I have read the information given above, or the inform	nation above has been read
and explained to me in a language I understand. I have	been given a chance to ask
questions concerning this study and the questions h	ave been answered to my
satisfaction. I now voluntarily agree to participate in thi	s study knowing that I have
the right to withdraw at any time without affecting futu	re health care services"
Name of respondent/participant:	
Participant's signature	Participant's left thumb
print	
OR	
Date://	
Name of witness Signature	Date:
	Deter
Name of Investigator Signature	Date:

Respondents unable to read or write need witnesses. Respondents who are able should sign and date personally

Part 2: CONSENT DECLARATION

"I have read the information given above, or the information above has been read
and explained to me in a language I understand. I have been given a chance to ask
questions concerning this study and the questions have been answered to my
satisfaction. I now voluntarily agree to participate in this study knowing that I have
the right to withdraw at any time without affecting future health care services"
Name of respondent/participant:
Participant's signature Participant's left thumb
print
OR
Date:/
Name of witnessDate:
Name of InvestigatorDate:

Respondents unable to read or write need witnesses. Respondents who are able should sign and date personally

.

Intended contraceptive use among pregnant women presenting at health facilities in the Mfantseman district of the Central Region of Ghana FACIITY CODE: 1_1 INTERVIEWER INITIAL: 1_1 CODE OF RESPONDENT: 1_1_1

A. RESPONDENT INFORMATION: ANC REGISTRATI

PLE	ASE CHECK: Has this pregna	ant women being in the station NO:		
1	Age	If YES, STOP INTERVIEW HERE	D Yes	a No
2	Highest completed	1. None 2 Primary 3 Middle/ISS 4 SSS Kushlaration		AGE
3	Ethnic group	5. Tertiary/Polytechnic 1-Fonti 2-Efutu 3-Awutu 4-Other Alan S./Polytechnic	_	EDU
4	Religion	7. Other(specify:		ETHI
5	Occupation	A. Other (Straditionalist, A. Other (Straditionalist, A. Other (Straditionalist, Straditionalist, A. Other (Straditionalist, Straditionalist, A. Other (Straditionalist, Straditationalist, Straditionalist, Straditiona		RELI
6	Number of children	5.Student 6.Other(OCU
7	Area of residence			CHIL
Ļ	Marital status	1. Solipond, 2.Biriwa, 3.Anomabo 5. Mankessim		RESI
10	8 Marital status 1-Marited through church/mosque wedding, 2-Married only by traditional rites 3-Engaged, yet to be married 4-Co-habitation (living together) 5-Divorced/Separated/Widowed 6-Single 7-Other:		-	MAS

B. (URRENT PARTNER (Respon	sible for current pregnancy) INFORMATION		
1	Age of partner?		<u> </u>	PAGE
2	Highest completed educational level	1. Nane 2. Primary 3. Middle/JSS 4. SSS/SHS/Vocational 5. Tertiury/Polytechnic		PEDU
3	Ethnic group	1-Fanti 2-Efutu 3-Awutu, 4-Öther Akan 5-Ewe, 6-Hausa, 7. Other[specify:]		ретні
4	Religion	1. Christian, 2. Muslim, 3. Traditionalist, 4. Other ()		PRELI
5	Occupation	1. Fishmanger 2. Farmer 3.Petty trader 4.Crvit/Public servant, 5.Student 6.Other		POCU
6	Previously married before present relationship?	1. Yes, 2.No. 3 May be/Don'i know	_	PMAR
7	Partner has any children beside those with you?	1. Yes, 2.No, 3 May be /Don't know		PCHD

C. R	ELATIONSHIP ISSUES			YRM
1	Years of marriage or relationship with present partner		10	
2	Does your partner stay in the same house as you?	1. Yes, 2.No		uv
-	Does your bushand/oartner have another wife/spouse beside	L Vac 7 No		OWI
3	yoursell?	1. 183, 2.110		

Version: 16.10.2011

NATE OF INTERVIEW: 1 1 1 1 1 11

Page 1 of 5
Inter	nded contraceptive use among pregnant wom	80 Drassel		
		Region of Ghana	listrict of the	Central
D. 5	OCIO-ECONOMIC BACKGROUND	CODE OF RESPONDENT: I	11	
4	What are the two main sources	Piped water :		
	drinking water for members of your household?	1=Piped into house, 2= Piped into compound, 3= Piped into neighbours' compound, 4= Public Tep Water from open well: 5= Open public well Water from covered well or bore hole: 7= Protecled well in compound, 8= Protecled public well Surface water: 9= Spring, 10= River/Stream, 11= Pond/Lake, 12= Dam, Other 13= Rainwater, 14= Tanker truck, 15= Bottled water, 16= Sachels of water, 17= Other		WAT
	within your house or compound? (One response only)	(1=Flush toilet, 2= Pit latine /KVIP, 4= None/ bush/field/Seashore, 5= Other (specify:	-	TOL
3	Does your household own any of the following?	Canoe (1=Yes, 2=No)	_	CAN
1	Read responses	Fishing net (1=Yes, 2=No)	-	NET
		Cold store (1=Yes, 2=No)		COL
		Refrigerator (1=Yes, 2=No)	_	RFR
		Outboard motor (1=Yes, 2=No)		ουτ
		Radio (1=Yes, 2=No)		RAD
ļ		Television (1=Yes, 2=No)	-	TEV
		Mobile telephone (1=Yes, 2=No)		MOB
		Com/.cassava mil (1=Yes, 2=No)	-	MIL
		Bicycle (1=Yes, 2=No)	-	BIĆ
		Motor cycle (1=Yes, 2=No)		OMT
		Car, Truck or Fishing boal (1=Yes, 2=No)		CAR
4	What type (s) of fuel does your household use for cooking?	(1= Electricity, 2= Gas, 3= Kerosene, 4= Straw, 5= Charcoal, 6= Firewood, 7= Dung, 8= Other specify;		FUE1
5	What is the main material of the floor of your house? (One answer only)	(1=Earth/Sand, 2=Terrazzo, 3=Wood Planka, 4=Parquel or polished wood, 5=Ceramic blas, 6=Cernent, 7=Carpet, 8=Linoleum, 9=Other (specify:)		FLR

Version: 16.10.2011

DATE OF INTERVIEW: 1 1 1 1 1 1 1

Page 2 of 5

Intended contraceptive use among pregnant women presenting at health facilities in the Mfantseman district of the Central Region of Ghana FACIITY CODE: 1_1 INTERVIEWER INITIAL: 1_1_1 CODE OF RESPONDENT: 1_1_1_1

PRODUCTIVE HISTORY & CURRENT PREGNAM		
How many times have you been pregnant?		
Have any of your pregnancies ended in miscarciage or abortion?		PREG
	1. Yes 2.No 99.Not applicable	ABOT
2a. If YES, how	many times? 99 if not applicable	ABOY
How many children have you delivered?		CDEV
How old is your last child?		
Have any of your children praced away?	99 if not applicable	
take any of your condition passed away?	1. Yes 2.No 99.Not applicable	PASS
you expecting to get pregnant when you noticed you were pregnant?	1. Yes, 2.No	EXPT
How would describe the current pregnancy?	Wanted and of the right time, Wanted but not at the time it came Not wanted and unexpected	DESB
	PRODUCTIVE HISTORY & CURRENT PREGNAN How many times have you been pregnant? Have any of your pregnancies ended in miscarriage or abortion? 2a. If YES, how How many children have you delivered? How old is your last child? Have any of your children passed away? With regards to your current pregnant, we you expecting to get pregnant when you noticed you were pregnant? How would describe the current pregnancy?	PRODUCTIVE HISTORY & CURRENT PREGNANCY How many times have you been pregnant? Have any of your pregnancies ended in miscarriage or abortion? 1. Yes 2.No 99.Not applicable 2a. If YES, how many times? 99 if not applicable How many children have you delivered? How old is your last child? 99 if not applicable Have any of your children passed away? Have any of your children passed away? With regards to your current pregnant, we you expecting to get pregnant when you noticed you were pregnant? How would describe the current pregnancy? 2. Tranted bui not at the time it came 3. Not would and unrespected

F. KI	NOWLEDGE & USE OF VARIOUS FAMILY PLANNING METH	ODS		
		AWARENESS	EVER USE	
1	PILL (Women can take a pill every day to avoid becoming pregnant)	1. In: 2.80 II	1.76 2% II	PEL1 PEL2
2	IUD Women can have a loop or coil placed inside them by a doctor or nurse	1.Te 2.So II	1.7α 2% II	IUDI IUD2
3	INJECTABLES Women can have an injection by a health provider that stops them from becoming pregnant for one or more months).Ia 154 I]	1.Tet 2.No I1	ונאם 202
4	IMPLANTS Women can have several small rods placed in their upper arm by a doctor or nurse which can prevent pregnancy for one or more years.	1.1a 280 II	1. Te: 2.No II	IMIP1 IMIP2
5	CONDOM Men can put a rubber sheath on their penis before sexual intercourse.	1.3a :>> II	1.7e 1% II	CONI CON2
6	FEMALE CONDOM Women can place a sheath in their vagina	1. Tes 2.50 II	1.Te 2Ne II	CON1 CON2
7	DIAPHRAGM Women can place a thin flexible disk in their vagina	1.Tet 2.So II	1.7a 2.8e II	DLAI DLA2
8	FOAM OR JELLY Women can place a suppository, jely, cream	1.7m 2.80 II	1.Yes 2.Su II	FOM
9	In their vagina before sectal methods before sectal methods and the sectal methods and the sectal methods and the sectal methods are set and the month she is most likely to get	1.1a 250 II	1.Te 250 1]	RHY1 RHY2
	pregnant.	1.Yes 2.No 1_1	1.7m 2.80 1_1	WTTH: WTTH:
10	WITHDRAWAL Men can be careful and pull out baloe careful			ENEI
11	EMERGENCY CONTRACEPTION As an emergency measure after unprotected sexual intercourse, women can take special pils after unprotected sexual intercourse, women can take special pils	1.7es 2.8e II	i.Te 2No II	EME

Version: 16.10.2011

DATE OF INTERVIEW: 1 VI I VI I VI

Page 3 of 5

.

Intended contraceptive use among pregnant women presenting at health facilities in the Mfantseman district of the Central Region of Ghana

.

	FACIITY CODE: 1_1 INTERVIEWER INITIAL: 1	Ghan	aith facilities in (CODE of an	he Mfantseman district o	the Central	a
L. KI (Loct	NOWLEDGE OF LACTATIONAL AMENORRHOEA			SPONDENT: II		- 1
1	Can a woman use exclusive breastfeeding to prevent	delay .	return to fertility	ofter delivery)		
2	Can a woman get pregnant when her mensor have	_		-Tes 2No 3-Don 1 horse	E	EXC
3	Can a woman get programmer in the menacy have no			-Ses 2-No 3-Don'i know		- NES
4	Can a woman get program within six months of delive	ering?		-Tes 2-No 3-Don's know	DE	EV3
J. A	CCEPTABILITY OF POST DADTUNA CANNING	lvery		Tes 2-No 3-Don's boow	DE	EV4
1	Do you think it is accessible for	ING				
К. Р	pregnancy after delivery?	y plar	ining to avoid	1 8 <u>~ 1.No</u>	F	FPA
	CONTRACTOR CONTRACTOR		ling			
1	Which of these methods have you used to avoid gettin pregnant too soon after delivery?	ng	Not applic:	ible (no previous child)	N	VAP
	(Mark 🗸 for method u	sed)	Done noth	ing to avoid pregnancy		ют
	(Multiple answers allowed to indi- various approaches used in the		Avo	iding sex after delivery	A'	VD
		(ICII)	Havir	g partner use condom	P	PAT
			E	xclusive breastfeeding	E	EXB
2	In the past ware you show a successful in mounting			Rhythm method	81	YM
2	pregnancy soon after delivery?		1. Tes 2.No	3 - Not of no previous child	S	ON
L. P	OST-PARTUM FAMILY PLANNING INTENTIONS					
1	After this pregnancy, how long do you wish to walt before getting pregnant again?		1. One year, 2. 1 4 - Four yea 6. Other	wo years, 3-Three years irs S. No more children	w	/AG
2	How do you plan (main way) to avoid getting pregnant too soon after delivery? (Read out the options)		2. 3. Having set 4. Adopting a 5. Other, spo	1. Avoiding sax Insisting on condom use only during safe period family planning method cily:	A	aft
3	If you wished to do family planning after delivery.		1.Đ	clusive breastfeeding	8	BRS
-	which three methods would you prefer			2.Condom	M	100
	(Rank as 1, 2, & 3, where 1 is best choice and 3 is least			3.Rhythm	===- <u>M</u>	
	preferred)		4. P	all (morning-morning)		JEC
		5.	njectaoles (every	6. IUD		
		-		7. Implant (long term)	T	٢NA
			8. Ster	ilization (permanent)		STE
		<u> </u>	9.Eme	rgency contraception	EM	MG

Version: 16.10.2011

DATE OF INTERVIEW: 1 1 1 1 1 1 1

Page 4 of 5

Intended contraceptive use among pregnant women presenting at health facilities in the Mfantseman district of the Central Region of Ghana

.

FACIITY CODE: 1_1 INTERVIEWER INITIAL: 1_1_1 CODE OF RESPONDENT: 1_1_1

M, I	POST-DELIVERY FOLLOW-UP			
1	We wish to be able to follow-up with you to kn planning intentions were actualized. Would yo	ow if your post-delivery family ou agree that we contact you?	1. Ja 2.No	WAG
	1a. If Yes, to Q1, can you describe where we	can locate you?		
	Telephone contact:	[TELC]		
	DETAILED DESCRIPTION	÷		

Intended contraceptive use among pregnant women presenting at health facilities in the Mrantseman district of the Central Region of Ghana FACITY CODE: 1_1 INTERVIEWER INITIAL: 1_1 CODE OF RESPONDENT: 1_1_1

		N. DETAILS			
1	Date of Birth (dd/mm/yyyy)	//			DOB
2	Religious denomination	05. Pentro 0 08. Tradi 10. Other:	01. Catholic 02 Anglican 03. Methodist 04. Presbyterian ostol/Charismatic 6. Other Christian 07. Moslem itional/Spiritualist 0.9. No religion	u)	RED
3	In your entire life, how many children (alive or dead) have you given (Put "00" if none a	birth to? and skip to ().5)	1_1_1	мсн
	d£	How many a	re males (sons)	II	SEA
	<u>عد</u>	How many are fema	les (daughters)		GUD
4	Have you ever given birth to a child wi died?	ho was born alive but later (11 No, skip to 5)	L Yes 2.No		ALI
	40	How many we	re males (sons)	II	SON
	46	How many were fema	des (daughters)	1_1	DAU
5	Name of the suburb where you live				SUB
6	Place of residence of partner (spouse)	1-Not applicable. We live toget 2-Doesnat live in same house 3-Lives outside of town bu 4-Lives outside of central region 5-Live	her in same house, e but in same town of in Central Region but with in Ghana s outside of Ghana	J	MAS
7	How many other wives/spouses does your husband/partner have?		Put '00' il none		ENO
8	In order of seniority, what is your rank among your partner's wives	tst-Not applicable. As I said 2 rd wife, 3 rd wife, 4 rd w	am the only wife		SEN

am / pm TIME: à DATE OF INTERVIEW: Suppi, 1

.

Version: 16.10.2011

l

269

3: Phase 2 Questionnaire

QUESTIONNAIRE PHASE 2

Intended contraceptive use among pregnant women presenting at health facilities in the Mfantseman district of the Central Region of Ghana

FACILITY CODE: I_I INTERVIEWER INITIAL: I_I_I CODE OF RESPONDENT: I_I_I_I

INTRODUCTION

The Centre for Health Research and Implementation Support (CHRIS-Ghana) in conjunction with the Municipal Health Directorate and Saltpond Government Hospital are conducting a study in the intention of pregnant women to use contraceptive methods after delivery in the Mfantseman municipality. The objective is to assess the intentions of women who deliver to adopt modern family planning methods. The findings of the study will help the Ghana Health Service and other health agencies to design post-delivery family planning services in the Mfantseman Municipality and other parts of Ghana. The study is not a part of routine health services.

You were interviewed a few months ago when you were pregnant. We are inviting you to participate in this follow up study because you agreed to do so during the first study.

PROCEDURES

A trained research staff will use a questionnaire to ask questions related your use of various family planning methods after delivery. He/she will ask you questions related to your behaviours after delivery related to family planning use, breastfeeding and sexual abstinence. You will also be asked questions on health related issues that are known to influence the use of family planning by women who have just delivered.

As you respond to the questions that staff will ask, he/she will fill the questionnaire. The interview will take about 20mins to complete.

As part of the study, you may be invited to participate in a group discussion with other women. Such discussion will be audio taped for the sole purpose of enabling transcription. They will be destroyed as soon as transcription is completed.

RISK AND BENEFITS

By taking time off your schedule to answer the questions, we will be causing you inconvenience; we will however minimize this by going strictly according to the questionnaire guide and completing on time to enable you go about your duties.

Some of the questions may require information that may be quite personal. Should you feel uncomfortable with such questions, you may opt not to answer them.

At the end of the questions, you will be offered the opportunity to ask questions that you may have on family planning. The staff will answer as much as he/she is trained to do so. No money or other reward will be given to you as result of your participation in the study.

CONFIDENTIALITY

All the information that you provide to complete the questionnaire will be considered strictly confidential. It will be available to only persons directly connected with the study. The forms on which the information will be collected will be kept under lock and key/ no information (e.g. name or hospital ID) that identifies and matches you with the responses on the form will be collected.

VOLUNTARINESS AND WITHDRAWAL

Your participation in the study is completely voluntary and you reserve the right not to participate, or even when you have started, to stop answering the question. This is your right and the decision you take will not be disclosed to anyone. It will also not affect the care that will be offered you at this facility now or future.

WHO TO CONTACT

This study has been approved by the Ethics committee of Ghana Health Services. If you wish to ask questions, or need further explanations later, you may contact Dr. Frank Baiden (0244591181) of the Ghana Health Sevice, Dr. Sebastian Eliason (0200975567) of the School of Medical Sciences in Cape Coast, Dr. Derek Bonsu of the Saltpond Government Hospital or Madam Yvonne Graham-Hayfron of the Municipal Health Directorate. You may also contact the administrator of the ethical review committee of the Ghana Health Service on Tel 0302681109

Part 2: CONSENT DECLARATION

"I have read the information given above, or the information above has been read and explained to me in a language I understand. I have been given a chance to ask questions concerning this study and the questions have been answered to my satisfaction. I now voluntarily agree to participate in this study knowing that I have the right to withdraw at any time without affecting future health care services" Name of respondent/participant: -----

Participant's signature		Participant's left thumb
print		
	OR	
Date:/		
Name of witness		SignatureDate:
Name of Investigator	{	SignatureDate:

Respondents unable to read or write need witnesses. Respondents who are able should sign and date personally.

RESPO	UNDENT INFORM	IATION CONFIRMATION		
ANC	REGISTRATION	NO:		
PLEA	SE CHECK: Has t	his women been interview to interview		
alread	y?			Yes
If YES	5, STOP INTERVIE	EW HERE	🗆 No	
1	Age Date of Birth		I I	AG E
2	Highest completed	 None 2.Primary 3.Middle/JSS 4.SSS/SHS/Vocational 	I 	ED
	educational level	5. Tertiary	-	
3	Ethnic group	 1-Fanti 2-Efutu 3-Awutu, 4-Other Akan 5-Ewe, 6-Hausa, 7. Other(specify:) 		ETH
4	Religion	1. Christian, 2.Muslim, 3.Traditionalist,	-	REL I
5	Occupation	1. Fishmonger 2. Farmer 3.Petty trader 4.Civil/Public servant, 5.Student		oc U
	Number Of		I_	СН
6	children even	Males [] Females []	I	L

	born/Parity of				
	the last birth		I	Ţ	
7	Area of	1. Saltpond, 2 Biring 2			
	residence	4. Mankessim	-	RES	-
8	Preferred No. of		-	I	
	children	Males [] females[]		PNO	
9	Marital status	Single (Never married)			
		2-Married,			
		3- Co-habitation (living together) 4 -Divorced/Separated	_	MA	
		5. Widowed	-	S	
)			

PC	OSTPARTUM OUTCO	MES		
	What was the	1. Live term baby, 2.live pre-term baby	I	OU
1	outcome of your last	3. Stillbirth	I	TC
	pregnancy?	4.miscarriage	I	10
	In what facility did	Home 2.public hospital/clinic 3.		FA
2	you deliver?	Private hospital/clinic 4. Maternity		с
		home		

		5. Other		 _
	Sex of Child if			
3	outcome was a live			SE
	birth	1. Male, 2. Female, 3. Ambiguous	-	xc
	What was mode of			M
4	delivery?	SVD, 2. Assisted vaginal delivery, 3.		OD
		Caesarean section		Е
5	Any delivery			CO
	complications?	l, Yes, 2. No]	MP
	Date of delivery		I	LO
6				NG
			ļ	
	If outcome was live,			СН
7	is child still alive?		—	LI
		1. Yes, 2.No,	-	V
	If death, date of			DE
8	death			AT

C. 1	POSTPARTUM BEHAVIOURS	
1	Has your period 1. Yes 2.NO returned since your	AMEN
2	last birth? For how many I_I_I	AMED

	months after birth			
	did you not have			
	your period			
3	Did you ever			
	breastfeed after			BFED
	delivery	1. Yes 2.NO		
4	If yes, for how many			
	months did you			BFLG
	breastfeed		ii_	
5	Have you resumed			1
	sexual relationship			SEXR
	since your last birth?	1. Yes 2.NO		
6	If yes, when did you			SXLG
	start?		1_1_1	
6	Who initiated the	•		SEXI
a	first sex?	Partner, Myself		
7	Have you sort		1	
	information about			INFO
	contraception since			
	delivery?	1. Yes 2.NO		
8	If yes, from where?	Heath worker/service 2.		101000
		Media –TV, radio,		INFOS
	0 4	newspapers 3. Friends		

9	Have you had any		
a	discussion about		
	family planning with		DISC
	your partner since		DISC
	delivery	1. Yes 2.NO	
9	If yes, what was		
Ъ	discussed?		DISC1
9	Who initiated the		
c	discussion?		DISC2
		Partner, Myself	
9	What was the		
d	outcome of the		DISC3
	discussion/		

PO	POST PARTUM CONTRACEPTIVE USE STATUS							
	1. Yes 2.NO	I_						
1	Have you adopted a	-I ADOPT						
a	contraceptive	- T						
	method?							
		-						
1	If yes, how many	_I LADPT						
b	months after your	-						

	last birth did you			
	adopt contraceptive		_I	
	method			
	If yes, Was there any			
2	discussion with your			
	partner before			DISCP
	decision was taken?	1. Yes 2.NO		
	Which of the	1 nartner 2 mother in law		
3	following influenced	3 church		DECI
	your decision?	4. self approval, 5. other(
	If you have not	I got pregnant immediately		
	adopted any method,	Partner was against the idea		
	Why have you not	I want to get pregnant as soon as		
	done so?	possible		
My ch		My child did not survive		PREG
4		I am not interested in FP		
		It is too early to decide after		
		delivery	-	
		Other (Specify)	-	
			-	
\vdash	If you have not			DOPT
4	adopted any method,	1. Yes 2.NO		

	do you plan to adopt			
	one soon or in the		_	
	future?			
	If you have adopted	1. Pills, 2.injectable, 3. Implants, 4.		
	a method which of	IUD, 5. Diaphragm, 6. Female		
	the following	Condom, 7. Male Condom, 8.		
6	methods did you	Female sterilization, 9. Male		CMET
	adopt	sterilization, 10. LAM, 11	_	
		Periodic abstinence, 12.	_	
		Withdrawal, 13. Other traditional	-	
	Why did you adopt	it is suitable for me 2. It has fewer		
7	that method?	sideffects 3. My partner endorsed it		VAD
ĺ		4. My provider chose it for me 5.		
		Other(specify)		
	Have you spoken			
8	about the benefits of			ADVOC
0	family planning to			
	anybody?	1. Yes 2.NO		

9. Please give your views on family planning after delivery.

.

4:Published Articles

.

.



RESEARCH

Open=Access

Factors influencing the intention of women in rural Ghana to adopt postpartum family planning

Sebastian Eliason¹, Frank Baiden^{2*}, Gloria Quansah-Asare³, Yvonne Graham-Hayfron⁴, Derek Bonsu⁵, lames Phillips⁶ and Kofi Awusabo-Asare⁷

Abstract

Background: Uptake of postpartum family planning (PPFP) remains low in sub-Saharan Africa and very little is known about how pregnant women arrive at their decisions to adopt PPFP. This information is needed to guide the development of interventions to promote PPFP.

Methods: We conducted a survey among pregnant women attending antenatal clinics in a rural district in Ghana. We used univariate and multivariate logistic regression analysis to explore how knowledge of various family planning (FP) methods, past experience with their use and the acceptability of PPFP to male partners and close relations influenced the intention of pregnant women to adopt PPFP.

Results: We interviewed 1914 pregnant women in four health facilities. About 84% considered PPFP acceptable, and 70% intended to adopt a method. The most preferred methods were injectables (31.5%), exclusive breastfeeding (16.7%), and oral contraceptive pills (14.8%). Women whose first choice of PPFP method were injectables were more likely to be women who had had past experience with its use (O.R = 2.07, 95% C.I. 1.50-2.87). Acceptability of PPFP by the pregnant woman (O.R. = 3.21, 1.64-6.26), perception of partner acceptability (O.R. = 3.20, 1.94-5.48), having had prior experience with the use of injectables (O.R. = 3.72, 2.61-5.30) were the strongest predictors of the intention to adopt PPFP. Conversely women who knew about the diaphragm (O.R. = 0.59, 0.38-0.93) and those who had past experience with IUD use (O.R. = 0.13, 0.05-0.38) were less likely to want to adopt PPFP.

Conclusions: Acceptability of PPFP to the pregnant woman, male partner approval, and past experience with the use of injectables are important factors in the PPFP decisions of women in this population. Antenatal and early postnatal care need to be adapted to take these factors into consideration.

Keywords: Postpartum, Family planning, Contraception, Male, Ghana, Sub-Saharan Africa

Background

The United Nation's Millennium Development Goal (MDG) 5 aims at reducing maternal mortality by three quarters, between 1990 and 2015. An important intervention towards achieving this target is the promotion of modern family planning (FP) among women in sub-Saharan Africa (SSA) [1,2]. Uptake of modern FP methods remains low in SSA and this is associated with a high incidence of unwanted pregnancies, unsafe abortions, unplanned deliveries and maternal mortalities [1,3].

The periods of pregnancy and immediately after delivery are considered opportune for counseling women on

Centre for Health Research & Implementation Support, Accra, Ghana Full list of author information Is available at the end of the article

the adoption of modern FP methods. This is because this period is often associated with a woman's frequent encounter with the health system [4]. These encounters provide avenue to promote optimal spacing of births through postpartum family planning (PPFP) [5]. It has been estimated that PPFP can prevent about 30% and 10% of maternal and child mortalities, respectively [6]. Data from Demographic and Health Surveys (DHS) in 27 countries suggest that less than 35% of women who wish to avoid pregnancy during the postpartum period use any form of modern contraception [7,8]. Very little is known about how pregnant women in SSA arrive at their PPFP decisions. This information is nevertheless critical to the design of strategies to increase the uptake of PPFP [4,9].



© 2013 Eliason et al.; licensee BioMed Central Ltd. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/2.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Correspondence: baidenf@yahoo.co.uk

Ghana

In Ghana, unintended childbearing is estimated at about 0.7 births per woman [10]. The major goals of the 1994 Ghana National Population Policy are to reduce the total fertility rate from 5.0 to 3.0, and to increase the contraceptive prevalence rate from 15 to 50 percent between year 2000 and year 2020 respectively [11,12]. According to the 2008 Ghana Demographic and Health Survey [13], fertility rate, contraceptive prevalence rate (CPR) and unmet need in Ghana are 4.0, 17% and 35% respectively, with considerable rural-urban disparities. Women in rural areas have an average of two children more than those in urban areas (4.9 versus 3.1) and use modern FP less than their urban counterparts (15% versus 19%). The level of unmet need for FP methods shows similar trends (rural 38% and urban 32%). The 2011 Multiple Indicator Cluster Survey suggests improvement in the CPR from 17% to 23% and reduction in the level of unmet need from 35% to 26.4%. At current rates, however, the targets set in Ghana's population policy are unlikely to be met by 2020. The postpartum period remain neglected in FP research in Ghana, and very few studies have focused on the FP needs of women during this period of "window of opportunity" [4,9].

We conducted a survey among pregnant women attending antenatal clinics in rural health facilities in the Central Region of Ghana to identify the factors that influence their intention to adopt PPFP. We report here on how the intention to adopt PPFP is influenced by knowledge of various FP methods, past use of these methods, and the perceived acceptability of PPFP to male partners and close relations.

Methods

Study sites

The study was conducted in four health facilities in the Mfantseman District. This district is about 110 km west of Accra and lies along the Atlantic coastline. The district extends 13 kilometers inland, and is essentially rural, except for two towns, Saltpond, and Mankessim, which are semi-rural. The district was selected for this study because of recent reports of increasing incidence of unsafe abortions, high teen pregnancies (13.6% of all pregnancies in 2010) and low uptake of modern FP [14]. It was also based on the familiarity of the research team with the district.

All pregnant women attending antenatal clinic at the Saltpond Government Hospital (semi-rural), Mankessim Health Centre (both semi-rural) and the Biriwa and Anomabo Health Centers (both rural) between January 2012 and April 2012 were targeted and interviewed as they arrived at the facility on routine antenatal visits. The health facilities were purposefully-selected in order to provide a mix of semi-rural and rural settings.

We used a five-page questionnaire that explored questions related to the intention to adopt PPFP and the circumstances under which the current pregnancy occurred. Women who indicated that the pregnancy was unexpected, unwanted or both were considered to have had an unplanned pregnancy. Other areas covered in the questionnaire included knowledge and use of various FP methods, their acceptability of PPFP, as well as acceptability to male partners and other close relations (mother, mother-in-law, father-in-law and religious leader). The questionnaire consisted almost entirely of closed-ended questions and was pre-tested in health facilities in the district that were not selected to participate in the study. They were administered by trained research assistants who were familiar with the culture and traditions of the women attending the clinics. The minimum qualification of interviewers was polytechnic or university diploma. The questionnaires were administered in the language that respondents were comfortable to speak in.

Ethical consideration

Ethical and administrative approvals were obtained from the Ethics Review Committee of the Ghana Health Service (GHS) and the Municipal Health Directorate (MHD). Written informed consent was obtained from each participant.

The data were double-entered using EPI-DATA. The inputs were verified and cleaned to achieve a clean data set. This set was exported into STATA (version 11) for analysis. Descriptive and univariate analyses using logistic regression were performed to explore the influence of knowledge and past use of FP on the intention to adopt PPFP. The effects of perceived PPFP acceptability to male partners and close relations were similarly explored. Factors found in univariate analysis to be significantly associated (P-value <0.05) with the main outcome of interest (intention to adopt PPFP) were included in a multivariate model. Pregnant women were considered to have the intention to adopt PPFP if they indicated they will adopt a modern FP method to delay pregnancy after they deliver.

Sample size

A total of 1900 pregnant women were targeted to be interviewed on the assumption that 50% of them will have the intention to adopt PPFP. With 80% power, it was possible to estimate the proportion of women willing to adopt PPFP within a margin of error of 3%. The district recorded about 4000 deliveries in 2009.

Results

Background of respondents

A total of 1914 pregnant women with an average age of 25.6 years (standard deviation 6.5) were interviewed.

The majority were Christians who belonged to the Fanti ethnic group. About a fifth had had no formal education and the dominant occupations were petty trading and fish mongering (Table 1).

About a third (35.2%) of pregnant women did not have About a third (35.2%) of pregnant women did not have a child. Of those who had at least one child, the average number of children was 2 (standard deviation-2). The majority of women (70%) indicated that the pregnancy they were carrying was unexpected or unwanted or both at the time it occurred. About 70% of pregnant women expressed the intention to adopt PPFP. For women who preferred to have more children, the average desired time before a next pregnancy was 4.6 years (standard deviation 1.5 years). The most preferred PPFP methods were injectables (31.5%), exclusive breastfeeding (16.7%), and oral contraceptive pills (14.8%). Women whose first choice of PPFP method were injectables were more likely to be women who had had past experience with its use (O.R = 2.07, 95% C.I. 1.50-2.87). Past use was however less predictive in the case of the pill (1.19, 0.81-1.75). It actually appeared to be inhibitory in the case of exclusive breastfeeding (0.81, 0.42-1.57).

Effect of knowledge of various FP methods

The two most widely known FP methods were the male condom (86.4%) and injectables (80.5%) while the least known were the foaming tablets (20.7%) and the diaphragm (24.4%). Respondents who had heard about the pill (1.76, 1.40-2.20), injectables (2.20, 1.73-2.79), implants (1.60, 1.30-1.95), male (1.97, 1.50-2.59) and female

Varia	Variable		PFP method
		Yes (%)*	No
Age	≥20 years	1136 (72.6)	429
	<20 years	190 (56.7)	145
Respondent-partner age difference	Partner more than 5 years older	440 (68.9)	199
	Partner 1–5 years older	798 (69. 7)	347
	Partner younger than respondent	72 (77.4)	21
Highest completed educational level	Senior high school and above	127 (56.0)	100
Inglicat comprome care	Junior high school	595 (70.9)	244
	Primary	317 (74.4)	109
	None	286 (70.1)	122
Febrie evenue	Fanti	1217 (70.4)	513
Etinic group	Other	108 (63.5)	62
0 H J	Christian	1242 (70.1)	83
Religion	Other	529 (29.9)	46
	Peotecostal and others	649 (73.3)	236
Christian denomination	Protestant	324 (67.9)	153
	Catholic	131 (65.4)	68
	the manager and netry trader	1087 (73.2)	399
Occupation	Fishmunger und petry were	71 (57.7)	52
	Government and other once souther	46 (54.8)	38
	Other	68 (65.4)	36
	>Three	362 (77.9)	103
Number of children	Two	237 (76. 0)	75
	One	338 (73.8)	120
	None	389 (58.3)	278
	Pural	652 (71.6)	259
Area of residence	Comi Pural	673 (68.1)	316
	Serni-Ruiwi	790 (72.5)	299
Marital status	Mattea	380 (69.5)	167
	Engaged or cohabiling	152 (58.7)	107
	Single, divorced, separated or widowed		

Table 1 Sociodemographic background of respondents

Row percentages.

condom (1.62, 1.29-2.03) were more likely to want to adopt PPFP. Participants who were aware that exclusive breastfeeding (1.83, 1.34-2.50) could be a method of contraception were also more likely to want to adopt ppFP. On the hand, those who had heard of the diaphragm (0.74, 0.59-0.92) and foaming tablets (0.77, 0.61-(1.98) were unlikely to have the intention of adopting pppp. In multivariate analysis, knowledge of the use of implants (O.R. = 1.64, 0.11-2.41) and exclusive breastfeeding (O.R. = 1.56, 1.12-2.18) were associated with a intention to adopt PPFP while knowledge of the diaphragm (O.R. = 0.59, 0.38-0.93) was associated with the lack of intention to adopt PPFP (Table 2).

Past use of various methods

and a fifte of a film of the

The most widely used FP methods were the withdrawal (29.5%) and male condom (26.9%) while the least used were the diaphragm (1.0%) and foaming tablets (1.2%). Respondents who had used the pill (O.R = 2.53, 95% C.I. 1.83-3.49), injectables (3.83, 2.70-5.43) and emergency contraceptive pills (1.52, 1.05-2.20) in the past were

. .

more likely to want to adopt PPFP. On the hand, those with prior experience with the use of the IUD (0.39, 0.16-0.96) were unlikely to want to adopt PPFP. In multivariate analysis, women who had used injectables (O.R. = 3.72, 2.61-5.30) and pill (O.R. = 2.22, 1.59-3.11) were more likely to have the intention to adopt PPFP while those who had prior experience with IUD use (O.R. = 0.13, 0.05-0.38) were significantly less likely to have the intention of adopting PPFP (Table 3).

Acceptability of PPFP to partners and close relations

The majority (84%) of pregnant women considered it acceptable for women (self-approval) to use modern FP methods in PPFP and 70.0% had the intention to adopt a method. Although self-approval was strongly associated (8.55, 6.29-11.63) with the intention to adopt a method, 64% of women who considered the PPFP acceptable did not have the intention to adopt it. Further analysis show the more educated a pregnant woman was, the more likely she was to consider PPFP acceptability and yet not intend to adopt it (P-value for trend <0.01). Pregnant

lable 2 Effect of knowledge of family planning on intention to adopt P	PFI
--	-----

Method		Intention PPFP n	to adopt	Univariate	Multivariate
		Yes	No	O.R. (95% C.I.)†	O.R. (95% C.I.)†
Oral contraceptive pills	Yes	1069	405	1.76 (1.40-2.20)	1.16 (0.70-1.91)
	No	257	171	1.00	1.00
Intrauterine contraceptive device	Yes	490	232	0.87 (0.71-1.06)	
	No	836	344	1.00	N/A
Injectables	Yes	1124	413	2.20 (1.73-2.79)	1.37 (0.79-2.41)
	No	202	163	1.00	1.00
Implants	Yes	881	319	1.60 (1.30-1.95)	1.64 (1.11-2.42)
	No	445	257	1.00	1.00
Male condom	Yes	1184	466	1.97 (1.50-2.59)	1.18 (0.58-2.41)
	No	142	110	1.00	1.00
Female condom	Yes	1068	414	1.62 (1.29-2.03)	1.03 (0.60-1.75)
	No	258	162	1.00	1.00
Dianhraom	Yes	300	164	0.74 (0.59-0.92)	0.59 (0.38-0.93)
	No	1025	412	1.00	1.00
Foaming tablets	Yes	258	137	0.77 (0.61-0.98)	0.84 (0.53-1.32)
	No	1067	258	1.00	1.00
Rhythm (calendar) method	Yes	757	356	0.82 (0.67-1.01)	N/A
(calendar) method	No	568	220	1.00	
Withdrawalayati	Vac	854	345	1.21 (0.99-1.48)	N/A
minurawal method	No	471	231	1.00	
Emergen	Yor	630	262	1.09 (0.89-1.32)	N/A
energency contraception	Tes No.	695 314	1.00		
Eval.	NO	567	205	1.83 (1.34-2.50)	1.56 (1.12-2.18)
caciusive breastfeeding	Yes	136	90	1.00	1.00

condom (1.62, 1.29-2.03) were more likely to want to adopt ppFP. Participants who were aware that exclusive breastfeeding (1.83, 1.34-2.50) could be a method of contraception were also more likely to want to adopt ppFP. On the hand, those who had heard of the diaphragm (0.74, 0.59-0.92) and foaming tablets (0.77, 0.61-0.98) were unlikely to have the intention of adopting ppFP. In multivariate analysis, knowledge of the use of implants (O.R. = 1.64, 0.11-2.41) and exclusive breastfeeding (O.R. = 1.56, 1.12-2.18) were associated with a intention to adopt PPFP while knowledge of the diaphragm (O.R. = 0.59, 0.38-0.93) was associated with the lack of intention to adopt PPFP (Table 2).

Past use of various methods

. ...

The most widely used FP methods were the withdrawal (29.5%) and male condom (26.9%) while the least used were the diaphragm (1.0%) and foaming tablets (1.2%). Respondents who had used the pill (O.R = 2.53, 95% C.I. 1.83-3.49), injectables (3.83, 2.70-5.43) and emergency contraceptive pills (1.52, 1.05-2.20) in the past were

more likely to want to adopt PPFP. On the hand, those with prior experience with the use of the IUD (0.39, 0.16-0.96) were unlikely to want to adopt PPFP. In multivariate analysis, women who had used injectables (O.R. = 3.72, 2.61-5.30) and pill (O.R. = 2.22, 1.59-3.11) were more likely to have the intention to adopt PPFP while those who had prior experience with IUD use (O.R. = 0.13, 0.05-0.38) were significantly less likely to have the intention of adopting PPFP (Table 3).

Acceptability of PPFP to partners and close relations

The majority (84%) of pregnant women considered it acceptable for women (self-approval) to use modern FP methods in PPFP and 70.0% had the intention to adopt a method. Although self-approval was strongly associated (8.55, 6.29-11.63) with the intention to adopt a method, 64% of women who considered the PPFP acceptable did not have the intention to adopt it. Further analysis show the more educated a pregnant woman was, the more likely she was to consider PPFP acceptability and yet not intend to adopt it (P-value for trend <0.01). Pregnant

Table 2	Effect o	f knowledge of	f family	planning or	n intention t	o adopt	PPFF
						o ddopt	

Method		Intention PPFP n	to adopt nethod	Univariate	Multivariate
		Yes	No	O.R. (95% C.I.)†	O.R. (95% Cl.)†
Oral contraceptive pills	Yes	1069	405	1.76 (1.40-2.20)	1.16 (0.70-1.91)
	No	257	171	1.00	1.00
Intrauterine contraceptive device	Yes	490	232	0.87 (0.71-1.06)	
	No	836	344	1.00	N/A
Injectables	Yes	1124	413	2.20 (1.73-2.79)	1.37 (0.79-2.41)
	No	202	163	1.00	1.00
Implants	Yes	881	319	1.60 (1.30-1.95)	1.64 (1.11-2.42)
	No	445	257	1.00	1.00
Male condom	Yes	1184	466	1.97 (1.50-2.59)	1.18 (0.58-2.41)
	No	142	110	1.00	1.00
Female condom	Yes	1068	414	1.62 (1.29-2.03)	1.03 (0.60-1.75)
	No	258	162	1.00	1.00
Diaphragm	Yes	300	164	0.74 (0.59-0.92)	0.59 (0.38-0.93)
	No	1025	412	1.00	1.00
Foaming tablets	Yes	258	137	0.77 (0.61-0.98)	0.84 (0.53-1.32)
	No	1067	258	1.00	1.00
Rhythm (calendar) method	Yes	757	356	0.82 (0.67-1.01)	N/A
(calendar) method	No	568	220	1.00	
Withdrawal mash- d	Yes	854	345	1.21 (0.99-1.48)	N/A
	No	471	231	1.00	
Fmore	Vor	630	262	1.09 (0.89-1.32)	N/A
emergency contraception	les	695	314	1.00	
.	NO	567	205	1.83 (1.34-2.50)	1.56 (1.12-2.18)
exclusive breastfeeding	Yes	136	90	1.00	1.00

Table 3 Effect previous use of family planning on intention to adopt PPFP

O.R. (95% CI.)† 2.53 (1.83-3.49)	O.R. (95% C.I.)†
2.53 (1.83-3.49)	
2.53 (1.83-3.49)	
	2.22 (1.59-3,11)
1.0	1.00
0.39 (0.16-0.96)	0.13 (0.05-0.38)
1.0	1.00
3.83 (2.70-5.43)	3.72 (2.61-5_30)
1.0	1.00
0.82 (0.38-1.78)	
1.0	NA
1.00 (0.81-1.25)	
1.00	NA
0.73 (0.37-1.46)	
1.00	NA
0.74 (0.29-1.71)	
1.00	NA
0.70 (0.29-1.71)	
1.00	NA
0.87 (0.70-1.10)	
1.0	N/A
1.20 (0.97-1.50)	
10	N/A
1.52 (1.05-2.20)	1 26 (0 86-1 85)
10	1.00
1 54 (0 27-1 08)	1.00
1.00	NA
	$\begin{array}{c} 2.33 (1.83 - 3.49) \\ 1.0 \\ 0.39 (0.16 - 0.96) \\ 1.0 \\ 3.83 (2.70 - 5.43) \\ 1.0 \\ 0.82 (0.38 - 1.78) \\ 1.0 \\ 0.82 (0.38 - 1.78) \\ 1.0 \\ 1.00 \\ 0.82 (0.38 - 1.78) \\ 1.0 \\ 1.00 \\ 0.73 (0.37 - 1.46) \\ 1.00 \\ 0.73 (0.37 - 1.46) \\ 1.00 \\ 0.74 (0.29 - 1.71) \\ 1.00 \\ 0.70 (0.29 - 1.71) \\ 1.00 \\ 0.70 (0.29 - 1.71) \\ 1.00 \\ 0.87 (0.70 - 1.10) \\ 1.0 \\ 1.20 (0.97 - 1.50) \\ 1.0 \\ 1.52 (1.05 - 2.20) \\ 1.0 \\ 1.54 (0.27 - 1.08) \\ 1.00 \\ \end{array}$

women with higher education (beyond secondary school) were nearly four times (3.87, 2.57- 5.84) as likely to make this consideration as women who had had no formal education. This seemingly discordant consideration was neither associated with the age of a pregnant woman nor with the number of children she had.

The majority (76.2%) of pregnant women perceived that their partners will consider their adoption of PPFP acceptable. A higher proportion (82.0%) indicated that they will require the permission of their partners before they actually adopted a method. Among women who perceived that PPFP will be acceptable to the partners, 82.3% thought they will still need their permission before they could adopt a method. Also most (90.5%) women who indicated they personally approved of PPFP still considered that they will need the approval of their partners before they could adopt it. In univariate analysis, the intention to adopt PPFP was strongly associated with the perceived acceptability of PPFP by partners and close relations (P-value < 0.01 in all cases). After adjustments in multivariate analysis however, only self-approval of PPFP (3.21, 1.64-6.26) and perceived acceptability by male partner (3.2, 1.94-5.48) were significantly associated with the intention to adopt PPFP (Table 4).

Discussion

Although fewer than 30% of women in this study had ever used a modern FP method, 84% considered PPFP acceptable and 70% expressed the intention to adopt a method. This seemingly favorable basis for the promotion of PFPP needs however to be appreciated with caution. This is because of the existence of the welldocumented wide gap between FP intentions and actual uptake among women in SSA [8,15]. In this study, findings have been made that suggests that what is likely to happen among the cohort of women interviewed would not be substantially different from what has been found among non-pregnant women interviewed in the GHDS and others studies in SSA; namely a low level of uptake despite a high level of expressed intention to use. The wide disparity between pregnant women's level of knowledge of various modern FP methods, and their use of

and internion to adopt PPFP						
			Intention to ado	pt PPFP method	Univariate	Madelunation
stability of PPFP to respondent and relations	Pregnant		Yes	No	Q.R. (95% C L)+	
Acception	regnant woman	Yes	1244	372	8.55 (6 29-11 63)	3.21 (1.64.6.26)
	5	No	79	202	100	5.21 (1.04-0.20)
	Partner	Yes	950	193	778 (5.91, 10, 40)	22404540
	Mother	No	138	218	1.00	3.2 (1.94-5.48)
		Yes	625	175	1.00	
	Mother-in-law	No	192	175	3.30 (2.50-4.33)	1.60 (0.78-3.27)
		Yes	423	110	1.00	
	Father-in-law	No	168	170	3.69 (2.64-5.33)	3.06 (0.71-13.27)
		Yes	377	99	1.00	0.55 (0.10.0.55)
		No	151	166	1.00	0.50 (0.12-2.05)
	Religious leader	Yes	473	154	1.00	0.00 (0.43.1.07)
		No	198	177	2.75 (2.07-3.03)	0.90 (0.45-1.67)
Religious partner's permission before ado	pting PPFP	Yes	1136	187	2.23 (1.75-2.85)	0.94 (0.54-1.61)
		No	419	154	1.00	0.54 (0.54 1.01)
Willing to use postpartum without making p	oartner aware	Yes	614	127	3.04 (2,41-3.83)	1.37 (0.90-2.11)
		No	709	446	1.00	

Table 4 Effect of acceptability of PPFP and related factors on intention to adopt ppro

these methods in the past (Tables 2 & 3) is evidence in this regard. Furthermore the fact that even among the 85.2% of women who considered PPFP to be acceptable practice, 64% did not have the intention to adopt a method is pointer to the fact that the declared intention to adopt PPFP is predicated on a number of conflicting factors. An important limitation of this study therefore is the lack of follow-up of the pregnant women to establish how their PPFP intentions were met, and to compare the characteristics of those who take up PPFP methods and those who do not.

Personal conviction versus role of male partners

The study identified personal conviction as important in getting pregnant women to have the intention to adopt PPFP. This finding makes a case for continuing the education of pregnant women on the benefits of PPFP, and the advantages of spacing deliveries. Further evidence from the study however suggests that personal conviction is insufficient to ensure actual uptake of PPFP by the interviewed women. This is because, by even greater measure, the women in this study made the point about their need for partner approval before they could adopt a method of PPFP. Similar studies in SSA have pointed to the critical role played by male partners in the FP decision-making processes of women [16-18]. The finding in this study is further indication of how deep-seated the influence of male partners is on the decision of women in SSA to adopt FP. The evidence suggests that the personal conviction of women is likely to be superseded by the influence of partner approval.

What is however an irony is that in various surveys among men in SSA, it has been shown that except in except in polygamous societies, male partner acceptability of FP is not particularly different from that of women [19]. It has been shown that men are as desirous as their female partners to limit the number of children they father. It would appear therefore that women have an inaccurate perception of their male partners about FP. This has been hypothesized to be the result of inadequate spousal communication about FP [20]. Although the findings of the analysis of DHS data in Ghana and Chad [15,21] cast doubt on the likely effectiveness of spousal communication in causing an increase in the uptake of contraceptive methods, other studies [18], including recent intervention studies have provided convincing evidence about the effectiveness of using spousal communication to cause an increase in the uptake of FP. A trial in Malawi demonstrated that spousal communication on FP led to men actually facilitating contraceptive use by their partner [22]. Spousal communication is now considered to be an integral component of successful interventions to increase male involvement in FP in SSA [23]. The challenge that remains however is how to reach out to male partners and facilitate informed FP spousal discussions. Studies in Uganda and South Africa have shown that a simple intervention such as written letter of invitation to a male partner can lead to significant increase in male attendance at antenatal clinics and opportunities for couple counseling [24,25]. Consistent with the need to have a more aggressive approach to ensuring active male participation in PPFP, there is the urgent need to identify additional approaches that will contribute to assuring women of spousal support of their intentions to adopt PPFP.

past use of modern family planning

In this study, we found that the intention to adopt PPFP and the selection of injectables as the method of choice were both significantly influenced by past use of the method. The apparent appreciation of injectables is corroborated by the findings of the 2008 Ghana DHS and a recent review of the uptake of various FP methods in SSA [13,26]. An investigation of the factors that make women in Ghana prefer injectables would most likely provide indications of the interventions that would be required to increase the uptake of PPFP in general. The possibility that injectables make it possible for women to use FP discretely needs to be considered.

In the 2008 Ghana DHS, the proportion of married non-users of FP who indicated the intention to use FP ranged from 43.7% to 53.0% [13]. This range is much lower than the 70% found in this study and this could be due to the fact that all the respondents in this study were pregnant women who would reasonably be expected to have a greater desire to delay repeat pregnancy. Notwithstanding the limitation of using intentions to gauge likely uptake, this finding support the need for pregnant women to be particularly targeted in the promotion of FP in this population.

Implications for the health system in Ghana

The model for delivering antenatal (including communitybased) services in Ghana has seen very little innovation and responsive to the evidence on the need for male participation. The focus has remained on women despite clear evidence that the continued limited impact of the family planning program in Ghana is due largely to the continued neglect of men as equal target [27]. Using a cross-sectional design this study has demonstrated that male partner approval is central to getting pregnant women to adopt PPFP. To facilitate this, a mandatory session of couple counseling should be actively explored by health workers as part of the routine antenatal care of each pregnant woman. It should be made a part of standard ANC protocol and health workers required to ensure adherence during facility and community-based care.

Making the correct choice of PPFP method and receiving the appropriate counseling on it is critical to retaining confidence in the use of the injectables in PPFP. The use of injectables in Ghana is associated a number of health-related myths and misconceptions that need to be addressed [28]. A common misconception is that injectables reduce breast milk and should not be used postpartum until menstruation resumes. To the extent that the injectable is emerging as a most highlyfavored method of FP, including PPFP, there is the need

for injectable-specific, targeted training for health workers involved in PPFP counseling. Non-pregnant women who opt to either start or discontinue injectables in routine family planning clinics should be explicitly informed that the method can be safely used in PPFP.

Conclusion

Male partner approval, past experience with the use of injectables and personal approval of PPFP are the major determinants of the intention of pregnant women in this population to adopt PPFP. Procedures adopted at antenatal and postnatal clinics should be adapted to take these factors into consideration.

Competing interests

The authors declare no conflict of interest.

Authors' contributions

FB was responsible for the conceptualization of the study. SE, FB, YGH, DB were responsible for its design, implementation and production of the initial draft of the manuscript. SE and FB were responsible for finalizing the manuscript after critical review by GQA, JP and KAA. All authors read and approved the final manuscript.

Acknowledgements

We wish to acknowledge the cooperation of pregnant women in all four health facilities where the study was conducted. We also wish to recognize the work of the interviewers in the respective facilities, the heads and staff of the facilities who supported the conduct of the study In diverse ways. We are particularly grateful to the Mfantseman Municipal Health Directorate and the management of the Saltpond Municipal Hospital for allowing the study to be undertaken in their facilities. This research was supported by the University of North Carolina at Chapel Hill, Carolina Population Center, through the USAID-funded MEASURE Evaluation Population and Reproductive Health (PRH) project, under the sub-grant award to the Center for Health Research & Implementation Support (CHRIS) in Ghana.

Author details

¹Department of Community Medicine, University of Cape Coast, Cape Coast, Ghana. ²Centre for Health Research & Implementation Support, Accra, Ghana. ³Family Health Division, Ghana Health Service, Accra, Ghana. ⁴Municipal Director of Health Services, Mfantseman, Ghana. ⁵Medical Superintendent, Saltpond Government Hospital, Saltpond, Ghana. ⁶Columbia University Mailman School of Public Health, New York, USA. ⁷Department of Population and Health, University of Cape Coast, Cape Coast, Ghana.

Received: 25 April 2013 Accepted: 19 July 2013 Published: 22 July 2013

References

- Cates W Jr, Abdool Karim Q, El-Sadr W, Haffner DW, Kalema-Zikusoka G, et al: 1. Global development. Family planning and the millennium development goals. Science 2010, 329:1603.
- Sachs JD, McArthur JW: The millennium project: a plan for meeting the 2. millennium development goals. Lancet 2005, 365:347-353.
- Crossette B: Reproductive health and the millennium development goals: 3 the missing link. Stud Fam Plann 2005, 36:71-79.
 - Warren C, Mwangi A, Oweya E, Kamunya R, Koskei N: Safeguarding
- maternal and newborn health: Improving the quality of postnatal care in 4 Kenya. Int J Qual Health Care 2010, 22:24-30.
- Koblinsky MA, Tinker A, Daly P: Programming for safe motherhood; a guide to action. Health Policy Plan 1994, 9:252-266.
- Prata N, Passano P, Rowen T, Bell S, Walsh J, et al: Where there are (few) skilled birth attendants. J Health Popul Nutr 2011, 29:81-91. 6.
- Agadjanian V, Yabiku ST, Cau B: Men's migration and women's fertility in 7. rural Mozambique. Demography 2011, 48:1029-1048.

- Ross J, Winfrey W: Contraceptive use, intention to use and unmet need during the extended postpartum period. Int Fam Plan Perspect 2001, 27:20–27.
- 27:20-27. Adegbola O, Okunowo A: Intended postpartum contraceptive use among pregnant and puerperal women at a university teaching hospital. Arch Gymecol Obstet 2009, 280:987–992.
- Johnson FA, Madise NJ: Targeting women at risk of unintended pregnancy in Ghana: should geography matter? Sex Reprod Healthc 2011, 2:29-35.
- Council NP: National Population Policy (Revised Edition, 1994). Accra: Government of Ghana; 1994.
- 12 Ngom P. Men's unmet need for family planning: implications for African fertility transitions. *Stud Fam Plann* 1997, 28:192-202.
- Ghana Statistical Service: Ghana demographic and health survey, 2008. Accra Ghana: Calverton, Md.: Ghana Statistical Service; ICF Macro; 2009.
- 14. Mfantseman Municipal Health Directorate: Annual Report Saltpond: Ghana Health Service; 2010.
- DeRose LF, Dodoo FN, Ezeh AC, Owuor TO: Does discussion of family planning improve knowledge of partner's attitude toward contraceptives? Int Fam Plan Perspect 2004, 30:87–93.
- 16 Crissman HP, Adanu RM, Harlow SD: Women's sexual empowerment and contraceptive use in Ghana. Stud Fam Plann 2012, 43:201-212.
- Gebreselassie T, Mishra V: Spousal agreement on preferred waiting time to next birth in sub-Saharan Africa. J Biosoc Sci 2011, 43:385–400.
- 18. Bawah AA: Spousal communication and family planning behavior in Navrongo: a longitudinal assessment. Stud Fam Plann 2002, 33:185–194.
- Becker S, Costenbader E: Husbands' and wives' reports of contraceptive use. Stud Fam Plann 2001, 32:111–129.
- Cleland J, Bernstein S, Ezeh A, Faundes A, Glasier A, et al: Family planning: the unfinished agenda. Lancet 2006, 368:1810–1827.
- 21 Dodoo FN-A, Ezeh AC, TO O: Some evidence against the assumption that approval of family planning is associated with frequency of spouses' discussion of the subject. *Popul Stud* 2001, 55:195–198.
- 22 Shattuck D, Kerner B, Gilles K, Hartmann M, Ng'ombe T, et al: Encouraging contraceptive uptake by motivating men to communicate about family planning: the Malawi male motivator project. Am J Public Health 2011, 101:1089–1095.
- Hartmann M, Gilles K, Shattuck D, Kerner B, Guest G: Changes In couples' communication as a result of a male-involvement family planning intervention. J Health Commun 2012, 17:802–819.
- Byamugisha R, Astrom AN, Ndeezi G, Karamagi CA, Tylleskar T, et al: Male partner antenatal attendance and HIV testing in eastern Uganda: a randomized facility-based intervention trial. J Int AIDS Soc 2011, 14:43.
- Bhalla K, Adofo K, Mock CN, Afukaar F, Appiah N, et al: Non-traditional data sources for injury control: an agenda for action in Ghana. *Inj Prev* 2012, 18:277
- Ross JA, Agwanda AT: Increased use of injectable contraception in sub-Saharan Africa. Afr J Reprod Health 2012, 16:68–80.
- Ezeh AC: The Influence of spouses over each other's contraceptive attitudes in Ghana. Stud Fam Plann 1993, 24:163–174.
- Parr NJ: Discontinuation of contraceptive use in Ghana. J Health Popul Nutr 2003, 21:150–157.

doi:10.1186/1742-4755-10-34

Clte this article as: Eliason *et al.*: Factors influencing the intention of women in rural Ghana to adopt postpartum family planning. *Reproductive Health* 2013 10:34.

Submit your next manuscript to BioMed Central and take full advantage of:

- Convenient online submission
- Thorough peer review
- No space constraints or color figure charges
- Immediate publication on acceptance
- Inclusion in PubMed, CAS, Scopus and Google Scholar
- Research which is freely available for redistribution

Submit your manuscript at www.biomedcentral.com/submit () BioMed Central

Eliason et al. BMC Pregnancy and Childbirth 2014, 14:261 http://www.biomedcentral.com/1471-2393/14/261

RESEARCH ARTICLE

BMC Pregnancy & Childbirth

Open Access

Determinants of unintended pregnancies in _{rural} Ghana

sebastian Eliason^{1*}, Frank Baiden², Barbara A Yankey³ and Kofi Awusabo–Asare⁴

Abstract

Background: Unintended pregnancies may carry serious consequences for women and their families, including the possibility of unsafe abortion, delayed prenatal care, poor maternal mental health and poor child health outcomes. Although between 1993 and 2008, unintended births decreased from 42% to 37% in Ghana, the rate of decline is low, whilst levels are still very high. This raises the need to understand factors associated with unintended pregnancies, especially among women in rural settings where the rates and risks are highest to help improve maternal health.

Method: We collected data from 1,914 pregnant women attending antenatal clinic between January 2012 and April 2012 in four health facilities in the Mfantseman Municipal of the Central Region of Ghana. We used bivariate and multivariate logistic regression analyses to explore how socio-demographic characteristics, past reproductive health experiences, partner characteristics and relations, awareness and past experience with contraceptives, influenced the status of women's current pregnancy (whether intended or unintended).

Results: The mean age of the 1,914 respondents in this study was 25.6 ± 6.5 years. Seventy percent (70%) said the pregnancies they were carrying were unintended. The odds of carrying unintended pregnancy among women with five or more children were higher than those with one to two children [AOR 6.06, 95% CI (3.24-11.38) versus AOR 1.48, 95% CI (1.14-1.93)]. Women with other marital arrangements showed significantly higher odds of carrying unintended pregnancy compared to those married by ordinance (Muslim or Christian wedding). Women not living with their partners exhibited increased odds of having unintended pregnancies compared to women who lived with their partners (AOR 1.72, 95% CI: 1.28 - 2.30). Awareness of traditional methods of family planning (withdrawal and rhythm) was associated with lower odds of having unintended pregnancy compared to non-awareness (AOR 0.66, 95% CI (0.49-0.89).

Conclusions: In this study, important risk factors associated with unintended pregnancies were: parity, living arrangements with partner, marriage by ordinance and awareness of traditional, non-pharmacological contraceptive methods. Family planning interventions targeting different groups of women, especially during the postpartum period, would be essential to reduce rates of unintended pregnancies and promote positive health outcomes.

Keywords: Unintended pregnancy, Family planning, Parity, Contraceptive methods

* Correspondence: sakeliason@yahoo.co.uk Department of Community Medicine, University of Cape Coast, Cape Coast, Gnana

Full list of author information is available at the end of the article



© 2014 Eliason et al; licensee BioMed Central Ltd. This Is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/2.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly credited. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

Background

Unintended pregnancies refer to pregnancies that are not wanted or those that are mistimed at the time of conception [1]. Out of the 208 million pregnancies estimated worldwide, in 2008, 41% were unintended [2]. Rates of unintended pregnancies though declining world-wide are still high. Rates of unintended pregnancies declined by 20% from 71 to 57 per 1000 from 1995 to 2008 among women aged 15 to 44 years in Low- and Middle-Income Countries (LMICs) [3]. In 2008, 75 million women in LMICs reported that their pregnancies were unintended [4] with 23% of these pregnancies occurring in Sub-Saharan Africa [5]. Guttmacher Institute and the United Nations Population Fund (UNFPA) estimated the level of unintended pregnancies in 2008 at 49 per 1000 pregnancies in Asia, 72 per 1000 in Latin America and the Caribbean and for women aged 15 -44 years in Africa, 86 per 1000; that of Africa was rated as the highest [6] and stated that in Ghana, 37 percent of all births are unintended [7].

Unintended pregnancies may carry serious consequences for women and their families, including possible unsafe abortion, delayed prenatal care, poor maternal mental health, reduced mother/child relationship quality, poor developmental outcomes for children, physical abuse and violence against women, increased risk of low birth weight of babies as well as increased maternal morbidity and mortality [2,8,9]. Available data suggests that induced abortion and related complications are the most common outcomes of unintended pregnancies [10]. It is estimated that in Ghana, induced abortions account for about 12% of maternal deaths, third after hemorrhage (22%) and unclassified causes (14%) [10]; furthermore, the proportion of unintended births also showed a decreasing trend: From 1993 to 2008, unintended births decreased from 42% to 37% [11]. In-spite of this decrease, the rate continues to be high and is estimated to be about 0.7 per woman [12,13].

The high rate of unintended pregnancies in Sub-Saharan Africa, including Ghana, attests to poor access to reproductive health care especially family planning, inadequate reproductive health rights and low empowerment of women. Partly due to these prevailing situation, Ghana and most Sub-Saharan African countries are not likely to attain Millennium Development Goals (MDGs) 3, 4 and 5. Targeted interventions, especially during the postpartum period when an unintended pregnancy can be of great risk to mother and baby [14] would be essential, if the rates of unintended pregnancies are to be reduced, to promote positive health outcomes. To achieve this objective, factors that are associated with unintended pregnancies need to be investigated and understood. Studies conducted in the United States, Asia, Middle East and Latin America have revealed several demographic and socio-economic factors as predictors of unintended pregnancies; among them are contraceptive failure, lack of access to contraception, religious beliefs and poor knowledge on contraception, fertility and pregnancy, a history of previous unintended pregnancy, insufficient reproductive health education, desire for at least two children, parity of five, lack of communication or support within the relationship, husband's reluctance to limit family size, and sexual violence [15-18].

In Ghana, where the situation is critical, very few studies have been undertaken on unintended pregnancies. For instance, one study [12] only detailed analysis of the predictors of unintended pregnancies. Some are listed as age, marital status, abode, educational status, profession, gravidity and parity, poverty or inadequate resources for raising a child, stigma against unmarried mothers, a cultural preference for sons, completion of family size, disagreement between spouses about family size, poor access to family planning services, and poor understanding of risks associated with unintended pregnancy [3,12]. The objective of this study is to contribute to the search for predictors of unintended pregnancies in Ghana through a survey among pregnant women attending antenatal clinics in rural and semi-urban health facilities in the Mfantseman Municipal of the Central Region.

Methods

This study was part of a bigger study on the factors influencing the intention of women in rural Ghana to adopt post-partum family planning (PPFP). The method has been described in an earlier publication [19]. The study was conducted at four health facilities in the Mfantseman Municipal of the Central Region. The area was selected in the Central Region because of consistent reports of adverse maternal health/family planning (FP) outcomes: It reported the highest level of teenage pregnancy (13.6% of all pregnancies in 2010), high incidence of induced abortions and low level of unmet need [11,20].

For this study, all pregnant Ghanaian women living in the municipal and attending antenatal clinic at the Saltpond Government Hospital, Mankessim Health Centre (located in semi-urban settings - i.e.-demographically urban with population of about 42,000, economically agro-based with values, attitudes, tastes and behaviours characteristic of both urban and rural settings) and the Biriwa and Anomabo Health Centers (both rural) between January 2012 and April 2012 were targeted for interview within the premises of the health facilities, using a five-page questionnaire. Questions related to socio-demographic background, socio-demographic characteristics of male partners, issues pertaining to the nature of relationships between the respondents and the male partners, respondents' reproductive history including status of current pregnancy (whether wanted, unwanted or mistimed), awareness and ever use of various Family

Planning (FP) methods and the intention to use FP after delivery.

A total of 1900 pregnant women were targeted to be interviewed on the assumption that 50% of them will have the intention to adopt PPFP. With 80% power, it was possible to estimate the proportion of women willing to adopt PPFP within a margin of error of 3%. The municipal recorded about 4000 deliveries in 2009.

Cleaned data were exported into STATA/IC (version 11.2) for analyses. Descriptive and bivariate logistic regression analyses were conducted under the various subthemes of the questionnaire. Two models were used in the bivariate logistic regression analyses. In Model I, tests of association were conducted between eighteen (18) independent variables and the outcome variable (unintended pregnancy). Fifteen (15) of the independent variables were found to be significantly associated (P < 0.05) with the outcome variable; however, since P < 0.05 is not appropriately robust to determine which associations were real and which were by chance, given so many tests, a second model (Model II) was introduced.

In Model II, the significance level threshold was set higher to 0.003 by conducting a Bonferroni correction. Factors found to be significantly associated with the main outcome of interest, were included in a multivariate logistic regression model (Model III), to identify significant independent predictors of unintended pregnancy. Tests of covariance were conducted among all the significant variables and those found to show covariance were dropped from model III.

The outcome variable (unintended pregnancy) was defined as any pregnancy that was not wanted at all at the time it occurred or in the future, or mistimed i.e. wanted at a later time but not at the time it occurred. An intended pregnancy was defined as any pregnancy that was wanted at the time it occurred or wanted at an earlier time but occurred later.

Ethical approval was obtained from the Ethics Review Committee of the Ghana Health Service (GHS). Institutional approval was also obtained from the Municipal Health Directorate (MHD) and the heads of the facilities where the survey was conducted. Written informed consent was obtained from each participant before the administration of questionnaires.

Results

Background characteristics of respondents and their relation to overall pregnancy status

A total of 1,914 pregnant women were interviewed. The mean age of these women was 25.6 ± 6.5 years, with the majority (29.7%) aged between 20–24 years. Majority (70%) indicated that the pregnancies they were carrying were unintended (mistimed 39%, unwanted 31%). There were more unintended pregnancies reported among

younger (90%) than older women (80%), (P < 0.001). There was a trend towards reduced unintended pregnancies with increasing level of education. Prevalence of unintended pregnancies was high among all religious groups with the highest being among the traditionalists (82%). Prevalence of intended pregnancies was relatively higher amongst the Muslims (43%) and Catholics (36%), (P < 0.001). Expectant mothers with five or more children had high prevalence (61%) of unwanted pregnancies compared to those with up to four children. Of the 256 respondents who were single, only a tenth of the pregnancies were intended, in contrast to those who were married. A third of those who were married traditionally, engaged or cohabiting had intended pregnancies. The prevalence of unintended pregnancies among students (n = 124) was noticeably high (90%) compared to those who were employed in the formal sector as civil/public servants (32% of the 84 respondents). Of those in the informal sector (petty traders, fishmongers and farmers), three out of every four pregnancies were unintended. Intended pregnancies amongst those living in the two semi-urban settlements (Mankessim and Saltpond) were higher than those in the rural areas (Biriwa and Anomabo) (35% versus 20%), (P < 0.001) (Table 1).

Bivariate logistic regression analyses of unintended pregnancies on independent variables

Unintended pregnancy (outcome variable) was regressed on each of the identified independent variables (Table 2, Model I). Only those independent variables that were found to be significantly correlated (P < 0.05) with the outcome were subjected to Bonferroni's correction (Table 2, Model II). Women aged 20 years and above, had significantly lower odds of having unintended pregnancy (OR 0.83, 95% CI 0.77-0.89). Some factors that were found to be significantly associated with increased odds of unintended pregnancies are: not being married by ordinance (Muslim or Christian wedding) (OR 1.41, 95% CI 1.30-1.52); partner not living in the same house as the woman (OR 2.15, 95% CI 1.70 -2.72) and high parity (OR 1.20, 95% CI 1.12-1.29).

Respondents who were aware of modern and traditional family planning methods, and had ever used traditional methods showed significantly lower odds of carrying unintended pregnancy [OR 0.40, 95% CI(0.25-0.62); OR 0.50, 95% CI(0.40-0.64); OR 0.68, 95% CI(0.55-0.82) respectively]. Education was not found to be a significant factor influencing unintended pregnancies in this study.

Multivariate logistic regression analyses

Controlling for all factors listed in Model III, increasing parity was significantly associated with increasing odds of unintended pregnancy. The odds of carrying unintended pregnancy among women with five or more children were

Table 1 Background characteristics by pregnancy status

			Overall pregnancy status (%)			
Demographic characteristics	Sample size	Percent of total sample size (%)	intended	Mistimed	Unwanted	Unintended
Age**						
15-19	340	17.8	9.2	32.3	58.5	90.8
20-24	569	29.7	31.7	45.0	23.4	68.3
25-29	483	25.3	37.7	42.0	20.3	62.3
30-34	291	15.3	38.1	35.1	26.8	61.9
35-39	172	9.0	28.5	36.1	35.5	71.5
40+	56	2.9	19. 6	25.0	55.4	80.4
TOTAL	1914	100.0	29.6	39.1	31.4	70.4
Education level**						
None	414	21.6	23.0	39.0	38.0	77.0
Primary	429	22.4	24.8	39.3	36.0	75.2
Middle/J5S	843	44.1	30.0	40,4	29.6	70.0
SSS/SHS/VOC	166	8.7	41.2	36.4	22.4	58.8
Tertiary	62	3.2	69.4	27.4	3.2	30.7
TOTAL	1,914	100.0	29.6	39.1	31.3	70.4
Religion**						
Christlan	1,783	93.2	29.3	39.0	31.7	70.7
Muslim	88	4.6	39.8	36.4	23.9	60.2
Traditionalist	10	0.5	20.0	60.0	20.0	80.0
Other	33	1.7	22.6	41.9	35.5	77.4
TOTAL	1,914	100.0	29.6	39.1	31.3	70.4
Gravidity**						
1 -2	1,025	53.8	31.8	36.7	31.5	68.2
3 ~4	531	27.9	30.5	46.9	22.6	69.5
5+	348	18.3	21.6	34.2	44.3	78.5
TOTAL	1,904^	100.0	29.6	39.1	31.4	70.4
Parity**						
0	673	35.2	30.9	31.1	38.0	69.1
1 -2	77 2	40.3	34.1	46.3	19.6	65.9
3 -4	353	18.4	23.1	42.2	34.8	76.9
5+	116	6.1	11.2	27.6	61.2	88.8
TOTAL	1,914	100.0	29.6	39.1	31.4	70.4
Marital Status**						
Married by Ordinance (Church/mosque)	236	12.4	49.6	32.6	17.8	50.4
Married (Traditional)	857	44.8	29.3	42.1	28.6	70.7
Engaged	282	14.7	38.1	40.2	21.7	61.9
	267	14.0	24.2	45.3	30.6	75.9
Divorced/Sen/Midowed	8	0.4	0.0	62.5	37.5	100.0
Sinola	256	13.4	9.6	26.7	63.8	90.4
Other	6	0.3	0.0	20.0	80.0	100.0
	1 9174	100.0	29.6	39.1	31.3	70.4
VIAL	1,2 (A) (C)					

Occupation**								_
Fishmonger	318	16.6		21.5				
Farmer	67	35		21.5	38.6	39.9	78.5	
Petty trader	913	179		19.4	43.3	37.3	80.6	
Civil/Public Servant	84	47.0		28.1	42.3	29.6	71.9	
Student	174	4.4		67.9	23.8	8.3	32.1	
Other	124	20		9.8	30.9	59.4	90.2	
ΤΟΤΑΙ	406	21.2		39.0	37.0	24.0	61.0	
Area of residencets	1,912^	100.0		29.6	39.1	31.3	70.4	
Saltaand								
Disture	422	22.2		36.3	38.9	24.9	63.7	
biriwa	231	12.1		21.2	35.9	42.9	78.8	
Anomabo	324	17.0		20.1	41.1	38.9	79.9	
Mankessim	567	29.8		34.0	36,5	29.5	66.0	
Other	358	18.8		28.8	43.6	27.7	71.2	
TOTAL	1,902^	100.0		29.6	39.1	31 3	70.4	
Religious DenomInation**						5115	70.4	
Catholic	199	11.2		35.68	3216	32.16	643	
Protestant/Charis/pent	1,361	76,9		30.05	39.75	20.0	70.0	
Muslim	81	4.6		43.71	30.51	17.20	70.0	
Traditionalist	74	42		1767	10.00	17.28	56.8	
No/other religion	54	31		17.27	40.00	33./8	82.4	
TOTAL	1 7690	100.0	•	22.22	40.3	31.48	77.8	
	11.02	100.0		20.23	39.40	30.02	69.5	

Table 1 Background characteristics by pregnancy status (Continued)

*Unintended (mistimed + unwanted), (Pearson Chi2 Statistic - **p < 0.001), ^observed differences in total sample sizes (1914) are due to missing values.

higher than those with one to two children [AOR 6.06, 95% CI (3.24-11.38) versus AOR 1.48, 95% CI (1.14-1.93)]. Women with other marital arrangements showed significantly higher odds of carrying unintended pregnancy compared to those married by ordinance (Muslim or Christian wedding). Single women showed the highest odds of carrying unintended pregnancy [AOR 7.32, 95% CI (4.21-12.75]. Women not living with their partners exhibited increased odds of having unintended pregnancies compared to women who lived with their partners (AOR 1.72, 95% CI: 1.28 - 2.30). Awareness of traditional methods of family planning (withdrawal and rhythm) was associated with lower odds of having unintended pregnancy compared to non-awareness (AOR 0.66, 95% CI (0.49-0.89) (Table 2 Model III).

Discussion

Factors which were identified to be significantly associated with the tendency to consider the pregnancy which women were carrying at the time of the survey to be unintended, included parity, marital arrangement, living arrangement with partner and awareness of traditional methods of contraception.

High parity was significantly associated with unintended pregnancy. The expectation was that the level of

unintended pregnancy would be lower with increasing parity. The result indicating high odds of unintended pregnancies with increasing parity among women is an observation which would need further investigation despite similar findings from other studies. [12,21,22]. The relatively low exposure of rural women in Ghana to modern family planning [10] could partly explain this finding. Another possibility is that, couples looking for a particular gender may end up having more children than intended; there is evidence that parents wanting to balance the sex of their children will continue to give birth if all the children are of the same sex and especially if parents have a desire for a son. Chaudhuri, S, demonstrated from a study in India that the desire for sons, or not having any son, was associated with an increase in parity progression [23]. This finding supports prior research in South East Asia [24-27].

Studies have persistently demonstrated higher odds of unintended pregnancy among partners with other marital arrangements compared to married couples [28-31]. Lachance-Grzela & Genevieve Bouchard, explain that the advantage of married couples generally having favourable and healthier pregnancies than unmarried couples occurs only when the pregnancies are intended [28]. The finding that, women who reported other forms of marital

Table 2 Binary logistic regression analyses: models I&II (bivariate analyses) and model III (multivariate analyses)

	Model 1		Model II Model III			
Variables	OR (95% CI) Unadjusted P-value		Bonferroni adjusted Byvaluo	Adjusted Odds Ratio (AOR)	P-value	
Age (Ref: 15–19)				NA		
20-24	0.83 (0.77-0.89)	< 0.001	<0.001	INA	NA	
25-29			(0.001			
30-34						
35-39						
40+						
Educ status (Ref: none)				NΔ	NIA	
Primary	0.98 (0.92-1.03)	0.45	NA		IN/A	
Middle/JSS						
SSS/SHS/Vocational						
Tertiary						
Ethniclty (Ref: fante)				NA	NA	
Others	1.00 (0.98-1.02)	0.89	NA			
Religion (Ref: christian)				NA	NA	
Muslim	1. 04 (0.99 -1.09)	0.15	NA			
Traditionalist						
Others						
Parity (Ref: 0)						
1 – 2	1.20 (1.12-1.29)	<0.001		1.48 (1.14-1.93)	0.004	
3 – 4				2.64 (1.88-3.71)	<0.001	
5+			<0.001	6.06 (3.24-11.38)	<0.001	
Marital status (Ref: by ordinance)						
Traditional rites	1.41 (1.30-1.52)	<0.001	<0.001	1.81 (1. 3 3-2.45)	<0.001	
Engaged, yet to be married				1.58 (1.10-2.28)	0.014	
Cohabitation				2.91 (1.96-4.31)	<0.001	
Single				7.32 (4.21-12.75)	<0.001	
Partner age (Ref: 15–19)				NA	NA	
20-29	0.97 (0.95-0.9 8)	<0.001	<0.001			
30-39						
40+						
Partner religion (Ref: christian)				NA	NA	
Muslim	1.25 (1.06-1.47)	0.009	0.135			
Traditionalist						
Others						
Partner has Chn. from other women (Ref: yes)				NA	NA	
No	1.33 (1.08-1.64)	0.008	0,120			
Years of marrlage/relationship (Ref: <1Yr.)				NA	NA	
	1.02 (1.00-1.05)	0.019	0.285			
Partner lives in same house as woman (Ref: yes)						
No	2.15 (1.70-2.7 2)	<0.001	<0.001	1.72 (1.28-2.30)	<0.001	
^a rtner has other spouses (Ref: yes)				NA	NA	
No	1.47 (1.06-2.23)	0.019	0.285			

Table 2 Binary logistic regression analyses: mod	lels I&II (bivariate analycor) and model ut (
(Continued)	(multivariate analyses) and model III (multivariate analyses)

Gravidity (Ref: 1–2)					
3-4	1.08 (1.12-1.14)	0.004	0.050	NA	NA
5+	, <u>-</u> ,	0.004	0.000		
Previous abortions/miscarriages (Ref: yes)				NA	
No	1.00 (1.00-1.0045)	0.043	0.645	NA	NA
first pregnancy			2.013		
Awareness of modern FP (Ref: no)					
Yes	0.40 (0.25-0.62)	< 0.001	<0.001	0.70 (0.42-1.17)	0172
Awareness of traditional FP (Ref: no)				0.70 (0.92 1.17)	0.175
Yes	0.50 (0.40-0.64)	<0.001	<0.001	0.66 (0.49-0.89)	0.007
Ever use of modern FP (Ref: no)				NA	0.007
Yes	0.78 (0.64-0.95)	0.014	0.210	10,	110
Ever use of traditional FP (Ref: no)					
Yes	0.68 (0.55-0.82)	<0.001	<0.001	0.95 (0.75-1,21)	0.672

arrangements had higher odds of unintended pregnancies compared to those married under the ordinance, presents an issue for further investigation within the Ghanaian context. It is possible that forms of marital arrangements could have implications for stability of marriage and therefore the possibility of planning pregnancies. The high odds of unintended pregnancies among single or unmarried women are not unexpected. This is especially so when pregnancy is considered to be a prelude to marriage or for solidifying a relationship [29]. Fear of infertility in future marital unions is a major driver behind this in some communities.

Two non-pharmacological contraceptive methods proved beneficial in preventing unintended pregnancies in this study. Women who were aware of withdrawal and rhythm as protective measures against unintended pregnancy, were less likely to have unintended pregnancies compared to those who were not aware. The socio-cultural context within the rural setting, myths and fear of side effects of modern contraceptives possibly influenced this finding.

As observed by Ikamari and colleagues [21] also, formal education was not significantly associated with pregnancy intendedness, contrary to expectation on this correlate. There was however, a trend towards reduced unintended pregnancy with increasing level of education (Table 1), which is consistent with other studies [12,32]. It could be an emerging issue which would need further investigation within the Ghanaian context.

Conclusions

This study has highlighted several factors associated with unintended pregnancy: parity, marital arrangement, living arrangement with partner and awareness of traditional, non-pharmacological contraceptive methods.

These results indicate that various categories of women would need to be targeted differently for family planning messages and services. For instance, the National Centre for Civic Education (NCCE), Ghana Health Service, religious bodies and Non-Governmental Organizations need to intensify the campaign on the importance of couples to opt for marriage by ordinance, since it has several advantages over other forms of marital arrangement. This study revealed that if partners lived together, the probability of unintended pregnancies may reduce. Marriage by ordinance may further strengthen this relationship and help to reduce unintended pregnancies. Campaigns on sex balancing aimed at encouraging parents to accept the sex of the children they have could be carried out, in order to limit the tendency to higher parity progression. This could be fairly easy given the fact that there are no obvious sex preferences in Ghana.

Family planning programmes may need to consider promotion of traditional, non-pharmacological methods alongside the modern methods, especially in rural settings, to improve overall contraceptive prevalence rates. For clients who do not want to adopt modern contraceptives despite all reassurances, the option of traditional methods should be offered them. This implies that health workers may need to be trained adequately to provide such services. Commitment from the Ghana Health Service Family Planning Programme would be required if this is to succeed.

Unintended pregnancy may be of greatest risk to mother and baby during the postpartum period. Family planning interventions, especially targeting this period, would be essential if the rates of unintended pregnancies are to be reduced, to promote positive health outcomes. In connection with this, pregnant women attending antenatal clinic (ANC) need to be targeted for family planning counseling before they deliver. Couple counseling should be actively explored by health workers as part of the routine antenatal care of each pregnant woman. It should be made a part of standard ANC protocol and health workers required to ensure adherence during facility and community based care.

Study limitations

The threat of selection bias existed, but was highly mitigated, by ensuring that, the data collectors explained the study objectives and their implications very well to the respondents, before asking for consent. The time for the study was short; and this was imposed by limited funding and strict reporting requirements by funding agency. Some of the data collectors abandoned the study because of inadequate remuneration. New data collectors had to be trained to continue data collection. This brought about some delays in data analysis and reporting.

Abbreviations

ANC: Antenatal clinic; AOR: Adjusted odds ratio; CI; Confidence interval; FP: Family planning; GHS: Ghana Health Service; LMIC: Low-and middle-income countries; MDG: Millennium development goals; MHD: Municipal Health Directorate; NCCE: National Commission for Civil Education; OR: Odds ratio; PPFP: Postpartum family planning; UNFPA: United Nations Population Fund.

Competing Interests

The authors declare that they have no competing interest.

Authors' contributions

SE and FB were responsible for the conceptualization of the study. SE and FB were responsible for its design and implementation. SE and BY were responsible for production of the initial draft of the manuscript. SE was responsible for finalizing the manuscript after critical review by KAA. All authors read and approved the final manuscript.

Authors' information

SE is a Public Health Physician and Lecturer in the Department of Community Medicine, university of Cape Coast, Ghana; FB is a Public Health Physician and Senlour Researcher at Centre for Health Research and Implementation Support; BY is a Pharmacist and PhD Candidate at Georgia State University, Atlanta, USA; KAA is a Professor in Population and Health at the Faculty of Social Science, University of Cape Coast, Ghana.

Acknowledgements

We wish to acknowledge the cooperation of pregnant women in all four health facilities where the study was conducted. We also wish to recognize the work of the interviewers in the respective facilities, the heads and staff of the facilities who supported the conduct of the study in diverse ways. We are particularly grateful to the Mfantseman Municipal Health Directorate and the management of the Saltpond Municipal Hospital for allowing the study to be undertaken in their facilities.

Author details

¹Department of Community Medicine, University of Cape Coast, Cape Coast, Ghana. ²Centre for Health Research and Implementation Support, Accra, Ghana. ³Georgia State University, Atlanta, USA. ⁴Department of Population and Health, University of Cape Coast, Cape Coast, Ghana.

Received: 5 January 2014 Accepted: 28 July 2014 Published: 8 August 2014

References

- Centers for Disease Control: UnIntended Pregnancy Prevention: Home. http://www.cdc.gov/reproductivehealth/unintendedpregnancy/.
- Singh S, Sedgh G, Hussain R Unintended Pregnancy: Worldwide Levels, Trends, and Outcomes. Stud Fam Plan 2010, 41:241–250.
- Singh S, Sedgh G, Hussain R: Levels and trends In unintended pregnancy worldwide. In paper presented at the annual meeting of the Population Association of America, Detroit, MI, USA, April 30th – May 2nd. 2009.
- Singh S, Wulf D, Hussain R, Bankole A, Sedgh G: Abortion Wolldwide: A Decade al-Uneven Progress. New York: Guttmacher Institute; 2009.
- Guttmacher Institute: Special analyses. In Adding It Up: The Costs and Benefits of Investing in Family Planning and Maternal and Newborn Health. Edited by Singh S, et al. New York: Guttmacher Institute and United Nations Population Fund; 2009.
- Singh S, Daroch JE, Ashford LS, Vlassoff M: Adding It Up: The Costs and Benefits of Investing in Family Planning and Maternal and Newborn Health, New York: Guttmacher Institute and United Nations Population Fund; 2009.
- Sedgh G: Abortion in Ghana, In Brief. New York: Guttmacher Institute; 2010. Series No. 2.
- Santelli J, Roachat R, Hatfield-Timajchy K, Gilbert BC, Curiis K, Cabral R, Hirsch JS, Schleve L: The measurement and meaning of unintended pregnancy. Perspect Sex Reprod Health 2003, 35:94–101.
- U.S. Department of Health and Human Services: Health Resources and Services Administration, Maternal and Child Health Bureau. In Women's Health USA. Rockville, Maryland: U.S. Department of Health and Human Services; 2011.
- Ghana Statistical Service (GSS), Ghana Health Service (GHS), and Macro International: Ghana Maternal Health Survey 2007. 2009.
- Ghana Statistical Service: Report of the Ghana Demographic and Health Survey 2008. 2009.
- Omane-Adjepong M, Oduro FT, Annin K: A multinomial regression analysis of unplanned pregnancies in Ahafo Ano South District, Ghana. Am Int J Contemp Res 2012, 2:12.
- Johnson FA, Madise NJ: Targeting women at risk of unintended pregnancy in Ghana: Should geography matter? Sex Reprod Healthc 2011, 2(1):29–35.
- WHO, USAID, JHPIEGO, MCHIP: Statement for Collective Action for Postpartum Family planning, 2012.
- Goicolea and San Sebastian: Unintended Pregnancy in the Amazon Basin of Ecuador: a multilevel analysis. Int J Equity Health 2010, 9:14.
- Glasier A, Guimezoglu AM, Schmid GP, Moreno CG, Van Look PF: Sexual and reproductive health: a matter of life and death. *Lancet* 2006, 368(9547):1595–1607.
- Rosenfeld JA, Everett KD: Factors related to planned and unplanned pregnancies. J Fam Pract 1996, 43(2):161–166.
- Amin SF, Howden C, Peynan N: A Comparison Study: Risk Factors of Unplanned Pregnancies in a Group of Iranian and New Zealander Women. Eur J Sci Res 2009, 26(1):108–121.
- Eliason S, Baiden F, Quansah-Asare G, Graham-Hayfron Y, Bonsu D, Philips J, Awusabo-Asare K: Factors influencing the Intention of women in rural Ghana to adopt postpartum family planning. *Reprod Health* 2013, 10:34.
- Mfantseman Municipal Health Directorate: Annual Report. Saltpond: Ghana Health Service, 2010.
- Ikamari L, Izugbaru C, Oghako R: Prevalence and determinants of unintended pregnancies in Nairobi, Kenya. BMC Pregnancy Childbirth 2013, 13:69.
- Hamdela B, G/mariam A, Tilahun T: Unwanted Pregnancy and Assoclated Factors among Pregnant Married Women in Hosanna Town, Southern Ethlopia. PLoS ONE 2010, 7(6):e39074.
- Chaudhuri S: The Desire for Sons and Excess Fertility: A Household Level Analysis of Parity Progression in India. Int Perspect Sex Reprod Health 2012, 38(4):78–186.
- Bairagi R: Effects of sex preference on contraceptive use, abortion and fertility in Matlab, Bangladesh. Int Fam Plan Perspect 2001, 27(3):137–143.
- Arnold F: Son preference in South Asia. In Fertility Transition in South Asia. Edited by Sathar ZA Phillips JF. London: Oxford University Press; 2001:281–299.
- Das N. Sex preference and fertility behavior: a study of recent Indian data. Demography 1987, 24(4):517–530.
- Gupta MD, Zhenghua J, Bohua L, Zheming X, Chung W, Hwa-ok B: Why is son preference so persistent In East and South Asia? A cross-country study of China, India, and the Republic of Korea. In World Bank Policy Research Worlcing Paper. Washington, DC: World Bank; 2002. No. 2942.

- Lachance-Grzela M, Bouchard G: The Well-Being of Cohabiting and Married Couples During Pregnancy: Does Pregnancy Planning Matter? J Soc Pers Relat 2009, 26:141.
- Bouchard G: Adult couples facing a planned or an unplanned pregnancy: Two realities. J Fam Issues 2005, 26:619–637.
- Korenman S, Kaestner R, Joyce TJ: Unintended pregnancy and the consequences of non-marital childbearing. In Out of wedlock: Causes and consequences of non-marital fertility. Edited by Wu LL, Wolfe B. New York: Russel Sage Foundation; 2001;259–286,
- Thomson E. Partnership and parenthood: A comparative view of cohabitation, marriage, and childbearing. In The new population problem: Why families in developed countries are shrinking and what it means. Edited by Booth A, Crouter AC. 2005;129–149.
- Calvert C, Baisley K, Doyle AM, Maganja K, Changalucha J, Watson-Jones D, Hayes RJ, Ross DA: Risk factors for unplanned pregnancy among young women in Tanzania. J Fam Plann Reprod Health Care 2013, 39(4):e2.

doi:10.1186/1471-2393-14-261

Cite this article as: Eliason et al.: Determinants of unIntended pregnancies in rural Ghana, BMC Pregnancy and Childbirth 2014 14:261.

Submit your next manuscript to BioMed Central and take full advantage of:

- Convenient online submission
- Thorough peer review
- No space constraints or color figure charges
- Immediate publication on acceptance
- Inclusion in PubMed, CAS, Scopus and Google Scholar
- Research which is freely available for redistribution

() BioMed Central

SAM JONAH LIBRARY UNIVERSITY OF CAPE COAST

Submit your manuscript at www.biomedcentral.com/submit